USING VALUE CHAIN APPROACHES IN AGRIBUSINESS AND AGRICULTURE IN SUB-SAHARAN AFRICA



A METHODOLOGICAL GUIDE

Tools That Make Value Chains Work: Discussion and Cases

> Prepared for the World Bank J.E. Austin Associates, Inc. Principal Author: Martin Webber, J.E. Austin Associates, Inc. Program Director: Patrick Labaste, The World Bank

Acknowledgements

This Guide was prepared by J.E. Austin Associates for the Sustainable Agriculture Group of the World Bank. The work was directed by Mr. Patrick Labaste of the World Bank. The Principal Author and team leader was Mr. Martin Webber, Executive Vice President of J.E. Austin Associates, Inc.

Principal contributors and coauthors from J.E. Austin Associates included Marcos Arocha, Virginia Brandon, Lisa Carse, Grant Cavanaugh, Michael Ducker, David Feige, Michael Gorman, Carlton Jones, Jennifer Lynch, Jane Shearer, Justin Stokes, and Gina Tumbarello. Principal World Bank collaborators were Malick Antoine and Jean-Luc Bosio. Jean Michel Voisard also served as reviewer.

The authors convened a roundtable of industry experts, who provided suggestions and experiences that enhanced this Guide. The roundtable participants included : Graeme Dixie, Richard Henry, Steve Jaffee, Svetlana Meades, Paul Siegel, Yolanda Strachan and Uma Subramanian of the World Bank Group ; and Susan Bornstein (TechnoServe), Jeanne Downing (USAID), Paul Guenette (ACDI-VOCA), Olaf Kula (ACDI-VOCA), John Lamb (Abt), Frank Lusby (Action for Enterprise) and Lynn Salinger (AIRD).

Many, many people contributed their expertise and experience, particularly to the case studies. They shared their own stories, and their analyses and observations. Their personal experiences and generous willingness to share their stories and conclusions make this Guide a particularly rich source of ideas and information. We sincerely apologize for any omissions, which, of course, are inadvertent.

Table of Contents

Acknowledgements	i
Overview	vii
Section 1: Introduction	. 1
Agricultural Value Chains and Africa's Development Agenda Opportunities and Threats of Trade and Market Globalization for African Economies Comparison of Agricultural Productivity Objectives of the Guide Organization of the Guide	. 1 . 2 . 4 . 6 . 8
Section 2: Concepts and Definitions of Value Chains/Supply Chains	. 9
Value Chains: Definition of Concepts How Value-Chain Analysis Has Been Used Analyzing and Evaluating Value Chains Limitations of Value Chain Analysis Implications of the Operating Environment: Beyond the Value Chain	.9 10 11 12 14
Section 3: Review of Existing Literature on Value/Supply Chains	15
Summary of General Themes and Structure from the Literature Review Creating Trust Governance: What Type of Power Relationship Exists and Is Information Shared? Market or Bargaining Power (Buyer vs. Producer) Innovation, Information, Information Sharing, and Knowledge Intervention Entry Points and Initial Focus	20 20 22 23 25 26
Section 4: Discussion of Individual Tools	28
Introduction and Overview Theme One: Designing strategies and business plans—Obtaining and using informati	28 ion 28
Theme Two: Developing robust new businesses Theme Three: Supplying the market—Aligning supply to match market opportunity Theme Four: Reaching the market—Market positioning and market opportunities Theme Five: Improving the business and policy environment Theme Six: Building partnerships Theme Seven: Monitoring results in value chain development	28 28 29 29 29 29 29
Tool 1: Choosing Priority Sectors for Value Chain Interventions	32
Initial List	32

	How to Choose Value Chains for Intervention: The Example of Ser	legal's
Project C	roissance Economique	34
Market Ana	llysis	35
Domestic C	apacity and Economic Impact	35
Testing Cor	nmitment	35
A Step-by-s	Step Summary of 1001 1: Choosing Priority Sectors for Value Chain	36
Interventior	15	50
Case Study: Prio in Mozambique .	ritizing Value Chains by Using Comparative Analysis—Value Chai	n Selection 37
Case Study: Geo	omar - Innovative Approaches to Assessing the Competitiveness of V	alue Chains
Tool 2: Design	ing Informed Strategies Across the Value Chain	43
Enhancing '	Value Chain Productivity	43
Assessing th	he Overall Status of the Value Chain	44
Assessing S	strategic Productivity	48
Assessing C	Operational Productivity	48
Assessing the	he Quality of Supply Chain Management	49
Assessing F	Iuman Resources Across the Value Chain	50
8		51
Case Study: Und Domestic Catfish	lerstanding the Value Chain and Integrating Information Into Strateg	51 y : Nigerian 52
Case Study: Und Domestic Catfish Tool 3: Condu	erstanding the Value Chain and Integrating Information Into Strateg cting Benchmarking and Gap Assessments of Value Chains	51 y: Nigerian 52 59
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark	erstanding the Value Chain and Integrating Information Into Strateg cting Benchmarking and Gap Assessments of Value Chains	51 y : Nigerian 52 59 59
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark The Benchr	erstanding the Value Chain and Integrating Information Into Strateg cting Benchmarking and Gap Assessments of Value Chains ing Background narking Tool	51 Nigerian 52 59 59 59
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark The Benchr Box 2:	erstanding the Value Chain and Integrating Information Into Strateg cting Benchmarking and Gap Assessments of Value Chains ing Background narking Tool Uganda Benchmarking Constraints in the Coffee Industry	51 Nigerian 52 59 59 59 59 61
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark The Benchr Box 2: Box 3:	erstanding the Value Chain and Integrating Information Into Strateg cting Benchmarking and Gap Assessments of Value Chains ing Background narking Tool Uganda Benchmarking Constraints in the Coffee Industry Tanzania Cotton—Benchmarking Costs	51 Nigerian 52 59 59 59 61 61
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark The Benchr Box 2: Box 3: A Step-by-S Value Chair	erstanding the Value Chain and Integrating Information Into Strateg cting Benchmarking and Gap Assessments of Value Chains ing Background narking Tool Uganda Benchmarking Constraints in the Coffee Industry Tanzania Cotton—Benchmarking Costs Step Summary of Tool 3: Conducting Benchmarking and Gap Assess ns	 51 Nigerian 52 59 59 61 61 sments of 66
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark The Benchr Box 2: Box 3: A Step-by-S Value Chain Case Study: Uga	erstanding the Value Chain and Integrating Information Into Strateg cting Benchmarking and Gap Assessments of Value Chains ing Background narking Tool Uganda Benchmarking Constraints in the Coffee Industry Tanzania Cotton—Benchmarking Costs Step Summary of Tool 3: Conducting Benchmarking and Gap Assess ns	 51 v: Nigerian 52 59 59 61 61 sments of 66 68
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark The Benchr Box 2: Box 3: A Step-by-S Value Chain Case Study: Uga Tool 4: Upgrad	erstanding the Value Chain and Integrating Information Into Strateg cting Benchmarking and Gap Assessments of Value Chains ing Background narking Tool Uganda Benchmarking Constraints in the Coffee Industry Tanzania Cotton—Benchmarking Costs Step Summary of Tool 3: Conducting Benchmarking and Gap Assess ns nda Floriculture: Benchmarking and Gap Analysis ding and Deepening the Value Chain	51 v: Nigerian 52 59 59 61 61 sments of 66 68 76
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark The Benchr Box 2: Box 3: A Step-by-S Value Chain Case Study: Uga Tool 4: Upgrad Deepening Th	Increase Dustiness Environment Integrating Information Into Strateg Integrating Benchmarking and Gap Assessments of Value Chains ing Background Inarking Tool Uganda Benchmarking Constraints in the Coffee Industry Tanzania Cotton—Benchmarking Costs Step Summary of Tool 3: Conducting Benchmarking and Gap Assesses ns. Inda Floriculture: Benchmarking and Gap Analysis ding and Deepening the Value Chain rough Adding Operations	51 r: Nigerian 52 59 59 59 61 61 sments of 66 68 76 76
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark The Benchr Box 2: Box 3: A Step-by-S Value Chain Case Study: Uga Tool 4: Upgrad Deepening Th Box 4:	lerstanding the Value Chain and Integrating Information Into Strateg cting Benchmarking and Gap Assessments of Value Chains ing Background narking Tool Uganda Benchmarking Constraints in the Coffee Industry Tanzania Cotton—Benchmarking Costs Step Summary of Tool 3: Conducting Benchmarking and Gap Assess ns	51 r: Nigerian 52 59 59 59 61 61 sments of 66 68 76 76 77
Case Study: Und Domestic Catfish Tool 3: Condu Benchmark The Benchr Box 2: Box 3: A Step-by-S Value Chain Case Study: Uga Tool 4: Upgrad Deepening Th Box 4: Methods for D	In Dustness Environment Interstanding the Value Chain and Integrating Information Into Strateg Interstanding the Value Chain and Gap Assessments of Value Chains Ing Background Ing Background <tr< td=""><td> 51 v: Nigerian 52 59 59 61 61 sments of 66 68 76 76 77 78 </td></tr<>	 51 v: Nigerian 52 59 59 61 61 sments of 66 68 76 76 77 78

Conclusion	
Case Study: Ken	yan Green Beans and Other Fresh Vegetable Exports
Tool 5: Identif	ying Business Models for Replication
Box 6: Box 7: Chain	Replicable Business Models - Rwanda Coffee Washing Stations 85 Dairy Pakistan—Identifying and Replicating Business Models within the Value
A Step-by-Ste	p Summary of Tool 5: Identifying Business Models for Replication 87
Case Study: Ider	ntifying and Implementing Replicable Business Models-Mozambique Cashews
Tool 6: Captur	ring Value Through Forward and Backward Integration
Vertical Integ Consideration	ration from the National Value Chain Perspective
Box 8: Chain	Benefits of Vertical Integration—ZEGA and Zambia's Horticultural Value
Box 9:	Bulgaria Wine—Integrating Operations to Secure Sourcing of Raw Material
Case Study: Cap Skies Holdings.	turing Value Through Integration—The Ghanaian Pineapple Industry and Blue
Tool 7: Horizo	ontal Collaboration - Creating and Taking Advantage of Economies of Scale
A Step-by-Ste Advantage of	p Summary for Tool 7: Horizontal Collaboration - Creating and Taking Economies of Scale
Case Study: Crea d'Ivoire Experie	ating and Taking Advantage of Economies of Scale—The Ghana and Côte nces
Case Study: Crea Cashew Value C	ating and Taking Advantage of Economies of Scale Within the Mozambican Thain
Tool 8: Competitiveness	Positioning Products and Value Chains for Greater Value and 116
Box 10: Competit	Ecuadorian Cacao—Positioning the Value Chain for Greater Value and iveness
Box 11: Greater V	Thailand GAP Cluster—Positioning Products (and the Value Chain) for Value and Competitiveness
Leveraging O	ther Value Chain Initiatives 117

Product Positioning	118
A Step-by-Step Summary for Tool 9: Positioning Products and Value Chains for Value and Competitiveness	Greater 120
Case Study: Value Chain Strategies for Market Repositioning-Rwandan Coffee	121
Tool 9: Applying Standards and Certifications to Achieve Greater Quality	127
Vocabulary	127
Standards Are Part of Value Chain Strategy	128
Elements of Standards	129
Certification Bodies	129
Box 12: Thailand GAP Cluster—Use of Standards and Certifications to U	pgrade Value
	129
Importance of Implementing EurepGAP Through the Value Chain	130
Use of Certification and Branding for Local Markets	130
Information Channels	131
Management Oversight and Governance	131
Lead Firms	131
Associations and Cooperatives	132
Government Agencies	132
Box 13: Ecuador Cacao—Improving Quality at the Producer Level to Ach Market Prices	ieve Higher 132
In Summary	135
Case Study: Uganda Nile Perch Quality Management and Certification	135
Tool 10: Identifying Needed Support Services for the Value Chain	141
Box 14: Uganda Cotton—Enterprise-Linked Extension Services Model	141
Box 15: Sri Lankan Cinnamon	143
A Step-by-Step Summary for Tool 12: Identifying Needed Support Services for t Chain	the Value 145
Case Study: Identifying Needed Support Services for the Value Chain—Zambia Cotton	147
Tool 11: Improving the Operating Environment by Promoting Public/Private Dialo	ogue152
Enabling Environment	152
Public/Private Dialogue	152
- Box 16: Tanzanian Coffee and KII ICAEE: Productive Dublic/Driveto	
Dialogue	161

Case Study: Improving the Operating Environment Through Public/Private Dialogue- Cattle Producers Association	—Botswana 158
Tool 12: Achieving Synergies Through Clustering	. 164
Background on Clusters	164
The Relationship Between Value Chains and Clusters	165
Cluster Initiatives	165
Cluster Evaluation	. 167
Case Study: Achieving Synergies Through Clustering—Kenya Avocados	. 168
Tool 13: Monitoring Achievements in Value Chain Performance	. 181
Purpose of Monitoring	. 181
Implementing Monitoring in Value Chains	. 182
Box 17: Standards for Ghanaian Pineapples	. 183
Box 18: The PAID M&E Framework	. 185
Conclusion	. 179
SUMMARY AND CONCLUDING THOUGHTS	. 180
Bibliography	. 182
Acronyms	. 199
List of Text Boxes (in Order of Appearance)	. 202
List of Cases (in Order of Appearance)	. 203
List of Tables (in Order of Appearance)	. 204
List of Figures (in Order of Appearance)	. 205

Overview

This Guide to value chain approaches provides the user with actionable methods and tools to increase the productivity and performance of sub-Saharan African (SSA) agriculture.

The Guide is intended for use by a number of audiences. Most directly, it provides planners, decision makers, and implementers with practical tools for creating effective value and supply chain development programs. It also provides public and private stakeholders with a common framework for prioritizing decisions on sector and sub-sector competitiveness. Policy makers, business leaders, members of the development community, researchers, and practitioners can use these methods and approaches to promote the development of traditional and non-traditional value chains in sub-Saharan Africa.

Using concrete examples, mostly from African countries, this Guide presents, reviews, and systematically illustrates a range of concepts, analytical tools, and methodologies which, in turn, can be used to design, prepare, implement, assess, and evaluate agribusiness development initiatives. It presents and comments on various conceptual, methodological, and practical approaches to improving the competitiveness of agricultural supply and value chains. The Guide stresses the importance of value chain-based approaches and analysis for agro-enterprise and agro-food chain development in SSA.

Value chains are a key framework for understanding how a product moves from the producer to the customer. The value chain perspective provides an important means to understand the business-business relationships, mechanisms for increasing efficiency, and ways to enable business to increase productivity and add value. It provides a reference point for improvements in services and the business environment. It is a vehicle for pro-poor initiatives and for linking small businesses with the market.

Value chains reside at the core of high-impact and sustainable initiatives focused on improving productivity, competitiveness, entrepreneurship, and SME growth.

Africa's exports remain dominated by primary commodities, and the share of agriculture in SSA's total exports has declined sharply in the last 40 years. Only a few SSA countries have achieved significant diversification of their exports. Despite those trends, agriculture remains the main export-revenue source for many SSA countries and the largest income generator for their populations. Much, even most, African agricultural production is of low unit value and the result of low-productivity value chains. SSA's agriculture productivity measure of USD 343 value added per worker (2004) is the world's lowest - world agricultural productivity per worker averages three times that of SSA; Latin American productivity per worker averages nearly ten times the sub-Saharan African average.

Despite the successes of many African exporters in selling to new markets, without further improvements to their business environments and the competitiveness of their export commodities, many SSA countries risk being trapped by producing low-skill, low-value products and services, struggling to obtain a significant value-added share in global trade. The productivity and efficiency of agricultural value chains are basic to the success of SSA rural economies and to the incomes of SSA rural populations.

The Guide first examines core concepts and issues relating to value chains. A brief literature review then focuses on five themes, of particular relevance to African agricultural value chains, which can contribute to effective implementation tools and approaches:

- Trust and cooperation
- Governance
- Market power
- Innovation and knowledge
- Focus/intervention points.

These topics pertain to conditions and challenges faced by value chain participants and practitioners. They resonate throughout the many cases described in the Guide.

The bulk of the Guide presents methodological tools and approaches that incorporate important value chain concepts with the themes discussed and with sound business principles. The accompanying cases illustrate the application of the tools. The tools and case studies discussed in this Guide have been selected for their usefulness in directing and supporting market-driven, private-sector initiative and action.

This Guide offers 13 value chain implementation tools, presented within the implementation cycle of a value chain program.

- Designing strategies and business plans—Obtaining and using information
- Developing robust new businesses
- Supplying the market—Aligning supply to match market opportunity
- Reaching the market—Market positioning and market opportunities
- Improving the business and policy environment
- Monitoring results in value-chain development

Each tool is followed by descriptions of one or more actual cases. These cases illustrate the tool's application, and are coupled with embedded mini-cases for additional illumination. Roughly 60 percent of the examples are from Africa; others come from Europe, Latin America, and Asia. The Guide frequently returns to the same cases when describing different tools, which serves to highlight the multiple dimensions of a single value chain and the rich set of opportunities value chain initiatives can offer.

Section 1: Introduction

Agricultural Value Chains and Africa's Development Agenda

The development and business communities involved in the African agriculture and agribusiness sectors have recently experienced a tremendous resurgence of interest in promoting value chains as a way to add value, diversify rural economies, and contribute to increasing rural household incomes in most sub-Saharan Africa (SSA) countries. Value chains are increasingly recognized as a means to reduce the rural poverty prevalent in the region. For practitioners who have long been convinced of the need to look differently at agriculture – not just as a means of survival but as smaller or larger commercial businesses linked to domestic and global markets – and of the need to identify and tap into new sources of potential growth and value addition in the sector, this is a welcome development. Hopefully, renewed engagement will lead to a substantial increase in the flow of financial resources and assistance that is dedicated to supporting market-sustainable agro-enterprises and agricultural value chains throughout the African continent.

There is danger that this renewed enthusiasm may not last, or may even backfire if the high expectations for this approach are not met. In view of development literature's current lack of clarity about the concepts and methods relating to value chains, there is risk that sooner or later the benefits of the value chain approach will be overshadowed by unmet expectations and the concept may be categorically discarded. An absence of conceptual precision could lead governments and donors to once again move away from these promising areas for economic development that are especially relevant for sub-Saharan Africa.

There is no single way to mitigate such risks, but this Guide offers useful guidance to the businessmen, policy-makers, representatives of farmer or trade organizations, and others who are engaged in SSA agro-enterprise and agribusiness development. The Guide is particularly designed for those who want to know more about value chain-based approaches and how to use them in ways that can contribute to sound operational decisions and results for business development and policy-making.

Using concrete examples, mostly from African countries, this Guide presents, reviews, and systematically illustrates a range of concepts, analytical tools, and methodologies centered on the value chain that can be used to design, prepare, implement, assess, and evaluate agricultural and agribusiness development initiatives. It presents and comments on various conceptual, methodological, and practical approaches that focus on improving the competitiveness of agricultural supply and value chains. In doing so, the Guide stresses the importance of value chain-based approaches and analysis for agro-enterprise and agro-food chain development in SSA.

The tools and case studies discussed in this Guide have been selected for their usefulness in directing and supporting market-driven, private-sector initiative and action. While the Guide is designed to speak directly to the needs of the businesses and direct actors in the value chains, it has also been prepared to serve as a resource for those practitioners, planners, and program

implementers who work closely with value chain participants to improve the productivity of Africa's agriculture.

Opportunities and Threats of Trade and Market Globalization for African Economies

Fundamental changes in international commerce and finance, including reduced transport costs, advances in telecommunications technology and lower policy barriers, have fueled a rapid increase in global integration. The international flows of goods and services, capital, technology, ideas, and people offer great opportunities for African nations to boost growth and reduce poverty by stimulating productivity and efficiency, providing access to new markets and ideas, and expanding the range of consumer choice. At the same time, globalization creates new challenges, including the need to increase the quality and sophistication of African goods and services, for new regulatory reforms to take full advantage of international markets, and for cost-effective approaches to cope with the resulting adjustment costs and regional imbalances.¹

Since 1970, trade in sub-Saharan Africa (SSA) has grown at three-quarters of the world's rate and only about one-half of Asia's rate.

Africa's share in world trade has actually fallen from 4 percent in the 1970s to 2 percent in 2005 (see Figure 1). One of the most striking phenomena is the gradual marginalization of sub-Saharan Africa in international agricultural-export markets. While sub-Saharan Africa is host to 12% of the world's arable land, the region's share of global agricultural exports has declined gradually from almost 10 percent four decades ago to around 3 percent today.³ On the import side, the



Figure 1: Africa's Share of World Trade²

Source: IMF, Direction of Trade Statistics

opposite pattern emerges as sub-Saharan Africa is the only developing-country region that has seen its share of world agricultural imports increase rather than decrease.⁴

These patterns are manifest in measurements of Africa's trade openness⁵ (measured by the trade-to-GDP ratio), which has also grown more slowly than that of any other major developing region and has supplanted Latin America's place as the region least open to trade.⁶

¹ Bolnick B., Camoens A., & Zislin J. (of Nathan Associates, Inc.) "Tanzania Economic Performance Assessment," USAID Country Analytical Support Report, 2005. p. 14.

² Gupta, S. & Yang, Y. sic.

 ³ Food and Agriculture Organization of the United Nations (FAO). 2006. FAOSTAT Online Statistical Service. Rome: FAO. Available online at: <u>http://faostat.fao.org</u>.
 ⁴ The State of Food and Agriculture 2005: Agricultural trade and poverty: Can trade work for the poor? Series title:

⁴ The State of Food and Agriculture 2005: Agricultural trade and poverty: Can trade work for the poor? Series title: FAO Agriculture Series (SOFA)-36, 2005.

⁵ The trade ratio is widely used as a measure of integration into the world economy. Caution is needed, however, in interpreting it as an indicator of policy openness or competitiveness, because countries that are large or distant from major markets tend to have low ratios regardless of whether their policy regime is open. Similarly, small countries

However, the trends are not all negative. In fact, several SSA countries have recently improved their standing in terms of trade openness by taking advantage of new export opportunities. As a region, SSA's average trade openness in 2005, as measured by exports as a percentage of GDP, was 39 percent, an increase of 18 percent from 2002 levels (see Table 1). Within Africa, levels of trade openness fluctuate depending on geographic location, resource endowment, infrastructure quality, enabling environment and other pertinent issues.

Those blocks of countries showing notable increases in trade openness are non-oil producing countries (36 percent change since 2002); SSA excluding South Africa and Nigeria (33 percent change since 2002); and the countries of the Economic and Monetary Community of Central Africa (29 percent). Africa's oil-producing countries have measured a 61 percent increase in exports since 2002, but when compared to 1997-2002 levels, this is an 8 percent drop, perhaps indicating that recent growth is largely due to volatility in world oil markets (especially the drop in trade during 2002). Relative to other parts of Africa, recent trends of trade openness show poor performance by the West African Monetary Union (-6 percent) and the Common Market of Eastern and Central Africa (6 percent).

	Historical					Change: Historical	Change
	1997-2001	2002	2003	2004	2005	2005	2002-2005
Sub-Saharan Africa (SSA)	31.5	32.6	33.8	35.9	38.5	22%	18%
SSA excluding S. Africa & Nigeria	32.6	30.5	34.2	38.2	40.7	25%	33%
CFA Franc Zone	34.7	37.7	36.1	39.8	43.2	24%	15%
West African Economic Monetary Union	29.5	33.3	30.5	30.6	31.3	6%	-6%
Economic and Monetary Community of Central Africa	41.4	43.2	43.2	50.4	55.7	35%	29%
Common Market of Eastern and Southern Africa	30.4	31.6	31.4	31.8	33.5	10%	6%
Oil-producing Countries	45.7	26.1	33.9	39.6	42.1	-8%	61%
Non-oil producing Countries	26.7	46.7	50.4	56.6	63.4	137%	36%
C IME 2005							

Table 1: African Trade Growth – Export of Goods and Services (% of GDP)

Source: IMF 2005

Still, Africa's exports remain dominated by primary commodities, with fuels accounting for about 40 percent and agricultural products approximately 20 percent. SSA has seen a sharp decline in the share of agriculture in its total exports, from more than 60 percent four decades ago to around 20 percent today.⁷ Only a few countries, such as Zambia and Kenya, have achieved significant diversification of their exports, and the share of manufactured goods in Africa's total exports has stagnated at about 30 percent, well below that of other developing regions. In addition, manufactured exports from African countries have a narrow base and low value-added; often, they are semi-processed, raw materials, or products that have preferential access to industrial countries.

Perhaps indicating future prosperity, African trade has recently begun to expand to newer markets of Asia. In general, the high commodity prices of recent years and Asia's economic

typically have a high trade ratio even with protectionist policies. Oil exporters also may have high trade ratios despite restrictive policy regimes.

⁶ Gupta, S. & Yang, Y. "Unblocking Trade: To underpin growth, African needs to adopt a comprehensive approach to boosting trade." Finance and Development Magazine, December 2006, Volume 43, Number 4.

⁷ The State of Food and Agriculture 2005: Agricultural trade and poverty: Can trade work for the poor? Series title: FAO Agriculture Series (SOFA)-36, 2005.

growth have helped Africa expand its exports to Asia (which now imports 25 percent of all African exports). The growing economies of India and China alone account for 10 percent of Africa's exports. As these countries continue to grow and demand more natural resources, African exporters may be poised to increase revenues and expand their production in concert with the Indian and Chinese economic growth.

Even so, without further improvements to their business environments and the competitiveness of their export commodities, many SSA countries risk being trapped by producing low-skill, low-value products and services, struggling to obtain a significant value-added share in global trade. Without market knowledge, particular expertise, or competitive products or services, entire economies will essentially fail to take advantage of the potentially high benefits of global markets and the increases in global trade flows. SSA economies unable to claim a more significant share of global trade will find it difficult to achieve the sustainable and accelerated growth rates that are necessary to reach the Millennium Development Goals (MDGs) and significantly reduce poverty on the continent.

Table 2: SSA's Rank Among WorldImporting and Exporting Regions

		Destination
Exporting		
region	Year	Sub- Saharan Africa
World	1980	3
	1985	3
	1990	2
	1995	2
	2000	2
	2002	2
Importing		Origin
Importing region	Year	Origin Sub- Saharan Africa
lmporting region World	Year 1980	Origin Sub- Saharan Africa 6
lmporting region World	Year 1980 1985	Origin Sub- Saharan Africa 6 5
Importing region World	Year 1980 1985 1990	Origin Sub- Saharan Africa 6 5 4
Importing region World	Year 1980 1985 1990 1995	Origin Sub- Saharan Africa 6 5 4 4 4
Importing region World	Year 1980 1985 1990 1995 2000	Origin Sub- Saharan Africa 6 5 4 4 4 3

These threats and opportunities hold particularly true for agriculture, the main export-revenue source for many SSA countries and the largest income generator for their populations. Increasing production of, and export revenues from, agricultural goods entails developing marketing channels and outlets. Such development is essential to national strategies in order to increase incomes and eradicate poverty in SSA. Increased productivity in terms of value and profitability are clearly the way to generate increasing incomes in a sustainable manner—that is, without further depleting SSA's natural resource base.

The majority of sub-Saharan Africans are low-income, and often subsistence, farmers. Sixty-five percent of Africans in SSA live in rural areas,⁸ while 70 percent of the SSA labor force works in agriculture. Sub-Saharan Africa's share of the world's agricultural exports is approximately 2 percent, and imports measure approximately 3 percent of world trade.

Source: J.E. Austin Associates, Inc

Comparison of Agricultural Productivity

One way to increase the competitiveness of an industry or product on the global market is to produce more efficiently. Increases in efficiency are captured by measuring the agriculture value added per worker, which is also a proxy for agricultural productivity.⁹ For African producers to

⁸ World Development Indicators 2007, World Bank.

⁹ Value added in agriculture measures the output of the agricultural sector less the value of intermediate inputs. Agriculture comprises value-added from forestry, hunting and fishing as well as crop cultivation and livestock production. Data are in constant 2000 U.S. dollars.

capture more value and increase exports, they must increase productivity levels. SSA's agriculture productivity measure of USD 343 value added per worker (2004) is the world's lowest. In comparison, world agricultural productivity averages three times SSA at USD 919, and Latin America is nearly 10 times more productive at USD 3,183 per worker.

The sequence of steps and actors involved in the process from production to delivery of a product to the market is called a value chain (sometimes also referred to as supply chains or market chains). The productivity and efficiency of agricultural value chains are thus basic to the success of SSA rural economies and to the incomes of SSA rural populations.

Value chains compete globally, and African agriculture competes in international and domestic markets with the exporters and products of Asia, Europe, and the Americas. The table below shows average agricultural productivity across the continents. Much, even most, of African agricultural production is of low unit value and the result of low-productivity value chains.

							Change:	Change:
COUNTRY	1990	2000	2001	2002	2003	2004	1990-2004	2000-2004
East Asia & Pacific	295.8	389.7	400.1	411.2	424.5	446.1	51%	14%
Europe & Central Asia	1922.1	1767.5	1880.0	1957.7	2000.5	2173.9	13%	23%
Latin America & Caribbean	2177.9	2725.1	2829.3	2906.7	3036.4	3183.7	46%	17%
Middle East & North Africa	1525.9	1761.2	1833.0	1909.7	2014.2	2049.5	34%	16%
South Asia	339.4	389.3	401.9	375.6	401.6	399.5	18%	3%
Sub-Saharan Africa	317.2	331.1	339.5	342.1	329.2	343.9	8%	4%
World	748.6	863.9	867.3	868.3	889.8	918.9	23%	6%

Table 3: Historical Levels of Agriculture Productivity, by Region

Source: World Development Indicators 2007, World Bank

Such low levels of productivity hinder Africa's attempts at reducing poverty. SSA's agricultural value chains need to become more productive and competitive in the global market for agricultural goods, and its value chains need to achieve greater value within Africa, as well. Increases in competitiveness can assist those dependent on agriculture and agribusiness in increasing their incomes and wealth.

Within Africa, there are large discrepancies between different countries' average levels of productivity. Countries such as Tanzania, Mozambique, the Congo Republic, Guinea, Mali, Burkina Faso, Central African Republic, and Rwanda have shown sustained growth over the past 17 years. Others, such as Burundi, the Democratic Republic of the Congo, Madagascar, and Zimbabwe, have suffered significant decreases in agricultural productivity, which may be the result of insecurity and conflict.

				Change:	Change:
COUNTRY	1990	2000	2004	1990-2004	2000-2004
Angola	208.0	126.5	193.5	-7%	53%
Benin	350.1	541.6	622.0	78%	15%
Botswana	577.3	398.9	429.6	-26%	8%
Burkina Faso	127.8	168.8	188.9	48%	12%
Burundi	108.6	86.5	73.7	-32%	-15%
Cameroon	731.0	1024.3	1179.8	61%	15%
Cape Verde	1325.8	1593.5			
Central African Republic	288.5	376.4	421.8	46%	12%
Chad	141.0	208.0	214.0	52%	3%
Comoros	407.3	401.8	419.0	3%	4%
Congo, Dem. Rep.	185.8	164.5	150.6	-19%	-8%
Congo, Rep.	330.3	299.6	367.6	11%	23%
Cote d'Ivoire	605.6	804.6	835.7	38%	4%
Equatorial Guinea	478.4	706.5			
Eritrea		63.7	59.4		-7%
Ethiopia		148.9	154.2		4%
Gabon	1567.4	1513.3	1929.1	23%	27%
Gambia, The	231.4	254.2	239.8	4%	-6%
Ghana	301.3	320.9	354.2	18%	10%
Guinea	171.2	206.9	236.6	38%	14%
Guinea-Bissau	201.7	228.7	236.9	17%	4%
Kenya	352.2	306.4	326.2	-7%	6%
Lesotho	557.5	518.0	493.9	-11%	-5%
Madagascar	186.9	178.9	174.7	-7%	-2%
Malawi	73.6	135.9	136.7	86%	1%
Mali	205.3	205.5	229.3	12%	12%
Mauritania	554.5	441.6	337.7	-39%	-24%
Mauritius	3802.7	3774.8	4966.7	31%	32%
Mozambique	121.5	116.9	157.6	30%	35%
Namibia	786.7	1078.6	1099.4	40%	2%
Niger	165.6	156.4			
Nigeria	578.1	774.2	949.4	64%	23%
Rwanda	172.4	201.0	221.5	29%	10%
Sao Tome and Principe	181.4	216.0	220.8	22%	2%
Senegal	263.5	275.7	258.9	-2%	-6%
Seychelles	618.7	579.9	511.8	-17%	-12%
South Africa	1911.8	2270.7	2498.5	31%	10%
Sub-Saharan Africa	317.2	331.1	343.9	8%	4%
Sudan	328.6	658.5			
Swaziland	1296.3	1237.2	1211.8	-7%	-2%
Tanzania	245.8	264.9	302.6	23%	14%
Togo	360.6	395.1	411.8	14%	4%
Uganda	188.1	221.9	237.2	26%	7%
Zambia	178.3	211.2	219.3	23%	4%
Zimbabwe	267.7	325.8	235.8	-12%	-28%

Source: FAO STAT

Table 4: Historical Trends of African Productivity, by Country

Objectives of the Guide

In the above context, the World Bank has been giving increasing emphasis in recent years to issues of agricultural trade facilitation, export promotion, and competitiveness. The Bank seeks to transform African agriculture into a profitable business by focusing on increased productivity and by linking farmers with more lucrative markets. Under the recently approved African Action Plan, the World Bank/ESSD Africa has further engaged in generating and disseminating

knowledge concerning the potential development of high-value agricultural supply chains as a means to increase and diversify revenues in rural areas of SSA.

This Guide to value chain approaches is part of that effort, providing the user with actionable methods and tools based on value chain concepts that can increase the productivity and performance of SSA agriculture. Contrary to earlier thinking and practice – which tended to focus on development analysis, project appraisals, and classical economic analysis such as benefit/cost analysis – experience shows that business concepts and methods are also vital to developing value chains in an African context. A wide range of tools and approaches – traditionally considered to be the purview of private business outside of the economic development context – are, in fact, proving to be most useful when it comes to designing subsector, commodity, and value chain strategies, as well as helping private businesses and professional organizations develop their own strategies and implement their own business plans.

Worldwide, there has been a great deal of experimentation and learning in the search for reliable methods to achieve sustainable economic growth. Recognizing that this growth must be led by sound business strategy and operation and driven by market demand, practitioners have gradually abandoned early, state-focused models (i.e. government ownership of resources and enterprise) and have since recognized that both public and private sectors have vital roles to play in the economic-growth partnership. How well those respective public and private roles are defined, and how competently each sector performs its role, are significant determinants of economic growth.

For many years, private-sector development initiatives and programs have emphasized actions to increase businesses' access to training, skills development, financing, entrepreneurship, business services, and other important elements. More contemporary focuses have included helping small and medium enterprises (SMEs) link with global markets and improving economies' business environments. While these varied initiatives have all proven useful, each by itself is unlikely to generate significant changes in a country's economic performance.

In recent years, a great deal of useful work has focused on deal making—helping businesses identify new markets and make sales. This focus is also undoubtedly important but is often driven by donor and government impatience to achieve visible results.¹⁰ While it is difficult to argue with the resulting increased sales and investment, there also remains a recognizable need for long-term planning, implementation, and sustainable results.¹¹. It is only more recently that practitioners have emphasized the sound business strategies of refocusing and improving value and market chains.

This Guide describes individual tools, approaches, and case studies. To introduce these, the Guide also discusses several key themes that the tools embody or take into account. In this regard, however, the important "take-away" from this document is not the identification of

¹⁰ KITS.

¹¹ For a recent discussion of sustainability in supplier-buyer relationships, see "*The Ties that Bind: Making Buyer-Supplier Relationships Stick*". (USAID's Business Growth Initiative, J.E. Austin Associates, Inc., https://www.businessgrowthinitiative.org/ResourceCenter/Pages/TechnicalBriefs.aspx

common themes, but rather the insights and practical applications that value chain-based approaches provide for adding value, creating opportunity, and enabling SSA's agricultural populations and businesses to reach markets.

The Guide is intended for use by a number of audiences. Most directly, it provides planners, decision makers, and implementers with practical tools for implementing effective value- and supply-chain development programs. It also provides public and private stakeholders with a common framework for prioritizing decisions that will have a competitive impact on sectors and sub-sectors. It provides policy-makers, business leaders, members of the development community, researchers, and practitioners with methods and approaches that can be used to promote the development of traditional and non-traditional value chains in sub-Saharan Africa.

The Guide's primary content – its themes, tools, and cases – has been chosen to provide a selection of insights and approaches. However, it is not comprehensive and there are certainly many other aspects of value-chain implementation and strengthening that are worthy of exploration. Perhaps they will be the focus of subsequent work.

The literature concerning value chains, both academic and applied, is very substantial (see Section 3). Special mention must be made of the considerable body of work that is being developed by and with the sponsorship of international agencies. The U.S. Agency for International Development (USAID) has developed many useful materials centered on value chain analysis, and its work is well reflected in its modular value chain training program and by its ongoing value-chain work. Germany's Gesellschaft für Technische Zusammenarbeit (GTZ) has also pioneered excellent work on value chains, as has the UK's Department for International Development (DFID) and several other agencies. The World Bank and International Finance Corporation (IFC) are preparing other toolkits and guides with complementary purposes and focus areas.

Organization of the Guide

The Guide is organized as follows:

- Section 1 introduces the Guide, its scope, and its objectives.
- Section 2 includes the definition of value chains, a description of their structure, and a background on using and analyzing value chains.
- Section 3 reviews existing literature on value/supply chains, including current theories and applications.
- Section 4 discusses individual tools used in value chain analysis.
- The final section, Summary and Concluding Thoughts, summarizes the Guide's principles and lessons and suggests several future steps for achieving further experience in value chain competitiveness.

Section 2: Concepts and Definitions of Value Chains/Supply Chains

Value Chains: Definition of Concepts

Chains composed of companies (or individuals) that interact to supply goods and services are varyingly referred to as production chains, value chains, filières, marketing chains, supply chains, or distribution chains. These concepts vary mainly in their focus, in the activity that is emphasized, and in the way in which they have been applied. However, they all describe the interactions of firms and processes that are needed to deliver products to end users, and they all aim to identify opportunities for and constraints against increasing productivity. Although it is impossible to make fine distinctions among these often-overlapping concepts, it is still worthwhile to provide some basic definitions.

For example, a value chain describes the full range of activities required to bring a product or service through the different phases of production, including physical transformation, the input of various producer services, and response to consumer demand.¹² As such, value chains include the vertically linked interdependent processes that generate value for the consumer. In contrast, the term supply chain is used internationally to encompass every activity involved in producing and delivering a final product or service, from the supplier's supplier to the customer's customer.¹³

The primary focus of supply chains is thus on cost and efficiencies in supply, while value chains focus more on value creation, innovation, product development, and marketing. While both concepts describe the same network of companies that interact to deliver goods and services, the value chain is essentially about value. The issue is not so much about which approach is superior or preferable, since both can deliver improved business performance and productivity gains for the chain's participants.¹⁴ It must be noted, though, that practitioners often focus on reducing costs and marginal inefficiencies in supply at the expense of focusing on interventions that could lead to bigger additions of value. (It is also worthwhile to note that recent years have seen a great deal of emphasis and innovation in the logistics of supply chains.)

Similarly, clusters are composed of collections of firms and institutions that perform many of the functions segmented and described in value chains. They describe both horizontal and vertical links between the various businesses and other organizations that are instrumental in producing a product (or closely related products) or service. Thus, they often incorporate elements of many value chains. The literature on clusters stresses the benefits of enterprise agglomeration and geographic proximity, placing relatively more emphasis on the local environment (both policies and institutions, public and private) and context in which it operates. In the "chain" concept, emphasis on the local environment is secondary, at best. Value chain literature tends to have less emphasis on external relations, while cluster analysis often omits the distribution of value generation among links in a chain.

¹² Kaplinsky, Raphael and Michael Morris. "A Handbook for Value Chain Research," September 2000.

¹³ From Feller, Andrew et al., "Value Chains versus Supply Chains", BPTrends, March 2006.

¹⁴ In fact, Feller et al. argue precisely for the need to stop thinking of supply chains and value chains as different entities, but rather, for integration of the two.

Another concept related to the value chain is the Francophone *filière* (literally "thread" in English). The filière is used to describe the flow of physical inputs and services in the production of a final product, and is essentially similar to the modern value chain concept.¹⁵ Filière studies do not have a single unifying theoretical framework, and its practitioners have borrowed from different theories and methodologies for their analyses. The filière was initially used to study contract farming and vertical integration in French agriculture in the 1960s. It was, soon thereafter, applied to agriculture in developing countries (dealing mainly with local agriculture production systems and consumption). Over time, filière analysis focused more on how public institutions affect local production systems. The field has recently begun to deal more directly with issues of trade and marketing in order to discuss the workings of commodity chains within an increasing liberalized context.

All of the concepts discussed, whether chain, cluster, or filière, underscore the importance of linkages to gain value and advantages to compete in global markets. The term value chain is the one primarily used in this Guide, as it is inclusive and incorporates supply, value addition, transactions, and market linkages. We use other terms occasionally (particularly supply chain) where we believe that the Guide's conceptual focus or a specific case warrants.

How Value Chain Analysis Has Been Used

Interest in value chains is not new. Businesses have been using value chain analysis and implementation principles for years to formulate and implement competitive strategies. Corporations use value chain analysis to answer questions such as, "Where in the value chain should my business be positioned to improve its performance?" The value chain's popularity has been reinforced by many important business strategy themes, including core competencies, comparative and competitive advantage, outsourcing, vertical and horizontal integration, and best practices.

Businesses (individually and in groups, such as clusters) have focused on value chains while searching for alternative ways to remain competitive. Value chain approaches have been used to streamline processes that generate the goods and services that customers value and to guide product improvement and innovation. Further, there is increased awareness that procedures within a firm might not affect its own competitiveness unless other firms adopt similar or linked practices. This mode of thinking has paved the way for businesses to use collaborative value chain concepts to identify efficiencies and competitiveness both within and among firms, acting on opportunities to build win-win linkages and collaboration. Recent technological developments that permit high levels of information sharing have reinforced businesses' realization of value chain efficiencies, especially in supply management.

More recently, governments and donors, realizing that upgrading the performance of individual firms may have little impact, have shown significant interest in value chain analysis and implementation. In their effort to devise interventions that reposition entire industries, build business competitiveness, and spur economic growth, governments and donors can use value chain-based approaches as robust tools to protect threatened links, facilitate upgrading of others

¹⁵ Kaplinsky, "A Handbook for Value Chain Research," p. 7.

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

to generate greater returns, and to promote Foreign Direct Investment (FDI) programs. Additionally, value chain analysis has been used to examine constraints in the enabling environment in which the chains operate.

Value chains have also been used as a tool for SME development, with new methods of linking SME suppliers and service providers to the value chains of lead processors or marketers. Value chain analysis sheds light on the size of the firms participating in each link, how they are participating or could be participating in the chain, and opportunities to facilitate or improve those linkages.

Analyzing and Evaluating Value Chains

Value chain analysis focuses on segmenting the different activities that add value in the production and sale of a product or service. It differs from traditional industry sector analyses in many ways, including the following:¹⁶

- Identifying dynamic linkages between productive activities, value chain analysis transcends traditional economic and industry sectors by showing where value is added in a production process.
- Value chain analysis focuses on net value added instead of overall size and gross output.
- Traditional industry sector analyses often do not sufficiently determine the distribution of value added between activities, both within and between countries, or a country's current insertion in local markets.
- Value chain analysis thoroughly examines information flows among actors in the value chain unlike typical industry analysis.
- Segmenting the value chain allows for better understanding of the constraints and opportunities within each segment, as well as the context in which the chain operates.

There are many ways to analyze or evaluate a value chain. Analysis can stem from research of secondary information, such as government or industry data, to interviews with industry participants. It can also be derived from participatory market assessments and market observations. Once the information is gathered, numerous tools and processes help interpret and inform the resulting analysis.

In general, an in-depth value chain analysis considers the following:¹⁷

- What are the economic costs along the value chain?
- Where is the most value added to the value chain?
- Who are the most import actors within the value chain?
- What is the institutional framework of the value chain?
- Where are the bottlenecks in the value chain?

¹⁶ Some of these differences were adapted from Kaplinsky and Morris, "A Handbook for Value Chain Research", IDRC, 2000, p. 46-47.

⁷ SNV, "Synthesis of the 1st cycle of CVRD debate", Dec. 2004.

- Where is there market potential for growth?
- What is the size of the sector/chain?
- What is the potential for upgrading?
- What possible synergies exist?

Limitations of Value Chain Analysis

As mentioned, there are many ways of analyzing a value chain. For example, value creation can be disaggregated between each link in the chain, as well as within each link. Some chains are merely a directional map (such as the one in Figure 5), which is, in itself, valuable for beginning to understand the actors and processes that intervene to create value for particular consumers. However, agencies and other sponsors that commission value chain analysis often find that the analysis as implemented is insufficient and cannot be used to guide them in making informed decisions—particularly in deciding on actions that will greatly impact value added, rather than merely reducing costs.

Indeed, many of these analyses have a common weakness: the tendency to focus excessively on cost efficiency or breakouts of cost components. While efficiency in production is increasingly becoming a necessary condition for penetrating global markets, it will not ultimately determine sustained participation and increased incomes for value chain participants. The following are examples of some related analytical weaknesses and challenges:

a. Value chains are not fixed or static

It is important to recognize that value chains are not fixed in terms of composition, relationships, or market positioning, and that there is a competitive need to alter and improve the value chain in light of strategic choices that businesses can make regarding the markets in which they compete. While a value chain's purpose is to link production to the target market advantageously, it is the private sector that decides which markets and where to compete – and alters the value chain accordingly. Value chain analysis too often focuses simply on improvements within the given value chain, rather than on how value chains can be shifted to target different, more attractive markets and business strategies.

b. Market dynamics matter

Value chains can be helpful instruments for serving the needs of a particular market sector, but focusing on a static value chain can also mask the need to segment and customize products for different markets. The key elements of building sustainable competitiveness are a solid understanding of market dynamics and a thorough analysis of the attractiveness of potential market segments and the competition. Businesses must choose which products and which markets can be served competitively and base their goals and strategy on good market analysis.

c. Quality and service are also important

Similarly, excessive focus on delivering a product (especially a commodity) may hide opportunities to deliver a package of products and services that the market or customer will find

desirable. Too often, a value chain analysis is not designed to help businesses and planners weigh choices about delivering product quality, information, and service.

d. Considering the environment in which a value chain operates

Often, value chain analysts fail to properly consider the business environment in which the value chain operates. In doing so, the analysis can fail to identify potential interventions for improved business and value chain performance. Government regulations, international standards and agreements, and market forces typically shape the business environment. Michael Porter's Diamond for describing competitiveness, shown in the accompanying figure, is a useful framework for considering the factors that determine a value chain's business environment. (The Diamond is discussed further in Section Four, Tool 2.)



Figure 2: Competitiveness Diamond

e. A simple cost analysis will not do

Certain value chain analyses merely depict a cost build-up per activity in producing a value chain without mapping the actors involved or identifying the value that is captured at each link of the chain.

The cost build-up, and benchmarking it against competitors, will obviously provide ideas on areas for improvement. But the analysis will probably not shed light on which activities generate more value, whether the product can be produced at a competitive price for other markets, how well the chain is integrated, or how easily information flows throughout it. More importantly, a

simple cost build-up will tend to focus on interventions that improve on costs, rather than on repositioning the whole chain into more lucrative markets and products.

f. Shifting value within a value chain, rather than creating more value

As mentioned earlier in this section, donor agencies and governments have sometimes used value chain analysis to identify and protect threatened links along chains. Additionally, some stakeholders continue to look at value chain analysis as a zero-sum game focused on shifting value from one link of the chain to another. This cutthroat perspective obscures opportunities to upgrade the whole system to the benefit of all value chain participants.

Implications of the Operating Environment: Beyond the Value Chain

While value chain analysis is extremely useful, its weaknesses highlight the fact that many other important considerations are necessary to increase value for the entire chain or some of its participants. Beyond the value chain itself, it is important to understand market dynamics, competitive forces, and the operational environment.

Overall, there is a need to focus on fundamentals. Every market segment has key elements and drivers that are important for competing successfully. Interventions to build competitiveness should not attempt to make quick, comprehensive improvements throughout a value chain. Rather, a competitiveness-building strategy targets priority elements for improvement. In other words, improvements should not be made irrespective of a strategy that has set goals and objectives and is based on identified opportunities, given the value chain's relative position. It is crucial to identify success factors and driving forces in the target markets in question and subsequently benchmark these elements against top performers and competitors.

In sum, value chain interventions should aim to sell a higher-value product. They should not be predicated only on "productivist"-based assumptions that can lead to reductions in production costs and increases in quantities produced, but also those that competitors can and will easily imitate.

Section 3: Review of Existing Literature on Value/Supply Chains

This section, through a review of value chain literature, provides a summary of and commentary on the state of knowledge and available information on several main themes currently being discussed by researchers and practitioners regarding how value chain-based implementation is framed, focused, and realized. The literature review emphasizes key themes from value chain development and identifies the bases for appropriate tools used to guide decision-making and action in the agricultural sector. While not specific to Africa, the themes are highly applicable to African agricultural value chains and are woven throughout the tools and cases presented in Section 4.

This section highlights a number of issues, patterns, and themes; the discussion of their contexts and relevance establishes a good groundwork for understanding the implementation tools that follow. Although the documents reviewed here indicate the remarkable breadth of work that has been done on value chain development, they are only a small portion of the available literature. This section is not intended to comprehensively review the entirety of value chain literature; the reference documents presented and cited here were selected for their coverage of the themes most relevant to the tools and approaches presented in this Guide.

The Guide's bibliography includes a comprehensive list of the resources consulted. Those documents specifically mentioned below are highlighted for their usefulness in discussing themes, models, theoretical prescriptions, lessons learned from past program implementations, and illustrative anecdotes as their main message.

For information specifically emphasizing sub-Saharan African agricultural value chains, we turned most frequently to commodity-specific or program-implementation reports. These reports were useful for African perspectives and for understanding key geographic aspects of value chain program implementation. Some of the most pertinent reports that provided African perspectives were:

- "Successes and Challenges in Promoting Africa's Horticultural Exports" by Gabre-Madhin & Minot, World Bank.
- "Partnerships for Agribusiness Development, Agricultural Trade, and Market Access A concept note for NEPAD" by TechnoServe, November 2004.
- "The market for non-traditional agricultural exports" by Hallam, Liu, Lavers, Pilkauskas, Rapsomanikis & Claro, Commodities and Trade Division, FAO, 2005
- "Etude sur la Compétitivité des Filières Agricoles dans l'Espace", Union Economique et Monétaire Ouest Africain (UEMOA).
- "Exporting Out of Africa: The Kenya Horticulture Success Story" by Jaffee & Okello, World Bank.
- "Globalization of the Agro-food System: Success and Challenges for Promoting Africa's Horticultural Exports" by Gabre-Madhin & Minot, International Food Policy Research Institute.

- "Guide to Commodity-based Export Diversification and Competitiveness Strategies for African Countries" by Stryker & Salinger, Associates for International Resource Development.
- "High Value Agricultural Products for Smallholder Markets in Sub-Saharan Africa: Trends, Opportunities and Research Priorities" by Temu & Temu, ICTA
- Programme de Developpement des Marches Agricoles AgMarkets Sénégal, GEOMAR International Inc.

This literature review focuses on five themes, currently the focus of much consideration, that are of particular relevance to African agricultural value chains and that can contribute to effective implementation tools and approaches:

- Trust and cooperation
- Governance
- Market power
- Innovation and knowledge
- Focus/intervention points

These topics pertain to conditions and challenges faced by value chain participants and practitioners and can help answer questions such as:

- Why do certain firms find it in their interest to cooperate or develop win-win relationships?
- Why do certain firms have the most market power and the ability to determine price to their advantage?
- What is a strategy for maintaining market share?
- How does a supply or value chain get pulled by market demand?
- How is innovation allowed to flow among members of a chain?

These common themes were extracted or developed from various works. A partial list (and their emphasis) is shown in Table 5. We particularly note:

- A Handbook for Value Chain Research, Kaplinksy & Morris, International Development Research Center.
- Global Commodity chain analysis and the French filière approach: comparison and critique, Raikes, Jensen, Ponte Royal Dutch Veterinarian & Agricultural University
- Globalization and the Small Firm: A Value Chain Approach to Economic Growth and Poverty Reduction, Downing, Field, Kula, United States Agency for International Development.
- Governance in Global Value Chains, Humphrey & Schmitz, Institute of Development Studies at the University of Sussex.
- Implementing Sustainable Private Sector Development: Striving for Tangible Results for the Poor: The 2006 Reader, Miehlbradt, McVay, Tanburn, International Labor Organization of the United Nations.

- Info-Cadena: Instruments to Foster Value Chains, Springer-Heinze, German Agency for Technical Cooperation (GTZ).
- Integrating SMEs in Global Value Chains: Towards Partnership for Development, Kapinsky, Readman, United National Industrial Development Organization.
- AMAP BDS Knowledge and Practice Task Order: Lexicon General, Dunn, United States Agency for International Development.
- Commodity Chains and Global Capitalism, Gereffi, Korzeniewicz, 1994.

In Table 5, each document's technical focus has been distinguished according to focus on private-sector development (PSD), value chain (VC) and agribusiness. The geographic focus of each document has been split indicated: documents that focus on Africa and those that focus on general best practices or methodologies.

Тί	able	5:	Key	Documents	U	tilized	in	the	L	iterature	Rev	view

				Relevant V	alue Chain Th	neme Covered			
Value Chain or Agribusiness Resource Document	Spatial Focus	Technical Focus	Trust	Governa nce	Market Power	Innovation & Knowledge	Interven tion	_Author (s)	Affiliated or Donor Organization
Addressing marketing and processing constraints that inhibit agrifood exports: A guide for policy analysis and planners	General	Agribusiness		\checkmark	\checkmark			Westlake	FAO
Agri-Supply Chain Management: To Stimulate Cross-Border Trade in Developing Countries and Emerging Economies	General	Agribusiness	√	√				Roekel, Willems, Wageningen	World Bank
AMAP BDS Knowledge and Practice Task Order: Lexicon	General	VC	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	Dunn	USAID
Commodities, Diversification and Poverty Reduction	General	Agribusiness	\checkmark	\checkmark	\checkmark			Humphrey	FAO
Commodity Chains and Global Capitalism	General	VC	\checkmark	\checkmark	\checkmark			Gereffi, Korzeniewicz	
Competitive Strategies for Agriculture- related MSES: From Seeds to Supermarket Shelves	General	Agribusiness					\checkmark		USAID
Compilation of insights on the online debate Value Chains in Rural Development (VCRD): The role of Donors in Value Chain Interventions	General	VC		\checkmark			\checkmark	Roduner, Gerrits A., & Gerrits A	SDC
"Customized Competitiveness" Strategies for Horticultural Exporters: Central America Focus with Lessons from and for other Regions	General	Agribusiness						Reardon	USAID, MSU
Etude sur la Compétitivité des Filières agricoles dans l'espace UEMOA	Africa	VC		√	\checkmark		\checkmark	Faivre Dupaigre, Baris, Liagre	ECOWAS
Exporting Out of Africa: The Kenya Horticulture Success Story	Africa	Agribusiness		\checkmark	\checkmark			English, Jaffee, Okello	World Bank
Globalization and the Small Firm: A Value Chain Approach to Economic Growth and Poverty Reduction	General	VC	\checkmark	\checkmark	\checkmark		\checkmark	Downing, Field, Kula	USAID
Globalization of the Agro-food System: Success and Challenges for Promoting Africa's Horticultural Exports	Africa	Agribusiness						Gabre- Madhin, Minot	IFPRI
Governance in Global Value Chains	General	VC		\checkmark	\checkmark			Humphrey & Schmitz	IDS

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

The Governance of Global Value Chains	General	VC		\checkmark	\checkmark	\checkmark		Gereffi, Humphrey, Sturgeon	Rockefeller Foundation
Guide to Commodity-based Export Diversification and Competitiveness Strategies for African Countries	Africa	Agribusiness		\checkmark				Stryker, Salinger	AIRD
A Handbook for Value Chain Research	General	VC	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	Kaplinksy & Morris	IDRC
High Value Agricultural Products for Smallholder Markets in Sub-Saharan Africa: Trends, Opportunities and Research Priorities	Africa	Agribusiness						Temu & Temu	ICTA
Implementing Sustainable Private Sector Development: Striving for Tangible Results for the Poor: The 2006 Reader	General	PSD	\checkmark				~	Miehlbradt, McVay, Tanburn	ILO
Integrating SMEs in Global Value Chains: Towards Partnership for Development	General	VC	\checkmark			\checkmark	\checkmark	Kapinsky, Readman	UNIDO
Participatory Market Chain Approach	General	VC	V	\checkmark				Bernet, Devaux, Ortiz, Thiele	
Promotion of Commercially Viable Solutions to Subsector and Business Constraints	General	Agribusiness					\checkmark	Lusby, Panlibuton	USAID
Strategies for Diversification and Adding Value to Food Exports: A Value Chain Perspective	General	VC		\checkmark				Humphrey & Oetero	UNCTD
Successes and Challenges in Promoting Africa's Horticultural Exports	Africa	Agribusiness	\checkmark	\checkmark	\checkmark	\checkmark		Gabre- Madhin, Minot	World Bank
Trade, Micro and Small Enterprises, and Global Value Chains: microREPORT #25	General	VC		\checkmark	\checkmark		\checkmark	Barber & Goldmark	USAID
Upgrading Global Value Chains	General	VC		\checkmark	\checkmark	~~~~		Humphrey	ILO
Value Chain Analysis for Policy-Makers and Practitioners	General	VC		\checkmark	\checkmark		\checkmark	Schmitz	ILO, Inst. Dev. Studies
Value Chains and their Significance for Addressing the Rural Finance Challenge	General	VC		\checkmark	\checkmark		\checkmark	Akin, Fries	USAID

Undoubtedly, academics, international organizations, bilateral development institutions, and nonprofit organizations have done a great deal of work related to these common themes. There is broad agreement on the importance of these identified themes and concepts. But the nomenclature is still unclear. Attempts have been made to define and provide a lexicon, but inconsistencies persist. The concepts and definitions used below result from experience, generally admitted expert terminology, and, in some cases, ad-hoc definitions that will be made explicit.

Summary of General Themes and Structure from the Literature Review

It should be a goal of the value chain participants to be competitive and agile in terms of identifying and taking advantage of market opportunities. The factors determining value chain competitiveness are often contextual and should be reviewed when considering the value chain as a complete entity. These factors include, but are not limited to, the business environment, supporting and end markets, and inter-firm or firm-level cooperation.¹⁸

Of course, the reality is that value chains are not always well managed and integrated entities. Even when market opportunities or mutual benefits are apparent, there are forces within a value chain that disrupt or impede its optimal management. These internal relationships and behaviors must be understood in order to upgrade the value chain as a whole, meet market demand, and increase competitiveness. The internal relationships affecting value chains include, but are not limited to, power, learning, innovation, and benefits. Attempts at externally managing value chains – for example through government regulation or marketing boards – have generally not produced stellar results, and their weaknesses often accumulate over time.

Any economic development initiative that aims to help a segment of a value chain or the entire value chain to upgrade, increase efficiency, lower transaction costs, capture more value, and become more competitive needs to consider the value chain's particular local competitiveness factors, as well as the internal relationships previously mentioned. There are, of course, numerous iterations of "steps in value chain development" that are dependent on the focus or goal of a project, which may be territorial and needs-based or market-driven and private sector-based.

As such, these foci certainly change the steps to be included in an effective value chain development initiative. However, there are general practices, and there is a "typical" structure and form to structuring such an initiative. It entails working out the following: Industry Selection; Industry Analysis; Industry Competitiveness Strategy; Implementation Action Plan; Performance Monitoring and Assessment;¹⁹ and Feedback for Improvement.

Creating Trust

At the heart of value chain development is the effort to strengthen mutually beneficial linkages among firms so that they work together to take advantage of market opportunities. Nearly all of the documents on value chain development contain this notion of the importance of inter-firm cooperation and creating economies of scale through better interactions.

¹⁸ PSD Reader 2006.

¹⁹ PSD Reader 2006.

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

Most value chain-based initiatives work with a range of business types to strengthen both vertical linkages (between firms that buy from and sell to one another) and horizontal linkages (between firms that serve the same functions in the value chain). These inter-firm connections are especially important to consider when examining how agile a value chain can be regarding market developments, or how able it is to link to markets. Positive outcomes undoubtedly result when there is a strong market drive for linkages, strong investment from many businesses in the chain, and a market system in place to replicate improved models and practices.²⁰

More specifically, inter-firm cooperation refers to the joint action between two or more firms in a value chain. It includes horizontal and vertical linkages between firms and can be formal or informal. Examples include information sharing, bulk purchasing of inputs, subcontract farming, and industry branding campaigns. Unfortunately, value chain participants frequently do not work cooperatively, and market conditions sometimes propel firms to adapt cutthroat measures while competing for highly segmented market rents.

Too often, many participants in a value chain choose not to collaborate among themselves due to lack of leadership, mistrust of competitors, weak information, or lack of scale. Without a strategic direction for the value chain and effective management of its economies, a cutthroat and zero-sum mentality can take hold among value chain participants, who then ignore or cannot see the benefits of cooperation, including the mechanisms that foster it. Competition is, of course, useful, but that usefulness is limited if it blinds the participants to productive collaboration and incentives. For example, if a lead exporting firm or monopoly is able to concentrate its buying power, it can rely on multiple suppliers for inputs and ignore price incentives for service and quality. Given its situation, the lead firm might restrictively determine the price, erect barriers to entry, and prevent the dissemination of information or opportunities to innovate.

A situation that prevents collaboration (beyond simply transactional relationships) can leave producers/suppliers competing among themselves for less lucrative rents and with little opportunity to capture more value. In such low-trust value chains, the lead firm(s) may perpetuate its short-term advantage by switching (or threatening to switch) suppliers in constant pursuit of cost advantages.

These characteristics of value chain relationships have the largest effect on the level of trust between participants:²¹

- Length of trading relationship
- Ordering procedures
- Contractual relationship
- Inspection
- Degree of dependence
- Technical assistance
- Communication
- Price determination
- Credit extended

²⁰ ILO 2006 PSD Reader, p. 43.

²¹ A Handbook for Value Chain Research.

• *Outsourcing payment terms*

The literature frequently emphasizes the idea that building trust by rewarding collective action among stakeholder participants is crucial for upgrading a value chain. Indeed, working within value chains requires establishing relationships in order for participants to gain the "win-win" perspective. When trust, learning, and benefits are shared among firms (vertically and/or horizontally), there is a greater likelihood of generating collective efficiency and scale. Increased trust has also proven to lead to greater specialization by the value chain, as well as eventual outsourcing that provides cost advantages.²² For instance, the more that suppliers can be trusted to meet quality, quantity, and time requirements, the less time and money need be invested to monitor performance. High-trust situations enable lead firms within the value chain to assist each other in achieving common objectives.

Value chain development initiatives have orchestrated stakeholder meetings by providing neutral outsiders to help build trust among the participants. Oftentimes, removing the "zero sum" mentality among value chain participants requires more objective assistance. An outside and neutral facilitator can drive home the concept of participating in a mutually beneficial commercial relationship, but it remains necessary that the value chain participants drive the entire process.

More recently, observers have questioned the sustainability and utility of non-profit-motivated organizations and government agencies acting as market advisors to value chain participants. Their argument states that, for sustainability purposes, market conditions, rather than government agents or non-profit professionals, should determine the role that participants take and the relationships that are formed. There is substance to this criticism, but depending on the focus of the development intervention, some (temporarily) subsidized support services may be needed in order to stimulate or create markets in under-serviced areas, or where market demand has not yet been recognized.²³

Governance: What Type of Power Relationship Exists, and Is Information Being Shared?

"Governance" is a description of the dynamic distribution of power, learning, and benefits among a value chain's firms. While the term can have many meanings, in this instance we use it to describe the sharing of information and systematic standards promoted by the "governing" entity in a value chain.

Governance can be characterized along a continuum of four types of relationships that center on information and the use of market power:²⁴

Market relationship: Arms-length transactions in which there are many buyers and many suppliers. Repeat transactions are possible, but little information is exchanged between firms, interactions are limited, and no technical assistance is provided.

²² Adapted from: Theodore H. Moran. August 2001. "Parental Supervision: The New Paradigm for Foreign Direct Investment and Development." *Policy Analyses in International Economics, Issue 64.* ²³ PSD Reader 2006.

²⁴ Lexicon, 2005.

- **Balanced relationship**: Both buyers and suppliers have alternatives, i.e., a supplier has various buyers. There are extensive information flows in both directions, with the buyer often defining the product (i.e. design and technical specifications). Both sides have capabilities that are hard to substitute, and both are committed to solving problems through negotiation rather than threat or exit.
- **Direct relationship**: Main buyer takes a large percentage of supplier's output, defines the product (i.e. design and technical specifications), and monitors the supplier's performance. The buyer provides technical assistance and knows more about the costs and capabilities of the supplier than the supplier does about the buyer. The supplier's exit options are more restricted than those of the buyer.
- **Hierarchical relationship**: Vertical integration of value-added functions within a single firm. The supplier is owned by the buyer or vice versa, with junior firm having limited autonomy to make decisions at the local level.

Governance ensures that interactions between firms along a value chain exhibit some level of organization rather than simply being random. Value chains are governed when the parameters requiring product, process, and logistic qualification that are set have consequences up or down the value chain, encompassing bundles of activities, actors, roles, and functions.²⁵ Of course, one objective of value chain development is to engender informed, incentive-producing governance targeted at achieving high-value results.

In many sub-Saharan Africa cases, certain key actors – the lead firms or "governors of value chains" – have the capability and power to define and set the parameters of contracts and subcontracts in their supply chains. For example, they can define chain-wide product and process standards, quantities, and conditions of delivery. This power may be based on ownership of well-established brand names, proprietary technology, or exclusive information about different product markets, which enable the firm to act as a system integrator.²⁶

The governing entity is often a lead firm that is closer to the market or business environment that the value chain inhabits. Leadership or being a "lead" firm implies success, efficiency, a competitive composure, or direct information. As such, working with lead firms is often the most efficient and effective way for a development program or market partner to reach a large number of chain participants. The advantages are an easy entry point, a possibly guaranteed market or reliable supplier, leveraged technical expertise, and marketing connections. The disadvantages are that the lead firms too often seek exclusive producers, resist replication, and limit competition, while small value chain participants risk becoming too dependent on one lead firm.²⁷

Market or Bargaining Power (Buyer vs. Producer)

All approaches to value or supply chains identify the crucial impact of power relations among different actors. Power relations determine how economic gains and risks are distributed among value chain actors (see the accompanying figure) and to what extent dominant firms may set and enforce standards with the aim of raising entry barriers for competitors and achieving market

²⁵ A Handbook for Value Chain Research, p. 29.

²⁶ Altenberg, Tillman, "Donor approaches to supporting pro-poor value chains," p. 10, 2006.

²⁷ PSD Reader 2006.

foreclosure. The concept of "governance of value chains" implies that "there are key actors in the chain who take responsibility for the inter-firm division of labor, and for the capacities of particular participants to upgrade their activities."²⁸



Figure 3: Power Relations in Value Chains

"Market power" refers to the idea that one firm in the market may be able to exert significant influence over the goods and services traded or the price at which they are sold. Governance plays a large part in determining and explaining various firms' market power. However. commercial competence. market forces. and technical capabilities also determine the market power of value chain participants.

A commercial transaction's price is determined by the bargaining power of the transacting entities. In

economics, bargaining power refers to the ability to set prices or wages, usually arising from some sort of monopoly, monopoly-like position, or non-equilibrium situation in a market. The economic actor with greater bargaining power has the greater decision-making freedom. Typically, value chains feature two types of bargaining power relationships: buyer-driven and producer-driven.

Buyer-driven value chains refer to a market context where producers have few options for selling their goods or services. These chains typically have low barriers to entry at the producer level, or they may have locational/logistics limitations on who the producer can sell to (e.g. "captive" tea producers for a tea estate or cotton producers for a ginner). This type of market condition is referred to as a "buyer's market" (if not a monopolistic one).

International brand-name and retailing companies ("buyers"), therefore, define the "rules of the game" in particular industries and appropriate the largest share of the gains from those industries' production. In the developing world, buyer-driven value chains are often characteristic of labor-intensive industries like agriculture, clothing, and furniture. The typically low barriers to entry on the production side of buyer-driven value chains mean that a multitude of suppliers compete for very low rents. Essentially, the buyer has an advantage because these numerous producers compete to offer goods and services at the lowest cost. Quality and other standards tend to be imposed by the buyer or appear as entry requirements to the buyer's geographic marketplace (e.g. EurepGAP).

Producer-driven value chains are often characterized by knowledge intensity, relatively higher levels of technology or skills, scarcity, high levels of marketing, or capital-intensive production practices. These high-level factors, scarcities, and differentiations produce barriers to entry for competition. These barriers to entry include the large amounts of capital needed for investment and, therefore, entry, limiting the number of producers. Similarly, environmental factors, such as

²⁸ Raphel Kaplinsky (2000) "Spreading the Gains from Globalization: What can be learned from value chain analysis?," Unpublished paper, Institute for Development Studies, University of Sussex, March

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

location, can present physical barriers. The products of producer-driven value chains often require high research and development (R&D) expenditures or have been branded with costly marketing efforts in order to increase barriers to entry and protect market shares.

Examples of producer-driven value chains are perhaps most readily seen in complex technology industries, such as the commercial airline, automobile, or computer industries. But they are also often present in agricultural sectors when freshness standards and protected varieties are important, when there is high product differentiation, when packaging and logistics are complicated, or when R&D and other knowledge elements in production or processing are critical. In such situations, the producer can capture a higher profit because sales are high-margin and based on factors such as quality management or differentiation, rather than on strictly costbased competition. Examples from the agricultural sector include organic products like cotton, branded products like processed and packaged agricultural products, quality-differentiated products like specialty coffees, or high-value processed products like essential oils. For these products, those with knowledge of and access to retail distribution chains are often the governors of value chains, and routine activities are outsourced to networks of suppliers and distributors whose profits can be squeezed.

Power and governance in value chains are common themes in the literature, but their relation to the prospects of upgrading value chains in developing countries is inconclusive. The literature seems to recognize that, in reality, value chains are not purely buyer-driven or producer-driven, and that they can and do change. Indeed, the creation of competitive advantage by elements of the value chain is a means to alter the power balance among actors within the chain, or vis-à-vis competing chains.

Innovation, Information, Information Sharing, and Knowledge

In competitive markets, innovation helps maintain or grow market share or profits and can be a route to competitiveness and the development of competitive advantage. Innovative production and processing can create cost efficiencies and improved service that translate into higher margins or more competitive pricing. Innovations in logistics can also provide cost efficiencies, as well as improved service. In terms of value chain development, innovation must be viewed as necessary for overall chain competitiveness by capturing more value or upgrading the value chain. The production factor of "knowledge" Is one core factor regarding the upgrading of value chains.²⁹ A value chain's access to information (e.g. regarding market trends) can itself be a competitive advantage.

The way that knowledge is transferred is determined by the information flows or linkages between firms within a value chain. Targeted transfer of knowledge by the lead firms, using backward linkages, is usually seen in the development of product specifications (e.g. quality, preferences, or certifications) and other expectations (e.g. price, quantity, or time) that are communicated by the lead firm to its suppliers. This relationship is often part of an "embedded services" relationship between buyer and producer. In such relationships, firms or producers pay for the services of technical assistance or innovation in the price they get for their products.

²⁹ Porter, Michael, "Competitive Framework."

In general, learning processes among firms are most effective where they are located close to one another and are able to therefore benefit from a high level of communication, networking, and other exchanges of information. If the rate of innovation is lower than that of competitors (whether firms or competing value chains), this may result in declining market share and value added; in extreme cases it may also involve negative growth. Thus, innovation has to be placed in a relative context – pace compared to competitors – that can be referred to as upgrading.

Upgrading refers to the innovation that increases firm and/or value chain competitiveness. According to the AMAP BDS lexicon developed by USAID, there are five categories of upgrading. 30

- 1. **Process upgrading**: Increasing efficiency (i.e. more output for same level of inputs or same output for lower level of inputs)
- 2. **Product upgrading**: Improving product quality
- 3. **Functional upgrading**: Operating at a new level in the value chain
- 4. **Intra-sectoral upgrading**: Operating in a new market channel within the same value chain
- 5. **Inter-sectoral upgrading**: Producing a completely different product in a completely different value chain

As distinguished from innovation, the concept of upgrading explicitly recognizes relative endowments and, hence, the existence of value. The upgrading approach emphasizes issues of knowledge creation, transfer, and appropriation. Critical questions are raised regarding the manner in which knowledge flows along value chains, firms acquire information and upgrade processes, firms "unlearn" certain capabilities as they specialize, types of knowledge are transferred by technology proprietors, and firms disclose their core competencies. However, this field requires substantial further research.

Intervention Entry Points and Initial Focus

Many documents, including many project reports, describe methods that various types of stakeholders use to work with value chains. The choice of an initiative's entry points, partners, tools, and approaches strongly depends on the characteristics of the value chain, its participants, the business environment, and many other factors.

Much of the literature discusses thematic priorities – elements of value chain operation or strategy around which an initiative can take root. The literature discusses such properties as:

- **Business opportunity**: Creation or expansion of a market for a high-demand product
- **Commodity:** Promotion of a commodity through better collaboration or market actors
- **Targeting the poor**: Focus on improving the livelihood systems of a particular region or vulnerable population
- Geography: Promotion of products with good market potential and comparative advantage
- **Technology**: Adaptation of existing technologies to local area

³⁰ Lexicon, 2005.
- **Consumer interest**: Selection of products that have safety or nutritional benefits in order to promote food security, food safety, and/or consumer protection
- Ideas: Support for ideas generated by value chain participants or local stakeholders
- **Service**: Selection of a service with growth potential that is currently only provided by young businesses where demand may not yet be realized
- **Cost savings**: Collaboration on sourcing inputs at lower cost through combined purchasing or transport savings
- **Implementation:** Applying the themes to implementation tools and approaches

It should be axiomatic that interventions be based on well-considered needs or strategies of the value chain and its participants, rather than a "solution looking for a value chain."

As mentioned, we have selected themes for emphasis that are highly pertinent to upgrading and improving value chain targeting, linkage, and operation and that are the current focus of much innovative thinking and practice. These themes are particularly applicable to Africa's agricultural value chains.

These themes resonate throughout the many cases described in the following section of the Guide. The Guide's tools and approaches make use of the thematic principles that this section has highlighted, as well as several other intervention ideas.

Improving value chain competitiveness in Africa presents challenges, but is, of course, subject to the same worldwide trends, competitive forces, and business principles that are common to value chains around the world. The specifics of any value chain initiative are nonetheless highly contextual.

The second portion of this Guide, which follows, discusses methodological tools and approaches that incorporate important value chain concepts with the themes discussed and with sound business principles. The accompanying cases, mostly from Africa but also from the Americas, Asia, and Europe, illustrate the application of the tools.

Section 4: Discussion of Individual Tools

Introduction and Overview

This Section presents 13 value chain implementation tools. They are presented in 6 themes or elements within a program implementation cycle. Figure 4 illustrates how the focus of each tool and theme relates a program implementation cycle. The themes, and a brief explanation of the tools that fall within them, are shown below:

Theme One: Designing strategies and business plans—Obtaining and using information

Tool 1: Choosing priority sectors for value chain interventions

This tool is designed to help practitioners consider: Which are the priority value chains? Which ones should be supported? Why does comparative advantage matter, and how can it be assessed? How should public, private, and collective perspectives and interests be harnessed?

Tool 2: Designing informed strategies across the value chain

Offers analytical methods for understanding the value chain and integrating the information into sound strategy along various points of the chain.

Tool 3: Conducting benchmarking and gap assessments of value chains

This tool describes how to measure and compare a value chain's performance (whether in relation to itself, similar value chains, or to best practices) as a means of gaining insight into appropriate strategic choices.

Theme Two: Developing robust new businesses

Tool 4: Upgrading and deepening the value chain

Explains how to move the value chain in new directions that may add efficiency, improve product quality, add new operations, and more.

Tool 5: Identifying business models for replication

This tool focuses on opportunities to implement sound business models repeatedly within a value chain. Such replicability is useful in increasing value added volumes, intermediation, and access to service and inputs.

Theme Three: Supplying the market—Aligning supply to match market opportunity

Tool 6: Capturing value through forward and backward integration

This tool explains how vertical integration can help businesses ensure supply or otherwise control inputs, capture more value, achieve economies of scale, and/or ensure access to information.

<u>Tool 7: Horizontal Collaboration - Creating and Taking Advantage of Economies of Scale</u> This tool provides approaches to create economies of scale that help to increase production, ensure quality, improve access inputs, and achieve more market power.

Theme Four: Reaching the market—Market positioning and market opportunities

<u>Tool 8: Positioning products and value chains for greater value and competitiveness</u> This tool describes how competitiveness positioning considerations can enable businesses to choose wise, market based value chain business models and provide strategic direction to many value chain actions to improve competitiveness.

<u>Tool 9: Applying standards and certifications to achieve greater quality</u> How meeting (and exceeding) the quality and performance standards of desired markets can help achieve entry, market share, and higher unit values for a value chain's products.

Theme Five: Improving the business and policy environment

Tool 10: Identifying needed support services for the value chain

This tool discusses how improving the depth and breadth of services offered to a value chain can help member firms to be commercially sustainable and improve operations.

<u>Tool 11: Improving the operating environment by promoting public/private dialogue</u> This tool describes how value chains can improve their operating environments by engaging the public sector and other actors in effective public/private dialogue.

Tool 12: Achieving synergies through clustering

How cluster-strengthening and cluster-development initiatives can help value chain participants achieve results that unique emphasis on core value chains may not be capable of generating.

Theme Six: Monitoring results in value-chain development

Tool 13: Monitoring achievements in value chain performance

This tool explains how monitoring and evaluation methods can help value chain participants track implementation progress, evaluate value chain performance, and identify the impacts of initiatives.

Each of the tools presents value chain-based actions that can form part of a competitiveness-focused agriculture development initiative.

The tools were selected to provide a broad range of interventions across and within the value chain and its operating environment. Space restrictions have required the authors to limit the number of tools and approaches covered in this Guide. The thoughtful reader will recognize that many other tools are available and could be described, or that any one of the included tools could be considered in much greater depth. Perhaps these will be the subject of a second volume of this Guide.

While the Guide's emphasis is on implementation, actions are likely to be most effective when underpinned by sound information and analysis. Therefore, we begin with three largely analytical tools (selection, understanding the value chain, and benchmarking—Tools 1, 2 and 3); and point out in subsequent sections where a strong analysis is an important input or prerequisite.

Each tool is followed by descriptions of one or more actual cases that illustrate the tool's application, with embedded mini-cases offering additional illumination. About 60 percent of the examples are from Africa; others are from Europe, Latin America, and Asia. We return to the same cases as appropriate when describing different tools, which serve to highlight the multiple dimensions of a single value chain and the rich set of opportunities value chain initiatives can offer.

Figure 4: Value-Chain Program Implementation Cycle and Tool Themes

The Value Chain Framework: A Program Implementation Cycle with Corresponding Tools and Approaches*



* The Value Chain Implementation Cycle is in part adapted from Action for Enterprise's Value Chain Approach & JE Austin's Productivity and Value Enhancement Model (see Figure 6).

Tool 1: Choosing Priority Sectors for Value Chain Interventions

This tool describes how investors, governments, donors, and other organizations can prioritize among the value chains in which they can potentially invest their resources. The process can help businesses and domestic or foreign investors to determine where they can invest most profitably and can help planners in the public sector and development partner organizations to select which value chains to support.

In most cases, the process will start with an initial list of potential value/product chains, followed by various types of market analyses, an evaluation of the operational and strategic opportunities and their potential impact, and a willingness assessment of the chains leaders and other actors to work towards improving their collective competitiveness.

This process does not necessarily need to be carried out in a linear fashion. In fact, involving various stakeholders in the early stages of the analysis is an excellent means of gauging (and building) their willingness to invest time and resources in the resulting initiatives.

In the case of governments and donors, the process assumes that the program has not prescribed *a priori* which value chains it must support. Such pre-selection should be avoided. Selecting before testing can distort incentives and project dynamics among stakeholders, effectively limiting the project's prospects for a meaningful impact.

The discussion within this tool is not intended as an exhaustive list of criteria upon which value chains can be prioritized. It should be considered a guide for businesses, planners, and practitioners to identify which value chains offer the best opportunities to invest profitably, encourage economic growth, and enhance competitiveness. Each business or practitioner can add criteria depending on their particular focus. Typical additional criteria include involvement of small and medium enterprises, variability of returns, gender considerations, environmental impact, and many others. A multi-criteria approach is, in fact, a commendable approach to filtering alternatives.

Initial List

To begin, it is helpful to compile a list of products/chains that could become more competitive and thus enhance income growth. Narrowly speaking, and often perhaps too simplistically, competitiveness in such cases is simply interpreted as being able to sustainably export at a profit, or to compete favorably against imports (in a way that increases productivity and, hence, the incomes of affected populations).

There are many sources from which an initial list of products can be compiled. These include consultation with experts, previous assessments, products in industry sectors that have been identified as national priorities, and sectors being supported by other initiatives.

In addition to these sources, quantitative tools that rely on the concept of comparative advantage are often used to refine and prioritize the potential list of products/value chains. The comparative advantage concept focuses on the relative efficiency of producing different goods in the home country vis-à-vis the rest of the world. It must be emphasized, though, that the fact that a product

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

possesses comparative advantage does not necessarily mean that it can be produced and sold at a profit under prevailing economic conditions. Many other elements need to be in place—including appropriate marketing links and input supply channels, financing mechanisms, uniform product quality, and many other demand requirements. In other words, comparative advantages can be built into competitive advantages.

To identify products/chains with a comparative advantage, practitioners have often used the Revealed Comparative Advantage (RCA) Index. The RCA denotes relative efficiency indirectly, based on trading patterns that emerge from actual market transactions. RCAs identify the extent to which an exporting country captures world market share in a particular area relative to the degree at which it captures export market share for all traded goods.³¹ An RCA greater than 1.0 indicates a comparative advantage for that item, while an RCA lower than 1.0 identifies a comparative disadvantage. The RCA is, however, imperfect because it embodies not only the fundamental economic factors affecting relative efficiency but also government policies and institutions that may distort markets. The usefulness of RCA indexes is also limited because they indicate past performance but do not take into account current market dynamics and likely future trends and conditions in those markets. As long as these imperfections and limitations are recognized, RCAs can be helpful as analysis tools, since data are generally available in the trade record to gauge comparative advantage.

An alternative measure of comparative advantage is the Domestic Resource Cost coefficient (DRC), which compares the cost of domestic production with world prices. The DRC measures the dollar cost in domestic resources of earning or saving a net dollar of foreign exchange. Therefore, values below 1.0 indicate a comparative advantage and values above 1.0 a disadvantage. To the extent that a country persists in producing commodities whose DRCs are greater than one, its resources are being poorly utilized. The country would have little chance of increasing exports of those items or increasing domestic production to substitute for imports, since they are inherently non-competitive on world markets. Therefore, policies that encourage the production of products without a comparative advantage tend to drag down the country's agricultural growth rate and opportunities for employment creation.

Perhaps more importantly, the process of compiling DRCs can be done in such a way that disaggregates cost data by productive stages along the chain. Presenting data in this manner illustrates how income is distributed among the different links in the value creation process and allows for the analysis of other weaknesses and inefficiencies in each link. Tool 2 demonstrates how to use and benefit from this information.

A drawback of DRCs is that data required to calculate them are often not readily available for many commodities. The calculation of DRCs requires data on domestic prices, international prices, government subsidies, and taxes for the specific commodities being evaluated, as well as the shadow price of foreign exchange. In addition, these indicators require information about the

³¹The formula for calculating the RCA index is of a given product (p) is RCA_p = $(x_p/X_p)/(x_t/X_t)$; where X_p are total world exports of product (p), x_t stands for total country exports, and X_t for the world total of all kinds of exports.

proportion of tradable and non-tradable inputs used to produce one unit of each particular good. Given these requirements, it is difficult to assemble such detailed data for all but a few commodities in a limited number of countries. DRCs, though, are often the preferred indicators of comparative advantage when the focus of attention is restricted to a few commodities and/or trading areas. However, since DRCs are estimated on the basis of certain production conditions, their results most likely do not apply country-wide.³²

The RCA and DRC approaches provide useful but incomplete information and guidance. Thus, they must be tested using a sensitivity analysis – testing different assumptions to see how and to what extent the outcomes change – and considered in the light of other important decision criteria. Among these criteria are market strength, domestic capacity, and level of commitment.

Box 1: How to Choose Value Chains for Intervention: The Example of Senegal's Projet Croissance Economique

Before a detailed analysis can be performed to determine the most appropriate subsectors on which to focus interventions, planners must generate a list of possible subsectors from which to choose. The selection of items for the list can be conducted informally—for example, brainstorming the obvious sectors in a country or basing the list on the most widely grown crops. Or the process can be more rigorous such as in Senegal, where a more deliberate approach was taken to identify the list of possible target subsectors.

Senegal's Projet Croissance Economique is a five-year program supported by USAID* that began in 2005. The aim of the project is to help Senegal stimulate accelerated growth, competitiveness, and trade. Its initial work has focused on improving the value chains of a number of products produced in Senegal in order to achieve greater production and productivity, higher prices, and increased exports. The project's approach is to ensure that Senegalese stakeholders are engaged at every juncture along the production process so that improvements are sustainable.

At the start of the project, the GoS, donor, and consulting firm collaborated to create an initial long list of possible value chains on which to focus interventions. This initial list included many subsectors, but through a sequential refinement process, the list was shortened to six subsectors. This process involved five stages:

- 1) **Mandatory sectors:** If the Government prioritized development of a subsector, it was automatically given strong consideration. These priority subsectors were identified based on Government policies that emphasized the subsectors because they were considered vital to the country, had high value added, or were import substitutes. Using these criteria, cotton and horticulture-related products were priorities.
- 2) High economic impact sectors: Next, subsectors with extensive economic impact on the country were given priority. The team looked for subsectors with perceived competitive advantage, perceptions of high impact on rural incomes and employment, or export market potential, including sectors potentially benefiting from the U.S. African Growth and Opportunities Act (AGOA). Cashews, mangoes, dairy, and bissap (hibiscus tea) were identified through these criteria.
- 3) Sectors with private sector appeal: Finally, the team looked at subsectors that were of interest to private enterprise, were already a focus of the private sector, or that had a high likelihood of attracting private domestic and foreign direct investment. Banana production for export was already targeted by private sector investors, and export of neem seed to the United States (where it can be used as a raw material for organic agriculture inputs) was promoted by a foreign investor, so those subsectors were also considered priorities.
- 4) Applying additional criteria: Supplementing the criteria highlighted above were crosscutting themes like poverty reduction, women's empowerment, and employment creation. The final short list that was created included six subsectors: mangoes, cashews, bissap, fonio (a small millet), woven textiles, and fisheries.

³² Background on the concepts is taken from Norton, Roger and Alvaro Balcazar, "A Study of Colombia's Agricultural Competitiveness," published jointly by United Nations Food and Agriculture Organization, the World Bank, and United States Agency for International Development, November 2003.

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

5) Value chain analysis and feasibility analysis: Next, a value chain analysis and feasibility analysis were conducted to verify soundness and opportunities, to determine which three of the initial six subsectors would be selected for the interventions, and to guide the nature of the particular value chain emphasis within each sector. Cashew, bissap, and mango were the initial choices.

This project is now over a year old. As it has progressed, changes in market opportunities have caused the list of selected subsectors to be expanded to include bananas, gum Arabic, neem, and dairy.

Authors: Carlton Jones and Martin Webber – J.E. Austin Associates *Implemented by a consortium including International Resources Group and involving J.E. Austin Associates.

Market Analysis

Market analysis to examine the nature of demand – its size, tendencies, segments and potential niches due to seasonality and other factors, price tendencies, customer preferences, current competitors, market access, and other requirements – is particularly important in prioritizing product/value chains. A market analysis allows the investor, business, or planner to identify whether attractive opportunities to improve and upgrade a given value chain exist, using market conditions as the benchmark of what needs to be achieved.

Domestic Capacity and Economic Impact

Once an initial list of potential product/value chains has been vetted by confirming market opportunities, program implementers can evaluate the capacity to respond competitively to those opportunities, as well as the extent to which upgrading and change are needed to be able to produce according to market requirements and customer preferences. This evaluation involves the analysis of institutions, technology, service providers, policies, and other production conditions, in addition to the investments needed to take advantage of the identified opportunities. An assessment of capacity also provides guidance as to important program intervention points, where the opportunities available within the target value chain can be leveraged or unleashed.

Further, given the market analysis and the opportunities identified, practitioners are able to construct scenarios of the economic impact (profits, jobs, increased exports, and wages) that the proposed changes would have.

Testing Commitment

It is essential to take into account the commitment of the chain's leadership and stakeholders when deciding which value chains to prioritize. If stakeholders are not willing to invest time and resources in achieving the identified opportunities for upgrading, little will be achieved. Their motivation will rest on many factors, including their understanding of the likely returns, the time and level of investment required, perceived risk, and the leadership of opinion leaders and champions. It is generally a good idea to include stakeholders in the process as early as possible and conduct workshops to discuss the current conditions of the product/value chain. Their commitment to the process can be gauged by their repeated attendance and expressed enthusiasm. Their level of responsiveness will help practitioners prioritize value chains.

A Step-by-Step Summary of Tool 1: Choosing Priority Sectors for Value Chain Interventions

- Start the prioritization process by compiling an initial list of products/chains whose competitiveness could further be enhanced. Sources for an initial list include experts, previous assessments, national priorities, and sectors supported by other initiatives.
- Add to or refine the list using quantitative tools, such as Revealed Comparative Advantage and Domestic Resource Cost Coefficients, to assess comparative advantage.
- Conduct market analysis to identify if attractive opportunities to improve and upgrade a given value chain exist.
- Evaluate the capacity to respond competitively to those opportunities. Construct scenarios of the economic impact that the proposed changes could achieve using multiple assumptions about profits, jobs, increased exports, and wages among other variables.
- Make the stakeholders part of the process as early as possible, especially through workshops to discuss the current conditions of the product/value chain and to test their willingness and commitment to the process.

Case Study: Prioritizing Value Chains by Using Comparative Analysis—Value Chain Selection in Mozambique³³

Introduction

This case study uses our first tool to decide which specific value chains in Mozambigue are the best candidates for intervention. It illustrates how the DRC can help decision-makers select among value chain options.

Points to Consider

When reviewing this case, it is important to consider the following points:

- What are the steps involved in comparative analysis? •
- Which method of comparative analysis was employed, and why? •
- How do differing assumptions change the outcome of the analysis?
- How does information from the analysis shape future interventions?
- What limitations are there in a DRC-type analysis? How can the analysis be supplemented?

Background

Mozambique is a largely rural country located on the southeastern coast of sub-Saharan Africa. After gaining independence from Portugal in 1975, Mozambique adopted a socialist economic system and later plunged into a civil war that lasted over 16 years. Peace was established in the early 1990s, providing a foundation for economic recovery.

Currently, more than 75 percent of the population relies on agriculture for their livelihoods, although less than 20 percent of the arable land is cultivated, and yields of all major cash crops are less than desirable. Seventy percent of the population lives on less than one US dollar a day, and a third suffers from chronic hunger.

In the early 2000s, Mozambigue was set to receive IDA loan funds from the World Bank for a targeted agricultural program. In preparation, the parties agreed to carry out a study to identify the agricultural sectors with Source: FAO STAT

Table 6: Mozambique Commodities by Price, 2005

Rank	Commodity	Production Price (Int \$1000)	Production Volume (MT)	
1	Cassava	443,169	6,150,000	
2	Maize	168,490	1,450,000	
3	Indigenous Cattle Meat	78,296	37,856	
4	Pulses nes	54,382	205,000	
5	Groundnuts in Shell	53,156	110,000	
6	Rice, Paddy	42,815	201,000	
7	Indigenous Chicken Meat	40,472	34,698	
8	Sorghum	38,305	314,000	
9	Cashew Nuts	38,108	58,000	
10	Coconuts	23,967	265,000	
11	Tobacco Leaves	21,879	12,000	
12	Vegetables Fresh nes	19,703	105,000	
13	Fruit Fresh nes	18,344	115,000	
14	Cow Milk, Whole, Fresh	16,049	60,350	
15	Indigenous Pigmeat	12,977	12,815	
16	Bananas	12,826	90,000	
17	Hen Eggs	12,156	14,000	
18	Potatoes	11,604	80,000	
19	Tea	11,366	10,500	
20	Oilseeds	8,789	30,000	

the greatest comparative advantage. In 2005, an agricultural economist was commissioned to perform an analysis which culminated in a report entitled "Economic Analysis of Comparative Advantage for Major Agricultural Cash Crops in Mozambigue."34

³³ Authors: Carlton Jones and Martin Webber – J.E. Austin Associates

³⁴ The analysis was conducted and the report was prepared by Nicholas Gergely.

Table 7: Original and Revised					
Sectors for Intervention in					
Mozambique					

in containing a c			
Original List	Revised List		
Cashews	Cashews		
Rice	Rice		
Tobacco	Potatoes		
Flowers	Paprika		
Cotton	Cotton		
Export Fruits,	Export Fruits,		
(bananas &	(bananas &		
grapefruit)	grapefruit)		
Sugar	Sugar		

Creating an Initial List

Before the comparative study could be conducted, a list was generated that included possible sectors for intervention. The original list was prepared by World Bank staff and focused on sectors linked to domestic priority crops.

The list was vetted by the Office of Commercial Agricultural Sector Promotion within GPSCA (Gabinate de Promoção do Sector Comercial Agrário), a GoM Ministry. GPSCA felt that potato and paprika were very relevant in the Tete Province. The Bank and GPSCA agreed to remove tobacco and flowers from the list and replace them with these sectors (see the accompanying table).

The Analysis

As previously described, there are two principal means to determine the comparative advantage of an agricultural sector: Revealed Comparative Advantage (RCA) and Domestic Resource Cost (DRC). Of the two, the RCA method is the easier to use. Because it relies on trading patterns that emerge from actual market transactions to indirectly indicate relative efficiency, the data required are typically readily available. However, the RCA method requires data on international market transactions which were not available for Mozambique. Also, the RCA method paints a picture of past performance but does little to take into account future trends.

DRC and RCA are helpful tools, but they alone should not determine which value chain is selected for intervention. These tools should be used in conjunction with other analyses, and scenarios that when combined make the best case for desired and successful outcomes.

The parties, therefore, decided to employ the Domestic Resource Cost method. DRC is highly dependent on selected cost and other assumptions that, when changed, can affect outcomes. For example, real outcomes may vary in cases where unpredictable costs determine value chain profitability. In such instances, flexible cost assumptions allow stakeholders to glean the correlation between outcomes and actual costs. Assumptions regarding other variables, such as location, access to inputs, or extension and training, will similarly affect the accuracy of DRC estimates.

Findings

Keeping in mind that DRC values below 1.0 indicate a comparative advantage, and values over 1.0 indicate that resources are being poorly utilized, the analysis produced these findings:

- **Cashews**: The cashew sector possessed the highest comparative advantage, with a DRC coefficient well below 1.0, at 0.10. Local processing contributes two-thirds of total value added to the product, so local processing is desirable for exporting raw nuts. Cashews also provide a significant profit (32.8 percent) to producers.
- **Rice:** At present, neither of the local rice varieties (Chokwe and Zambezia) are financially or economically profitable, but with technical assistance and credit access schemes, this could be changed. The Chokwe and Zambezia DRCs are both higher than 1.0; with interventions they could achieve coefficient below this value.
- **Potatoes:** Although they command low prices, potatoes are profitable and have a comparative advantage thanks to low production costs. The potato DRC is 0.36, but with interventions to improve yields via better inputs, it could be improved slightly.
- **Paprika:** This crop is somewhat profitable for farmers, but better pesticide and fertilizer interventions could increase profitability. Yields could be increased with the further extension of credit schemes that allow for more input purchases. Paprika profits are lower than those of

tobacco and potatoes, which are the two competing crops in the region. However, paprika has a comparative advantage at the farm and export levels. Its DRC is .58.

- **Cotton:** In comparison to other agricultural crops, cotton is not profitable, due to relatively high family labor requirements. However, in some cases, cotton is slightly profitable, and is the only option for growing a cash crop. This profitability is very fragile, however, as it depends on world prices for cotton which fluctuate. Cotton's DRC is .76.
- **Export Fruits (Bananas and Grapefruit):** Grapefruit is highly profitable in economic and financial terms. Bananas, on the other hand, are barely profitable in financial terms and generate an economic advantage only when taxes are eliminated. Grapefruit's DRC is .15, with bananas at .40.
- **Sugar:** Sugar is not economically profitable but possesses a slight financial profitability because of domestic subsidization. Sugar has the highest DRC at 17.04. It is hoped that, through existing interventions, it can be lowered to .99 by the year 2012.

Of the seven sectors and eight cash crops studied, cashew possessed the lowest DRC (.10), with the grapefruit following close behind (.15).

The study noted that several sectors (rice, potatoes, paprika, and bananas) could have different profitability and DRC levels if technical assistance or different sets of inputs were available. This highlights how DRC can be a useful tool but should not be the only determinant for sector selection. DRC use should be weighed in light of the assumptions used to complete the analysis. Even though cashews



Figure 5: Mozambique Cashew Domestic Value Chain

Source: J.E. Austin Associates, Inc

However, the user or interpreter of DRC analysis should bear in mind the importance of recognizing that when assumptions are changed, the relative desirability of outcomes may change as well. DRC analysis provides important insights and a basis for comparing the profitability of value chains but should not be the sole tool used for sector selection. Complementary analytical tools should be employed to ensure that differing outcomes are identified and considered. For instance, DRC does not take into account market opportunities. Combining a market assessment with DRC analysis facilitates more informed decisions on agricultural interventions.

have the lowest DRC, it could be determined through further analysis that, for example, the relative cost of intervention for another product might make that other crop a preferable fulcrum for interventions.

Outcomes

Today, the Mozambican Ministry of Agriculture refers extensively to this report for planning and analysis. The report also serves as a useful reference for further exploratory research.

Conclusions and Future Steps

with Markets the best comparative advantage were identified based wellon considered cost and other assumptions.

As the World Bank continues supporting agricultural projects in Mozambique, more DRC analyses can be conducted to ensure the data sets remain current, especially with respect to labor costs. Recommendations for future analysis include working with the Ministry of Agriculture's Policy Analysis Department to validate and/or update production costs and prices used in the previous comparative analysis report. It will also be worthwhile to extend the analysis to other cash-producing agricultural sectors such as livestock, fiber, and fuel crops and to conduct regional level studies to show impact in areas where agricultural projects are underway.³⁵

³⁵ Gergely, Nicolas. "Economic Analysis of Comparative Advantage for Major Agricultural Cash Crops in Mozambique." April/May 2005.

Case Study: Innovative Approaches to Assessing the Competitiveness of Value Chains ³⁶

Introduction

The Geomar International Group³⁷, a Canadian-based global consulting firm, adopted innovative approaches to assessing the competitiveness of value chains. The firm analyzed Mali's agricultural sectors by applying a series of analytical tools to a broad range of data. Through this process, Geomar identified value chains for export markets, which provide a basis for import substitution, improve livelihoods for Malian small growers, and contribute to Mali's GDP.

Points to Consider

When reviewing this case, consider the following questions:

- How do the steps that Geomar took differ from other analytical approaches?
- If time and resource are factors, which steps are necessary, and which can be set aside?

Background

Mali sought to exploit its comparative advantage and diversify its agricultural output to increase incomes and decrease reliance on its two largest agricultural contributors, cotton and rice. The Government of Mali, with the support of a \$46.4m loan from the World Bank, launched the Agricultural Competitiveness and Diversification Project (PCDA) in 2005.

The Geomar Approach to Value Chain Analysis & Selection

Geomar created five modules to assess Malian agricultural competitiveness. Each module builds from the previous and helps practitioners progress from a comprehensive list of sectors to those with true marketability, competitive advantage, and comparative advantage. This process also takes into account the demand in existing end markets, identifying new potential end markets, regional climate and growing factors, production capacity, access to finance, infrastructure, and other determinants.

Module 1: Defining Mali's Broad Portfolio of Agricultural Sectors

- <u>Step 1:1</u> Create a comprehensive list of agricultural sectors in the country, including informal ones.
- <u>Step 1:2</u> Categorize each sector using criteria that defines its particular storage, and/or delivery needs, such as perishable, semi-perishable, durable, transformed, semi-transformed, or processed.
- <u>Step 1:3</u> Classify each sector based on potential end markets, such as export markets for consumption or processing, regional markets for consumption, or local markets for consumption.
- <u>Step1:4</u> Summarize the structure of the various value chains by organizing the categories defined earlier, incorporating upstream-downstream relationships and key factors of value addition.

Module 2: Analyzing Market Demand and Market Entry Conditions

- <u>Step 2:1</u> Create a market demand data sheet for every value chain listed in Module One, providing a comprehensive snapshot of that chain's viability and market opportunities.
- <u>Step 2:2</u> Chart opportunities in each end market for all identified classifications. For example, identify European markets for perishable product-value chains intended for export markets, including value and quantity. Do the same for other end markets and sector classifications.

³⁶ Authors: Carlton Jones and Martin Webber – J.E. Austin Associates based on work conducted by Geomar International Group and a presented in a report entitled "Profil Stratégique pour le Développement de La Compétitivité des Filières Agricoles du Mali : Identification des Opportunités Stratégiques de Valorization et de Diversification de l'Agriculture Malienne, Version Préliminaire."

³⁷ Geomar International Group was acquired by SNC-Lavalin in 2006.

Module 3: Analyzing the Competitiveness of Potential Malian Offerings

- <u>Step 3:1</u> Determine the production potential for each sector using information from the first two modules. Add data on number of producers, production, farm yields, unit price, and revenue.
- <u>Step3:2</u> Analyze regional potential based on comparative advantages. Map the key production regions to determine target areas for select agricultural products, highlighting geographic advantages (e.g. access to water or growing seasons); constraints (e.g. distance from main markets, distance from Bamako for transport, pollution, or poor climate); and producible crop sectors.
- <u>Step 3:3</u> Create sub-regional identification sheets that present annual rainfall, temperature, sun exposure, and humidity for each sub-region to determine the suitability of crop selection.
- <u>Step 3:4</u> Illustrate the growing months and seasonal market demands for all products to show areas of opportunity for producers based on their production cycles and periods of crop availability.
- <u>Step 3:5</u> Conduct an analysis that shows constraints and subsequent interventions that would improve value chain competitiveness.

Module 4: Defining Priority Sectors

- <u>Step 4:1</u> Prioritize value chain criteria by triangulating the interests of various stakeholders. Rank the priorities for each value chain based on production sophistication, number of solvent operators that could be integrated into the strategy, strategy duration, social impact, market appropriateness, existing professional organizations, and existing programs.
- <u>Step 4:2</u> Create a matrix that identifies priority sectors. Put priority levels from the final stage of Step 4:1 in columns, and categorize the rows based on various end markets. Insert sectors into their respective boxes, thus identifying the sectors with the highest probability for success and impact. The matrix can be used to balance strategies for a variety of sectors.
- <u>Step 4:3</u> Organize the priority sectors by area using the regional analysis conducted in Module Three that shows which value chains should be implemented in which regions of the country. This table also provides an at-a-glance view of the crops that can be grown in several regions.
- <u>Step 4:4</u> Create a reference index showing growth and economic impacts for certain priority value chains, highlighting all indicators used during the previous modules of the analysis.

Module 5: Competitiveness Planning: Putting the Analysis into Action

- <u>Step 5:1</u> Determine which approach will improve the competitiveness of the sectors in question. In this instance, Geomar recommended that Mali's competitive strategy include short-, medium-, and long-term objectives, each with pragmatic and obtainable interventions.
- <u>Step 5:2</u> Address the issues with solutions, taking into consideration the constraints highlighted in Module Three and their associated investment needs. For example, "Improve technical skills and human productivity by introducing new technologies and training facilities."
- <u>Step 5:3</u> Wrap the entire strategy together by providing implementation guidelines and a framework that encompasses standard, specific activities that fall under four stages: provisioning, production, logistics, and marketing. Overlap these stages with cross-cutting activities.

Looking Ahead

The value chain strategy that Geomar recommended to Mali is currently being evaluated and implemented. It is as yet unclear what the outcomes will be from this comprehensive approach, though it is likely that elements of the strategy could be implemented in their entirety while others could be augmented to fit the ever-changing market demands Mali seeks to address. In the end, Malians will have invested in a strategy that takes into account the many elements of value chain analysis that have often overlooked in previous strategies. This, in itself, will prove to be beneficial in increasing the institutional knowledge, not only for the Government, but also for all stakeholders taking part in the implementation of PCDA.

Tool 2: Designing Informed Strategies Across the Value Chain

A strategy is an integrated set of choices about which markets a firm or set of firms should serve; how to compete in each market; and how to allocate resources to the most valuable production, marketing, and support activities. In increasingly integrated regional and global markets, firms don't merely compete against firms – value chains compete against value chains. Increasing productivity and competitiveness requires a comprehensive strategy. In vertically integrated industries, firm strategy and value chain strategy are synonymous. However, in fragmented supply chains, buyers, suppliers, and intermediaries must work together to increase their productivity and to raise the value of the end product.

This section outlines a selection of fundamental analytical tools for understanding the quality and

efficiency of a cascading series of activities that increases value added.³⁸ A vast set of tools exist that can be used to describe. analyze, and evaluate the value chain. Several are highlighted here demonstrate the types to of and information analysis that typically guide the strategies and actions to enhance the productivity of a value chain. Since the analytical tools outlined below are point-in-time assessments, they should be updated on a continuous basis to ensure that any changes in market dynamics the or performance of the business are quickly integrated into a new strategy.

Figure 6: Enhancing Productivity and Value Across the Value Chain



Source: J.E. Austin Associates, Inc

Enhancing Value Chain Productivity

Competitiveness is determined by the productivity (value per unit of input) with which a firm or value chain uses its human, capital, and natural resources. Productivity depends on the value, uniqueness, and quality of a product, in addition to the efficiency with which it is produced. Productivity can be strategic, as well as operational, in nature. The overall productivity of a value chain can be segmented into various components, each of which can be analyzed and addressed in concrete, tangible ways to increase the overall return to businesses within the chain.³⁹ The graphic above and the more detailed explanations below outline a framework and key questions

³⁸ This value chain productivity framework was developed by J.E. Austin Associates for use in developing and implementing industry strategic agendas with public and private sector leaders.

³⁹ Productivity is the efficiency by which a firm uses its natural resources, labor and financial resources. It is measured in terms of output per unit hour of labor, capital-output ratios and total factor productivity. These drive returns to labor and capital, which in turn drive wage levels and overall prosperity.

that can be used for analysis and evaluation in guiding the development of a comprehensive, integrated value chain strategy.

Assessing the Overall Status of the Value Chain

An awareness of the distribution and flow of profits in the value chain is fundamental to understanding the relationships among and behaviors of each segment, as well as the opportunities for increasing productivity and profit across the chain of activities.



Figure 7 illustrates a standard value chain, or value system analysis and graph. This analysis is the quantitative building block for understanding the cascade of value added activities in a particular industry. By summarizing the price build-up for each activity in the chain, the graph tracks the distribution of net revenue for one unit of a good from raw material to point of sale. It is important to note that this graph does not incorporate costs at each activity, and so does not give any insight into the market economics and profitability of each activity. It is useful in mapping the roles of each segment and the incremental value that the market assigns each role based on additional inputs and services to the product.

Figure 7: Mongolia: Net Revenue per Kg of Cashmere for Each Component of the Value System



Source: Nathan Associates Inc. and J.E. Austin Associates, Inc

analysis is not a very precise tool, it is a good way to provide a general characterization of the current state of the industry, identify issues, and generate discussion. It is particularly useful as a neutral facilitation tool to focus an initial discussion on the perceived state of the value chain or to perform initial brainstorming on the potential opportunities and risks.

The example in Figure 7 depicts the build-up of the net revenue of one kilogram of Mongolian cashmere from herder to retailer. Over sixty percent of the value of the end product is captured by the knitting, wholesaling, and retailing activities.

SWOT Analysis

Complementing the value system analysis above, SWOT a (strengths, weaknesses. opportunities, and threats) analysis is the general, qualitative starting point for any competitiveness strategy or other analysis for decision-making purposes. While the SWOT

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

SWOT analysis is simple and can be used at differing levels of focus (examining a single firm, a segment of the chain, or the chain overall). It is also a good way of identifying areas to examine in greater detail. Thus, it is an excellent analytical starting point.

It is important to identify the right focus for the SWOT analysis and to keep the discussion on point. SWOT analysis is not an effective tool for identifying alternate strategies, nor is it terribly rigorous. It is, therefore, most valuable when supported by other forms of analysis rather than as a standalone tool.

Figure 8: Pakistan Dairy SWOT Analysis

Strengths Weaknesses + Large population and high percapita Consumers are price sensitive & ignorant of consumption of dairy products, ahead of other quality requirements - Significant cost variation in milk production regional markets in Asia + A high ratio of agricultural land to agricultural across sizes and types of farms po pulatio n Poor quality of milk A long chain of middlemen operating without + Large aggregate supply and breed potential of indigenous cattle/buffalo regulations leading to signific ant value + Regular culling of less productive / unproductive destruction animals The existence of the bandi system (under which the midd leman predetermines the + Internationally cost competitive farm types producer's sale price for the entire year at the already exist in Pakistan rate most beneficial to him) + Willingness of processors to develop infrastructure linking them to the farmer because of - Low animal productivity, no breed consumer preferences against reconstituted milk. management and little productivity support Larger and smaller processors are willing to invest services to farmers given the right support + Central geographic location with easy access to large, potential regional markets Farmers notorganized, general lack of able management structures in sector Poor quality of animal health care and breeding services. Support structure and industry is not well developed to support animal productivity. Opportunities Seasonality leading to fluctuations and umertainty in feed supply and quality Consumption can be increased with the right emphasis on nutritional value, quality and price. Capitalize on traditional nutritional value placed in Under-developed cold chain to consumers milk and related products Poor physic al infrastructure TExploit potential to improve animal yield through Small-scale and fragmented animal holdings better feed availability and breeding. Potential to do Lack of remunerative producer price for milk (maintain/establish) feasible larger farms. Interest - Low utilization of installed capacity of dairy in investing in larger model farms. plants [↑]Induce productivity & quality specific farmer Lack of a well-defined national policy for cooperation around collection and possibly milking d airy development and support efficient collection at larger scales. [↑]Develop extension services model based on NRSP Threats × Farmer cooperation around collection and to promote farmer cooperation [↑]Encourage large farm types as well as give central milking is expensive and difficult to manage incentive to subsistence farmer to invest in productivity ×Lack of breed management at the macro level and low potential of indigenous animals †Emergence of commercial dairy farms on a large × Unregulated imports of dairy products & scale [↑]Large aggregate production of milk. Develop rural additives and urban markets to sell milk and dairy products × Vested interests in perpetuating the from adjacent milk pockets dependence on imports of dairy commodities †Joint ventures based large potential and access to regio nal markets [†]Government interest and willingness to contribute †Build on consensus to develop and implement quality stand ards Consolidation and improvement in retail segment in large cities

Source: J.E. Austin Associates, Inc

Once a broad overview of the structure and characteristics of the value chain has been established, a more in-depth understanding of the drivers of productivity is required to uncover the highest value strategies to raise productivity in the value chain.

Using the Competitiveness Diamond

The Competitiveness Diamond⁴⁰ is a more rigorous analytical tool for evaluating an industry or value chain. The Competitiveness Diamond framework has been validated by numerous analytical and case studies and is now used by industries and governments worldwide to assess industry cluster competitiveness and to develop strategies for improving competitiveness. It is structured around four pillars:

- 1. Factor (input) conditions: skilled labor, infrastructure, etc.
- 2. Demand conditions: size and type of accessible demand
- 3. Related/supporting industries: presence of supplier and supporting industries
- 4. Context for firm strategy and rivalry: conditions for conducting business

Within each sector and across the economy, these four pillars interact to form a foundation for building a competitive system of firms. For developing economies like those in Africa, building a competitive value chain starts with understanding the demand for products and services in key markets and then organizing and investing in the factor inputs (supply) to best meet the needs of those markets. Only by accessing and understanding the specific needs of the market can the commercial sector utilize its resources in the most efficient way that creates the most value for both the producer and the customer. As a more in-depth understanding of the market is developed and the supply is upgraded accordingly, the value chain then must concentrate on upgrading its suppliers and supporting industries and working with the government to improve its business environment.

In contrast to the SWOT analysis, the Competitiveness Diamond is an analytical tool that poses very specific questions to the economic leadership about a specific industry. It can also be used as a framework for evaluating an overall regional or national economy.

The following diagram outlines many of the key questions to be posed when conducting a diamond analysis:

⁴⁰ Often referred to as the "Porter Diamond", The Diamond and many of its applications were developed by Michael Porter of the Harvard Business School.

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

Figure 9: Competitiveness Diamond Analysis—Key Questions



Once a general understanding of the competitiveness of a value chain has been achieved, value chain analysis must then delve more deeply to highlight the underlying drivers of competitiveness: strategic productivity, operational productivity, supply chain management, human capital, and the business environment. Only at this level of assessment is it possible to uncover the investment and actions required to increase the performance of the chain in regional and international markets.

Assessing Strategic Productivity

The primary means of increasing value chain productivity is by choosing appropriate markets and products and by adding new product features and service components that enable companies to increase the value of their product and, hence, the price. This "strategic" or "market pricerelated" productivity can be as important as "operational" productivity in determining competitiveness. In practice, understanding the strategic position of a firm or value chain in key markets helps to orient and focus investment on the highest value activities. By (re)positioning products in new markets or to new customers or by adding additional value (real and perceived), companies increase their sales and profits per unit of capital and labor. (Positioning is further discussed in Tool 8.)

When evaluating a value chain's strategic productivity, the key questions to consider are:

- Which product segments are currently being offered by the companies in the chain? Do they represent the full range of segments that could be offered? Do they represent the highest value segments in the industry?
- Which markets are currently being served by the companies in the chain? Do they represent the highest value markets?
- Of the spectrum of potential customers in each key market for the chain, which customers is the chain best positioned to target?

A number of analytical tools can be employed in answering the questions above:

- Market Segmentation and Assessment
- Customer Profiling
- Customer Benefit Analysis (product attribute gap analysis based on customer perceptions)
- Mapping the consumption chain (from sale to point of consumption)

Assessing Operational Productivity

A second way to improve value chain productivity is through operational productivity improved technology, manufacturing, and service processes within specific segments of the value chain. Operational productivity focuses on the efficiency of each appropriate segment of activities on its own. By introducing new technology that improves processes and management systems, key players in the value chain can lower their costs and raise the productivity of their businesses and the value chain overall.

The following questions are useful in evaluating the operational productivity of a specific segment of the value chain and developing strategies to enhance it:

- How do our costs compare to the price for the product in different markets (i.e., which markets are profitable to serve based on the current cost structure)? Are we excluded from competitively serving key markets because of our cost structure?
- How do the costs of the value chain compare to other competing value chains?
- What are the key trade-offs between cost and quality for the product?
- Who is in control of the cost drivers?
- What are the opportunities for lowering costs without compromising quality?

Analytical tools that can assist in determining the answers include:

- Per-unit economic cost-driver analysis (fixed vs. variable costs within each activity)
- Cost trend analysis (historical and projected changes in cost drivers)
- Per-unit activity cost benchmarking

Assessing the Quality of Supply Chain Management

Focusing on supply chain management in terms of costs of raw materials, transportation logistics, communications, and information technology – aspects of the chain that have generated great efficiencies in manufacturing, retailing, and other industries – is a third means to fully understanding the underlying drivers of competitiveness. Supply chain management is different from operational productivity. It focuses on the flow of goods and information along the chain of activities, the efficiencies of these flows, the transactions that facilitate value added, and the economic relationships that underlie each set of transactions.

Questions to consider when evaluating the supply chain include:

- Is the relationship among buyers and suppliers in the value chain cooperative or adversarial?
- How effective is the flow of information along the value chain (market trends, changes in price, external cost pressures)? How aware are the producers of the downstream market dynamics of the industry (market trends, demand conditions, pricing)?
- How sensitive is the overall cost structure to the cost of raw materials?
- How do logistics services affect the cost of raw materials and intermediary products?
- What is the availability of supporting services (financial, logistics, administrative) across activities in the value chain?
- How long does it take for a product to go from initial production to end market? How does this compare with competing value chains?

Analytical tools that can assist in determining the answers include:

- Supply chain analysis:
 - Distribution of players across chain activities
 - o Farm-to-market mapping: number of transactions and time required
- Benchmarking other supply chains
- Mapping of supporting services across the supply chain
- Mapping of information flows along the chain
- Supplier assessments

Assessing Human Resources Across the Value Chain

The next driver of value chain productivity is the quality of human resources available for the chain to tap.⁴¹ Thus, a fourth way to improve productivity is by investing in human resources. This may involve enhancing motivation, management, and training at the firm level, both by upgrading the overall educational system and through utilizing specialized institutes.

Improving the overall quality of the workforce is often seen as the mandate of government and the educational institutions, outside of the direct control of industry. Similarly, specific technical and management skills are viewed as the concern of specific firms or industries. Neither perception is accurate. Upgrading the technical and management skills of an entire value chain requires close cooperation of the firms along the chain and the supporting government and academic institutions.

The following questions can be considered when assessing the level of human resources across the spectrum of activities in a value chain:

- What incentives are present to encourage firms to invest in the technical and management skills of their employees?
- Do firms have difficulty retaining trained talent?
- What supporting educational services are available to firms locally to increase the skill levels of their staff? (industry certifications, IT training, technology application)
- How well do the academic institutions know the needs of industry? Is the curriculum aligned with the specific skills requirements of the industry?
- Are there industry standards for industry skills levels? Do the academic institutions teach to these standards?
- How do the skills levels of the value chain's workforce compare to competitors along key skills categories?

Analytical tools that can assist in determining the answers include:

- Industry workforce assessments to evaluate the demand- and supply-side conditions for labor in the industry and across the labor market in general
- Skills segmentation, classification, and benchmarking analyses for the value chain against competitors
- Bridge analyses⁴² to assess the strength of linkages between industry and academic institutions

⁴¹ Many tools and methodologies have already been developed to evaluate and address this area. A detailed consideration of workforce issues is largely beyond the scope of this Guide. Perhaps they can be addressed more fully in a subsequent volume. For more detailed information on workforce development, readers are directed to the extensive body of literature available publicly on the World Bank website.
⁴² Derived from J.E. Austin Associates, Inc.'s "10 Bridges Approach," a methodology for building workforce

⁴² Derived from J.E. Austin Associates, Inc.'s "10 Bridges Approach," a methodology for building workforce competitiveness through university-industry collaborations.

Assessing the Business Environment

The quality of the business environment ultimately serves a gate-keeping function, and often a negative one. Productivity increases are achieved at the firm and value chain levels by improving the quality of business strategy and operations – for example, by forming new partnerships with international firms in their value chains that provide access to markets, technology, finance, and know-how. Interventions designed to enhance productivity will differ by value chain and will require a different balance across the four components (strategic productivity, operational productivity, supply chain management, and human resources) discussed above. However, a factor common across all value chains and components is that the business environment contributes importantly to the ability of firms to succeed in enhancing productivity. These "business environment" factors, also called the micro-economic foundations of growth are perhaps especially important in the context of Africa:⁴³ they include the level of bureaucracy and red tape; the extent to which the rule of law is enforced and commercial courts are functional; the quality of infrastructure; the level of financial sector modernization and regulation; levels of trade access; and the ability of the country to attract foreign investment and to ensure the proper functioning of land markets (through reliable registries) and labor markets (through policies that encourage job creation, labor flexibility, and overall productivity).

Numerous methodologies and analytical tools exist to evaluate the business environment of a particular value chain.⁴⁴ In most cases, the analytical tools are targeted at a specific area of the business environment. The following are a few general tools that can be used:

- Regulatory impact analysis
- Mapping of the time and cost of bureaucratic processes
- Per-unit costing of rent-seeking activities across the value chain
- Total annual costing of weaknesses in infrastructure relevant to the industry

Means by which businesses can participate in improving the business environment are described in Tools 10 and 11.

A Step-by-Step Summary of Tool 2: Designing Informed Strategies Across the Value Chain

• Conduct a value chain product unit financial analysis and SWOT analysis. First, understand the distribution and flow of resources and profits along the chain. Collect key data at each segment of the value chain, and graph the unit costs, unit revenue, and unit profit for each step. To complement the financial and operational data, conduct a SWOT analysis in a participatory manner with value chain leadership to qualitatively characterize the state of the value chain.

 ⁴³ Dr. Michael Porter, the thought leader of business competitiveness, weights his "Business Competitiveness Index" in the influential World Economic Forum Global Competitiveness Report with 80 percent on the national business environment and 20 percent on "the quality of business strategy and operations."
 ⁴⁴ Annual reports on the conditions of the business environment in specific countries such as the World Bank Doing

⁴⁴ Annual reports on the conditions of the business environment in specific countries such as the World Bank Doing Business and the Global Competitiveness Report provide useful examples as they use many of the survey and analytical tools at the economy level.

- Conduct a Competitiveness Diamond analysis. For the value chain, assess the strengths and weaknesses of each of the four pillars of the competitiveness diamond: 1) demand conditions 2) factor inputs 3) context for firm strategy and rivalry 4) related and supporting industries.
- Assess strategic productivity of the value chain. Use analytical tools such as market segmentation and assessment, customer profiling, customer benefit analysis, and consumption chain mapping to understand the current strategic positioning of the value chain, as well as potential markets and product offerings.
- Assess operational productivity of the value chain. A variety of tools are available to analyze the operational productivity and performance of the value chain including per-unit economic cost-driver analysis (fixed vs. variable costs within each activity), cost trend analysis, and per-unit activity cost benchmarking.
- Assess the quality of supply chain management. To gauge the efficiency in the flow of goods along the chain, the following tools can be useful: Farm-to-market mapping (# of transactions and time) and distribution of players across the chain, supply chain benchmarking, mapping of supporting services, mapping of information flow along the chain, and supplier assessments.
- Assess the human resources across the value chain. To assess the quality of the value chain workforce, the following tools can be used: Demand- and supply-side analyses of the labor market for the value chain, skills segmentation and benchmarking analyses with competitor value chains (including labor productivity, # of skilled graduates), or the 10 Bridges analysis of the linkages between industry and academia.
- Assess the business environment. Numerous tools are already well developed to assess the business environment of a value chain. Among the most tested are regulatory impact analyses, mapping of time and cost of bureaucratic processes at each stage of the value chain, per-unit costing of rent-seeking activities along each segment of the value chain, and total annual cost of weaknesses in infrastructure.

Case Study: Understanding the Value Chain and Integrating Information Into Strategy - Nigerian Domestic Catfish⁴⁵

Introduction

Nigeria is the fifth largest aquaculture producer in the world. In Nigeria, domestic fish is a preferred protein that rivals red meat in consumer demand. Domestic demand for fresh catfish has grown as Nigerian incomes increase. Nigeria has become one of the largest importers of fish in the developing world, bringing in some 600,000 metric tons (mt) annually. There is a growing awareness of aquaculture in Nigeria, with more than 100 private commercial fish farms currently in production.

The Government has sponsored major investments in the infrastructure of some 40 additional public fish farms and hatcheries, but most of these are not operating at present. Aquaculture expansion has been a slow process, as private sector fish farmers have faced major constraints, including lack of supporting feed/seed companies.

At the New Partnership for Africa's Development (NEPAD) 2005 "Fish for All Summit" gathering, Nigeria's President Olusegun Obasanjo stated that, "if Africa's per capita consumption of fish is just to be

Figure 10: Map of Nigeria



Source: CIA World Factbook 2007

kept at its present level, though grossly low and unacceptable, then fish production must be increased by over 250 percent by 2015. This unhealthy situation calls for urgent action and indeed poses a great challenge to all of us." President Obasanjo further remarked in his closing address that although fishing brings Africa export earnings of USD 2.7 billion annually, "these benefits are at risk as the exploitation of African natural fish stocks is reaching its limits, and aquaculture production has not realized its full potential."⁴⁶

The Tool: Market Assessments

Value chain participants must consult the market when determining how to increase profits, reach new markets, innovate, and increase productivity. There are seven widely accepted means of developing market analysis that contribute to identifying opportunities to capture more value. They are detailed in the Market Assessment Methods section below.⁴⁷

In Nigeria, a market survey was used to gather quantitative and qualitative information about buyers, sellers, volumes, prices, market trends, market share, and market segments, along with qualitative information on competitors. The rationale behind this approach was that surveys can accurately represent the opportunities and demand characteristics of the markets studied. Value chain participants can use surveys to identifying broad problems in a market, such as lack of knowledge about a particular service, limited understanding of its benefits, or failure to assign appropriate value to outside assistance. This type of information is useful for establishing new products or services within the value chain.

http://www.seepnetwork.org/files/2991_file_TechNote4_c.pdf

⁴⁵ Authors: Michael Gorman and Martin Webber - J.E. Austin Associates

⁴⁶ Sources: EurekNews and BBC News (<u>http://news.bbc.co.uk/2/hi/science/nature/4187522.stm</u>

⁴⁷ Engelmann and Swisscontact "PLP Technical Note #4: An Inventory of BDS Market Assessment Methods for Programs Targeting Microenterprises" Swisscontact, June 2005,

Background

Africa produced 7.31 million tons of fish and seafood in 2002.⁴⁸ On the continent, aquaculture development has been most notable in Egypt, where a combination of tradition, market demand, available and well-managed water resources, marketable species, and private sector initiative resulted in dramatic growth during the 1990s. The top three African producer nations were Egypt (accounting for 85.6 percent of the total), Nigeria (6.5 percent), and Madagascar (1.8 percent). Egypt's aquaculture growth and development has been the most significant as it has increased production levels from 85,000 tons in 1997 to 376,000 tons in 2002, a 35 percent average annual growth rate.

Market Assessment Methods⁴⁹

- Secondary Research refers to the use of data that have already been collected, analyzed, and made available for other purposes. VC participants may find such research useful for identifying sectors experiencing growth, as well as for understanding government regulations and policy.
- *Group Discussions* permit value chain participants to explore issues in general terms, then seek more specific information by using focus group discussions.
- *In-depth interviews* provide qualitative and quantitative information on the VC and are particularly useful for different participants of VCs to understand their complementary relationships.
- Market Observation can be used by VC participants to obtain qualitative and quantitative data from local markets on transactions, interactions, processes, and embedded services. Observations are also a simple tool to cross-check information obtained from other sources.
- Interactive Workshops or Meetings can be used to validate and deepen previously gathered information. They can generate ideas for addressing constraints or opportunities in the market. In addition, these workshops may lead to a common approach in solving market problems among all of the VC participants.
- Product Concept Testing aims to gauge the demand for a service or product that the VC does not yet produce and about which the VC customers have no knowledge.
- *Market and Consumer Surveys* are useful (as evidenced by the Nigeria case study) to obtain an accurate picture of serviceable aspects of the market for the product or service.

By 2006, farmed catfish accounted for approximately 50 percent of Nigeria's domestic annual fish production. The catfish industry provides approximately USD 75 million in revenues at the farm gate and accounts for nearly USD 180 million in consumer spending. The sector contributes to the employment of nearly 25,000 people, with the majority (over two-thirds) employed as restaurant workers. Aquaculture and Nigerian farm-raised catfish have also been identified as a growing source of income for Nigerian farmers.

⁴⁸ Dixie and Ohen. May 2006. The Market for Catfish and Other Aquaculture Products in Nigeria.

⁴⁹ Engelmann and Swisscontact

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

To ensure success, any new investments or expansion of production must be driven by market demand and consumer preferences. This case study features the information gathered from a domestic market assessment for the farmed and fresh catfish value chain conducted by Dixie and Ohen⁵⁰ in 2006 in order to demonstrate the value of market information in determining future investments and/or interventions in a value chain.

The Nigerian Catfish Value Chain⁵¹

The domestic market demand for fresh and smoked Nigerian catfish is outstripping the supply. Tastes and preferences for catfish have been increasing due in large part to the availability, cost, consumer preference for white meat, perceived health benefits over substitute products, and some popular catfish dishes served by restaurants.

Production

According to Dixie and Ohen, the value of Nigerian catfish produced and sold by aquaculture farmers is about USD 75 million, which assumes a production level of about 30,000 tons of catfish per year. At the retail level, this tonnage rises in value to about USD 120 million. Restaurant operators capture an additional USD 60 million dollars on catfish, for a total of USD 180 million in sector-wide consumer spending. Estimates indicate that approximately 70 percent of fresh fish in Nigeria is sold in restaurants, especially in *bukas*.⁵² The major cost of production for farmers is European catfish feed, which amounts to approximately USD 39 million. Nigeria currently relies on imports because a sizeable fish-feed industry has yet to develop. The majority of domestic feed supplies are farm-made with only a few animal feed millers providing domestic commercial feed Source: J.E. Austin Associates, Inc pellets.53

Figure 11: Nigerian Domestic Catfish Farming Value Chain



Supply

Farmed catfish are almost universally sold at the farm to primary wholesalers, retailers, or buka owners. As indicated in Figure 11, primary wholesalers have been chiefly responsible for developing the trade from the major fish farms in Nigeria. Traditionally, these wholesalers transport live catfish to secondary wholesalers who sell on to other retailers, restaurants, or consumers. It is worth noting that, in the major production areas, retailers and restaurants are increasingly buying directly at the farm, thus cutting out middlemen and vertically integrating. A new trend in the industry is that the majority of fresh catfish, especially larger fish, are now sold directly to bukas, restaurants, and hotels.

Demand

Dixie and Ohen cite Nigerian consumers' requirements and demands with respect to catfish and catfish products that are fairly common across regions and markets. As indicated in accompanying figure, key

⁵⁰ Dixie and Ohen, 2006.

⁵¹ Dixie and Ohen, 2006.

⁵² The word "buka" refers to a "canteen or eating house" in the local Hausa language.

⁵³ Moehl, J., M. Halwart and R. Brummet. 2005. "Report of the World Fish Center Workshop on Small-Scale Aquaculture in SSSA." FAO.

consumer motivations for consuming catfish were 1) availability, 2) taste/smell, and 3) familiarity. Over 90 percent of consumers said that they were increasing their purchases of catfish, and 15 percent identified

Figure 12: Consumers' Reasons for Purchasing Live Catfish



Source: Dixie & Ohen

asing their purchases of catfish, and 15 percent identified bird flu as affecting their purchases. Comments were made about the health benefits of white meat (e.g. catfish), and a significant number made reference to a perception that after a certain age (normally about 35), red meat should be given up entirely. Nearly 80 percent said that they would use catfish to prepare pepper soup—by far, the most popular use—and about 40 percent said it would go into other soups or stew dishes.

Opportunity

According to traders and retailers, product losses (fish death, quality loss) and finance are the most common challenges cited by fresh catfish traders and retailers. Fresh and living catfish command a retail price twice that of frozen fish. Traders and retailers do not use proper storage facilities, do not change the water sufficiently, and often leave the fish in the sun for long periods—all leading to product loss and lower quality, which translate to lower revenues. Traders indicate that limited finance

constrains the ability to develop and invest in business operations or expansion. Secondary issues are problems with debt recovery, market location, electricity (mainly for refrigeration), skin damage to the fish (which reduces the sale price), and transport costs.

Nigerian catfish farmers report that the marketing of fresh catfish is becoming more difficult because of increasing competition. Some farmers in the Southeast also recognize that the difficulties in selling fish result from a general lack of organization (e.g. farms all try to sell fish on the same day). To control marketing problems, farmers say that they would like to have direct contact with traders further down the marketing chain. Farmers have indicated that information sharing within the value chain is sparse and that there is frustration that primary wholesalers shield their sources of supply from secondary

Figure 13: Consumers' Views of What Factors Would Lift Their Consumption of Catfish



Source: Dixie & Ohen

wholesalers and retailers. In addition, in some locations, retailers feel that they have to pay higher prices than necessary because they are unable to buy direct, or at least to have better information about the selling prices at the farm.

The Nigerian catfish market projections provided by the surveyors imply that the undersupply of farmed catfish may amount to some 5,000 mt per year, which could be rapidly addressed by expanding production. The urban markets, where farmed fish is primarily sold, will grow by about 3 percent annually purely through population growth.

Once these sources of demand have been met, additional supplies will have to be absorbed by stimulating sales through lower retail prices, the addition of value-added operations, or exports. Fish farmers and their value chain colleagues will have to take a much more proactive approach to learning about the market and initiating sales

and marketing activities. This is likely to involve conducting their own market research and establishing closer contacts with retailers, traders, restaurants, and processors. These will be new activities for producers.

Interventions in the Nigerian Fresh Catfish Value Chain

The survey results point to many possible value chain initiatives. Unmet and clearly increasing market demand should be addressed by increasing supply from producers ("VC Possible Action #1" in Figure 14), taking advantage of the opportunities to vertically integrate operations ("VC Possible Action #2"), and developing more sophisticated marketing

and distribution systems in order to capture more value ("VC Possible Action #3"). There is also ample opportunity for producers to band together and gain a competitive advantage through economies of scale. Costly inputs can be purchased at significant savings with effective costshare agreements and in larger quantities. Also, associations of producers or the firms linked to them can combine to act in groups to attain the scale and market power necessary to create permanent marketing links and secure contracts to provide restaurants with guaranteed supply and quality.

These new distribution systems will have an opportunity to increase the marketing of catfish and improve the safe transport of products. The market assessment conducted by Dixie and Ohen confirms that consumers prefer catfish based on its purported health qualities, its cost, and its



availability, as well as enthusiasm for its taste and preparation in popular dishes and as a complimentary product to beer. There is an opportunity to capitalize on these preferences in marketing initiatives while





Source: J.E. Austin Associates Inc.

Figure 14: Nigerian Catfish Farming Value Chain – Possible Actions

simultaneously expanding the market supply.

Currently, losses during transportation are estimated to be between 10 to 15 percent of value, with practices varying widely across the sector. With better organization, appropriate post-harvest methods suited to Nigerian conditions can be propagated to the retail, trading, and restaurant sectors. Producers or wholesalers should seek technical expertise and assistance in order to lower these loss rates and improve the quality of the catfish that they distribute.

Financing is still not widely available for restaurants and

bukas to increase their working capital and expand their businesses. With growing demand and falling prices, there is an opportunity to provide financing to promising end-market businesses that may pull the value chain towards higher growth vectors.

Case Study: Understanding the Value Chain and Integrating Information Into Strategy

Conclusion

The market assessment survey has furnished important information that allows the value chain actors to identify growth opportunities and priority interventions. The Nigerian domestic catfish market is clearly rich with opportunity for increasing sales; there are numerous constraints and other issues that can be addressed, and there is room for value chain collaboration centered on production, investment in transport and infrastructure, marketing, commercialization, and vertical integration.

Tool 3: Conducting Benchmarking and Gap Assessments of Value Chains

Benchmarking Background

At both the company and value chain levels, benchmarking is crucial. Benchmarking compares the performance of a company or value chain to itself at different points in time, to another value chain in the country, or to a value chain in another country in order to establish the current baseline position and provide comparative data to guide decisions and actions. Usually quantitative indicators are used, such as total GDP, time to market, pricing data, etc. Qualitative data can also be used, although such information is harder to measure clearly and objectively.

Benchmarking allows practitioners and stakeholders to understand the performance of particular value chains in comparison to competitors, and especially to global best performers. Knowing what comparators do differently, and whether these differences are important drivers of value chain performance, gives clues as to beneficial changes that could be made to improve performance of the value chain.

Benchmarking can be used by all members of the value chain to build a common understanding and vision as the basis for prioritizing objectives and decisions. Members can benchmark against each other to determine whether their performance is up to par and pinpoint areas of improvement. They can also benchmark the entire value chain against other value chains in the same industry or other global value chains.

The Benchmarking Tool

It is possible to benchmark almost anything, so, too often, stakeholders do not adequately narrow down their field of focus when benchmarking. A scope that is too broad does not allow for real drivers of performance to be analyzed. Therefore, it is necessary to first refine the scope and select appropriate comparators. Those conducting the benchmarking must keep in mind that the results should lead to clear courses of action. Therefore, if the comparators that are selected are not appropriate, the results will not be useful and may even be misleading.

In assessing a value chain's competitiveness, several key aspects can be usefully examined. First, it is important to benchmark overall value chain performance, meaning how well the actors in the value chain deliver products to the final consumer in comparison to other value chains and other countries. Once the overall value chain is analyzed, it can be broken down into key performance components, such as yields, transport efficiency, market access, unit price, and many others. These key components and the underlying processes can then be individually benchmarked to identify relative strengths and weaknesses. Industry experts can identify important, detailed performance and competitiveness indicators; once these indicators are quantified, comparisons between different industries and countries can be made. The experts and the value chain leadership can then analyze how the better performance is being achieved.

Once benchmarking is completed and differences in performance have been identified and analyzed, the next step is to determine courses of action. In order to spur action by stakeholders, the results of the benchmarking exercise can be disseminated so that stakeholders understand their position and become a part of the action process. The information should generate dialogue aimed at increasing participants' insights into the need for intervention and achieving consensus on decisions and actions.

What Benchmarking Does

- 1. Establishes a Baseline of Current Performance
 - a. Once the benchmarking exercise has been completed, a company, an industry, or a country understands its position relative to its comparators.
- 2. Identifies Areas and Targets for Improvement
 - a. If benchmarking against best practices, then differences in performance along the entire value can be identified. Shortcomings can be examined steps can be taken to improve performance.
- 3. Pinpoints Potential Critical Factors for Success
 - a. Areas for improvement can be prioritized.

Steps to Effective Benchmarking

The benchmarking process is straightforward in nature. It generally includes the following steps:

- Determine what indicators and measures to benchmark
- Determine the benchmarking target groups
- Gather and analyze the data
- Convert benchmark data into action through:
 - o Managed Discussion
 - Prioritization and design
 - o Implementation

Many widely available indexes measure elements of one country's competitiveness relative to other countries. Examples involve the World Economic Forum's Global Competitiveness Report, the World Bank Doing Business Report, the World Development Indicators, and a variety of standard economic indicators available through governments, universities, and international organizations. Although the data they provide can be general in nature, such data are a useful guide for specific discussions and action planning later in the benchmarking process.

In addition to the "broad strokes," decision-makers and industry leaders need to understand the more detailed basis for the value chain's performance, the structure of the value chain, availability of services, and the value chain's operating environment. This targeted benchmarking is done in several steps, outlined below.

Determine What to Benchmark

First, the value chain's performance must be accurately and objectively measured in the performance and operational areas most relevant to its competitiveness (e.g., availability of inputs, time to export in national ports, and local transportation costs). Items that should be

Using Value Chain Approaches in Agribusiness and Agriculture in Sub-Saharan Africa

benchmarked will vary from value chain to value chain based on priorities determined by the value chain's strategy. It is important to select actionable items for a benchmarking exercise. It is not sufficient just to know that some process or step in the value chain is slow or costly. Selected indicators need to point to why the process is slow or exactly where costs are added. This should be measured at steps all along the value chain and in the services provided to the chain. It is important that the entire value chain be included so as to identify bottlenecks and added costs that may not be readily apparent.

The rationale for using data, and the data points themselves, can be gathered from industry and trade publications, local government sources, and through surveys. Surveying relevant firms and industry groups, and participants along the entire value chain, can provide a qualitative component that enhances understanding of the processes being measured and costs being tracked. How to conduct value chain performance gap analysis is discussed later in this section. The greater the level of specificity in the benchmarking activities, the more likely it is that the results will lead to practical applications.

Box 2: Uganda Benchmarking Constraints in the Coffee Industry

Uganda, a major coffee producer, has only recently begun a concerted campaign to add value to their production. Grown in many areas of the country, Uganda's coffee is generally transported to Kampala and sent to Mombassa Port for shipment overseas.

In 2003, amidst a national decentralization initiative, coffee growers and processors were faced with increasing numbers of procedures as individual districts imposed levies on investments and shipments within and between the districts. Procedures were not combined or streamlined—indeed, the increasing fragmentation led to the need to devote substantial time to petty transactions and, reportedly, an increased incidence of "facilitation payments." These circumstances were echoed in other sectors, such as fish and wood products.

Reports indicated that once a shipment reached Kampala, it was not uncommon for the container to take a period of 20 days to reach Mombassa—but only two or fewer days are required for actual transport time. What accounted for the remaining 18 days? Lengthy and non-streamlined border procedures accounted for some of the time, but most was taken up with multiple inspections and customs procedures. Reports also implied that numerous informal taxes were being levied. The impact on the industry in terms of product quality, losses, and the risk of missed deliveries was significant.

Ugandan stakeholders benchmarked this situation against those in other countries in order to see if these delays and costs were normal. Information from coffee exporters in Colombia, Costa Rica, and Vietnam indicated that instead of 20 days, the norms for delivery to port were between one and seven days!

This information helped the industry and government to recognize the impact of a poor system of regulation and implementation and to focus attention on putting a streamlined system in place.

Source: J.E. Austin Associates, Inc. Interviews Author: Lisa Carse – J.E. Austin Associates

Determine the Benchmarking Target(s)

Benchmarkers must determine comparator value chains (target comparators) and specific performance indicators that will provide the best basis for comparison. The target comparators should be value chains operating in other countries, or possibly in different regions of the same country. Once the home value chain data are obtained, the benchmarkers can turn their attention to the targets.

Including a knowledgeable outsider can facilitate the process, provide additional perspective, strengthen the objectivity of the data, and speed identification of key elements to benchmark. In choosing benchmarking targets, it is important not only to consider countries and industries that demonstrate global best practices but also regional competitors or those countries that are operating successfully in a competitive space that the home value chain would like to occupy (see Tool 8). The purpose of the benchmarking exercise within the context of the goals and strategy of the nation and/or value chain will determine which comparator indicators are examined; the key point is that the indicator should have high relevance as a standard or driver of performance. Objectives of the benchmarking might include:

- Operational improvements
- Reduced transaction costs
- More advantageous market positioning

Collecting the Data—From Where is Benchmarking Information Obtained?

A surprising number of public sources will provide useful information; sometimes, the benchmarkers will have to dig more deeply. Some common sources include:

- Publicly available reports
- Published sources, especially trade publications and databases
- Asking the target comparator value chain (and participant companies and clusters) directly
- Industry experts, consultants, and researchers
- Asking suppliers, service providers, and buyers

Box 3: Tanzania Cotton–Benchmarking Costs

Cotton is a primary commodity in several West African countries. A field visit to Tanzania in 2004 examined Tanzanian ginning, benchmarking companies against a theoretical West African cotton company and against actual costs in Burkina Faso and Cameroon. The theoretical costs were calculated for a West African cotton company operating at full capacity (with a volume of 50,000 tons of seed cotton) with the buy price of seed cotton at a level of FCFA 160 per kg, which is the actual price in Burkina Faso, Cameroon, and Mali.

Below, the price breakdown of lint cotton production is set against comparator countries, including the West Africa theoretical model.

Comparison Between Tanzania, Theoretical Costs in West Africa, and Actual Costs in Burkina Faso and Cameroon

	USD/kg of lint cotton				
	Tanzania	W. Africa theoretical	Actual Burkina	Actual Cameroon	
Collection of seed cotton	0.094	0.083	0.103	0.097	
Processing costs	0.082	0.135	0.163	0.134	
Financing costs (short term)	0.023	0.038	0.067	0.013	
Cost from ginnery to FOB	0.100	0.128	0.155	0.162	
Sub-total	0.299	0.383	0.488	0.406	
Capital costs (on investment)	0.009	0.035	0.036	0.036	
---	-------	-------	-------	-------	
Taxes	0.042				
Overhead and contingencies	0.009	0.052	0.034	0.053	
Dagris** fee			0.012	0.012	
Total intermediary costs	0.358	0.470	0.569	0.507	
Purchase cost of seed cotton	0.833	0.755	0.755	0.755	
CDF levy (passbook)	0.055				
Critical functions (extension, research, seeds)		0.019	0.029	0.047	
Total FOB cost	1.247	1.244	1.352	1.308	
Minus: value of seeds	0.079	0.038	0.038	0.050	
Net FOB cost	1.168	1.206	1.315	1.259	

Tanzania has a cost advantage over the theoretical figure and the actual prices from Burkina Faso and Cameroon. Knowing this allows Tanzania to develop an industry growth strategy to take advantage of its favorable cost position. Tanzania's net FOB costs are 13 percent lower than Burkina Faso's and 85 percent lower than Cameroon's.

*The comparison, in US dollars, is based on the actual exchange rate (USD 1 = FCFA 504 and TZS 1200) at the time of writing the report in 2004.

**Dagris is a company involved in cotton marketing chains that holds shares in cotton enterprises in several African countries.

Author: Lisa Carse - J.E. Austin Associates

Utilizing the Data

Once the data have been gathered, participants can carry out a careful analysis and draw insightful conclusions. Many analytical tools can be used to create clear comparisons and help identify areas for targeted intervention. Creating a graphical representation, such as the "spider" chart below, can help decision-makers to easily understand certain strengths and weaknesses in their value chains so that actions can quickly be targeted in the appropriate areas.

Figure 16 below describes the extent of coordination within the supply chains of citrus fruits and tomatoes in Spain, Morocco, and Turkey.⁵⁴ In order to achieve EurepGAP certification, it is important for all members of the value chain to be in close communication so that quality and other elements required for certification are ensured. Morocco and Turkey were benchmarked against Spain in categories that contribute to seamless coordination along the supply chain. From the graph below, it is apparent that Spain is the best performer, followed by Morocco and Turkey. Turkey falls short of Spain in every category, while Morocco actually matches Spain's score in several categories. Illustrating this information in this type of graph helps value chain participants to weigh their options for which improvements to address first. In this case, Turkey needs to make improvements in every category, while Morocco can choose whether to focus on areas where it falls far short of Spain (such as market orientation, production flexibility, vertical integration, vertical coordination, and IT systems), or attempt to gain a competitive advantage by outperforming Spain in areas where it is already close to achieving that goal (such as in segregation or traceability systems).

⁵⁴ Garcia, M. 2003. "Benchmarking Safety and Quality Management Practices in the Mediterranean Fresh Produce Export Sector." Project Report. Department of Agricultural Sciences, Imperial College, London.



Figure 16: Morocco-Coordination Within the Citrus Fruit and Tomato Value Chains, Comparison with Turkey and Spain

Note: Higher value indicates better quality Source: Garcia, 2003

Gap analysis is a basic tool that is useful in understanding differences among comparator value chains, and helping value chain participants to identify the areas where interventions and reforms should take place. This analytical tool, when presented graphically as a comparison of elements of value chains (or clusters—see Tool 12 for comparison of the content of cluster maps) can provide strong visual impact in understanding and communicating a value chain's relative strengths and weaknesses.

Gap analysis can also be presented in a table format, as shown in Table 8 using data gathered in a benchmarking exercise conducted for the cigar industry in the Dominican Republic.

Critical Success Factors	Dominican Cigars	Cuban Cigars	Follow-on Questions
Sales Volume	120 million sold	80 million sold	At what price point? What are industry profits?
Flavor	#2 in blind taste tests	#1 in blind taste tests	What are the key determinants of flavor?
Packaging	Packaging Imported wrapper Local wrapper		How important is the wrapper to consumer choice? How does the wrapper affect production costs?
Research and Development Capacity	Weak (but improving)	Strong	What institutions are needed to develop R&D capacity?
Distribution Channels	Mostly sells to Davidoff, etc.	Controls European distribution channels	What kinds of distribution channels are most in line with the business and growth model? How can these be developed?
Final Market	Final MarketOver-reliance on U.S. embargo of CubaStrong European penetration		Where are current customers? Future/potential consumer bases?
Industry Management	Dynamic enterprises	State-owned enterprises	What are managerial weaknesses? How can they be improved?
Marketing	Rising image as a "cigar country"	Strong "Cuban" brand	How to develop an effective and differentiated branding strategy?

Table 8: Gap Analysis of the Dominican Cigar Industry vs. Cuban Cigars

Source: J.E. Austin Associates, Inc.

Gap analysis can also be based on the perceptions of value chain leaders through an exercise that quantifies what members of the value chain already know. In the illustrative example below, members of the value chain use a qualitative benchmarking exercise to begin to identify both the specific areas in which they believe the chain lags behind its competitors and ideal models or industry trends. This "quick and dirty" approach to gap analysis is based more on leaders' perceptions (correct or incorrect) than on hard data. But it is a useful means of quickly starting a benchmarking dialogue and a good strategist/facilitator can use the method and discussion to encourage the value chain participants to look more deeply into the assumptions and conclusions.

In the example illustrated in Table 9, value chain A is the country of interest, while value chain B is the best practices target or global industry standard. The participants identify the indicators that they believe are the current drivers of their industry's global competitiveness. Members of the value chain in country A then give themselves scores for how well the chain performs on each indicator. This score is compared to the best practices score, and the gap between the two scores is recorded.

Driving Forces of Industry	Country A	Country B	Gap
Health qualities	1	5	4
Branding and	2	5	3
promotion			
Convenience of	4	5	1
packaging and			
availability			
Guarantee of quality	3	5	3
and standards			
Logistics	2	5	3
Rating: 5=excellent:	4=good: 3=average: 2	=below average: 1=w	eak or poor.

Table 9: Illustrative Gap Analysis

Source: J.E. Austin Associates, Inc.

Once the participants recognize the gaps and understand the reasons for them, they will be able to make choices for a prioritized strategy change. The gap analysis and the proposed action prioritization can be used as an effective basis for public/private dialogue along with value chain and firm-level decision-making (for example, see the cases in Tools 2 and 8). However, because it is unlikely that value chain actors will be able to effectively manage everything at once, decision-makers will need to weigh the feasibility of possible initiatives, their implementation capacity, and the relative payoff of each intervention in determining priorities. Value chain stakeholders will initially need to target high impact gaps and objectives or ones that establish a platform for follow-on steps and successes.

A Step-by-Step Summary of Tool 3: Conducting Benchmarking and Gap Assessments of Value Chains

- Determine which value chain performance indicators and measures to benchmark. Identify the elements that are important. The indicators should point to where costs, time, quality, service, and value are added along the value chain to help identify bottlenecks, unnecessary costs, and upgrading opportunities.
- Determine the benchmarking target groups (comparators). Value chains and companies that represent global best practice, as well as regional value chains, should be compared against to identify areas of operational improvement, ways to reduce transaction costs, quality and service improvements, availability and quality of supporting services, and ways to improve competitiveness strategies.
- **Identify data sources.** There may be many sources, including studies and research, industry reports, trade publications, and other available material. Knowledgeable experts (within the industry and within comparator value chains and organizations), buyers, and service providers can be consulted. First-hand visits to observe operations may be warranted.
- Gather and analyze the data. A variety of tools are available to analyze and communicate benchmark data. These tools include, but are not limited to, GAP analysis, spider graphs, diamond analysis, and illustrative tables.

• **Convert benchmark data into strategy and action.** Accurate data from benchmarking exercises provide the private sector, practitioners, and policy-makers with relevant indicators highlighting the strengths and weaknesses of any value chain. These data can then be used as a springboard for discussions among stakeholders about which areas of a value chain should be priorities for intervention.

Case Study: Uganda Floriculture - Benchmarking and Gap Analysis⁵⁵

Introduction

Benchmarking and gap analysis serve to establish levels of performance vis-à-vis comparator industries, value chains, or firms. In 2002, the World Bank's "Regional Study on Agricultural Trade Facilitation/Non-Traditional Export Promotion in SSA: Uganda Horti-floriculture Sector Technical Note 2" benchmarked the cost structure and other elements of Uganda's floriculture industry against those of Kenya, a country with similar natural attributes, ranked number one in African floriculture exports to the Dutch auctions. Therefore, Uganda was not only benchmarked against its main regional competitor; it was also benchmarked against Africa's best performer in terms of total export value. The benchmarking was used to inform a gap analysis-the identification of areas of weakness relative to comparators-and create a strategy for future growth based on Uganda's competitive position.

Background and Industry Description



Figure 17: Map of Uganda

100 mi

Arua

Alher

Masaka,

Mbarara

Fort Portal

DEM. REP

OF THE CONGO

RWANDA

SUDAN

Lira

Bell

Lake

Moroto.

KENY/

Soroti

Mbale.

Jinia

Gulu

KAMPALA Port

Entebbe.

TANZANIA

Uganda, a landlocked country in East Africa, has achieved Source: CIA World Factbook 2007. considerable diversification from traditional agricultural crops such as coffee, tea, and cotton, to nontraditional exports including fish, tobacco, cut flowers, and a wide variety of vegetables. The push towards diversification of the economy

was particularly important in

response to falling prices for traditional goods.

Floriculture was first developed around 1993, so it is a relatively new industry for Uganda. (Floriculture specifically describes the cultivation of flowers and cuttings. which currently comprises a majority of Uganda's horticultural production.) The World Bank's 2002 assistance in studying the Ugandan floriculture sector was intended to help the industry and GoU assess arowth potential and design strategy for its а development.

Figure 18: Uganda Floriculture VC & Cluster Map



Source: J.E. Austin Associates, Inc. The value of Uganda's

horticultural exports,

comprising flowers, plant cuttings, fresh fruits and vegetables, vanilla, cocoa and papain, nearly guadrupled from USD 10.7 million in 1995 to USD 40.7 million in 2002 (see Table 10). Floriculture exports

⁵⁵ Authors: Lisa Carse and Martin Webber - J.E. Austin Associates

represented the largest share of Uganda's exports in the horticultural sector. However, experts estimated that Uganda was producing far below its potential.

Product	Value of exports (USD	Value of exports (USD	Total growth (%)	Annual growth (%)	Annual growth (%)	Annual growth (%)	Average annual growth
	'000 fob) 1995	'000 fob) 2002	1995- 2002	1999-2000	2000-2001	2001-2002	2000-2002
Roses	2.3	14.1	513.04	-8.34	26.32	22.40	13.46
Plant							
cuttings	0	7.03		28.21	24.44	25.54	26.06
Fresh							
produce	0.63	4.24	573.02	0.96	1.27	32.50	11.57
Vanilla	0.24	9.43	3829.17	34.67	183.17	64.86	94.23
Cocoa	0.64	4.97	676.56	-27.86	33.66	84.07	29.96
Papain	4.46	0.71	-84.08	-76.67	-26.53	-1.39	-34.86
Other	2.4	0.2	-91.67	-28.57	20.00	-66.67	-25.08
Total	10.67	40.68					

Source: World Bank, Uganda Horti-floriculture Sector Technical Note 2

In 2002, commercial floriculture was a major non-traditional agricultural export sector, valued at USD 21 million FOB. There were 20 commercial farms in production with a total acreage of 122 hectares. Export volume had increased from 721 tons in 1995 to 3,820 tons in 2002.

The Ugandan floriculture industry mainly produces sweetheart roses and chrysanthemum cuttings. Almost all exports are destined for Europe, with the Netherlands receiving the largest share. While most of the flowers produced in Uganda are sold through the Dutch flower auction, small quantities are also sent directly to the UK, Belgium, and Germany. In 1998, Uganda ranked 11th among the suppliers of roses at the Dutch auction market; by 2002, it had moved up to 6th place.

Producers in the floriculture industry in Uganda are large commercial farms that are vertically integrated and sell directly to importers in Europe. The value chain at the producer level is consolidated mainly because of the large amount of capital that must be invested in growing flowers suitable for export and putting a working cold chain in place.

Source: Verenig	Source: Vereniging van Bloemenveilingen in Nederland (VBN)							
Rank	Country	Value	Stems	Av. Price				
		€mil.	mil.	€ct /stem				
1	Kenya	148.5	1,027.9	14.4				
2	Israel	133.1	866.4	15.4				
3	Zimbabwe	67.2	474.0	14.2				
4	Ecuador	21.7	75.4	28.8				
5	Spain	15.5	168.4	9.2				
6	Uganda	12.8	136.6	9.4				
7	France	10.0	63.5	15.7				
8	Zambia	10.0	79.1	12.6				
9	South Africa	9.1	31.1	29.3				
10	Tanzania	7.0	55.1	12.7				
	Others	46.8	299.1	15.6				
	Total	481.7	3,276.6	14.7				

Table 11: Rose Sales at Dutch Auction in 2002

The 2002 assessment included analyses of the Ugandan floriculture value chain and production cost structure, which were then benchmarked against those of Kenya. The value chain analysis for roses that, due to showed strong competition and exchangeability of suppliers, profit margins are very small in the flower value chain. Producers receive around 40 percent of each euro spent by a consumer on flowers. At the next step of the value chain, the exporter keeps another 40 percent of the final consumer price, although the exporter pays the associated airfreight costs. When flowers reach the Netherlands, an importing agent unpacks the flowers, re-hydrates them, and palletizes them for sale to the Dutch flower auction, which adds a 5 percent fee. Wholesalers and other retailers then add another 15 percent before the final consumer pays the corresponding final price.

Using Benchmarking: Value Chain Analysis





The final consumer pays 100% of the final asking price. If the value chain is traced back from there, it shows how much cost each step of the value chain is adding as it builds up to the final price to the consumer. Producers and exporters, in combination with the fees they have to pay for airfreight and handling costs, make up the largest segment of the final cost to the consumer.

To help assess Uganda's competitiveness in floriculture, the World Bank benchmarked part of Ugandan producers' value chain and production cost structure against those of Kenyan producers. Kenya was selected because it is located in the same geographic region (although with slight a difference in climatic conditions, especially the altitude for growing similar products) and is a competitor across many industries. Table 12 benchmarks Uganda's sweetheart rose production cost structure with that of Kenya's; this is demonstrated graphically in Figure 20. (See Table 13 and Figure 21 for a comparison of the Cost Structure for Cuttings Production.)

Table	12: Sx	veetheart	Rose	Production	Cost	Structure	ner	Hectare:	Ugand	la vs.	Kenva	(in	UST))
I avic	12.01	veetheart	NUSC	1 I Ouucuon	CUSI	Suuciare	per	nectare.	Uganu	1a vs.	пспуа	(III)	USL	"

Description	Uganda	Kenya	Difference (Uganda minus Kenya)
Production (stems)	3,750,000	3,250,000	500,000
Average price per stem	0.088	0.094	-0.006
Gross sales	330,000	303,875	26,125
Air-freight costs	71,205	63,503	7,702
Handling costs (NL)	9,821	9,408	414
Handling costs (Local)	1,473	2,352	-879
Auction/Agent fees	52,800	48,620	4,180
Total marketing costs	135,300	123,883	11,417

Net sales	194,700	179,992	14,708
Packing material	6,250	5,559	691
Fertilizers/Chemicals	30,000	25,500	4,500
Electricity Fuel	15,000	11,250	3,750
Labor	12,775	16,425	-3,650
Staff/management	20,000	25,000	-5,000
General prod. costs	5,000	5,000	0
Royalty fee	15,400	15,400	0
Repairs & maintenance	4,000	4,000	0
Operational costs	108,425	108,134	291
Gross margin	86,275	71,858	14,417
Depreciation investments	30,000	30,000	0
Interest loans	18,000	18,000	0
Net margin	38,275	23,858	14,417

Source: Uganda Horti-floriculture Sector Technical Note 2





Source: J. E. Austin Associates, Inc, Uganda Horti-floriculture Sector Technical Note 2

Uganda produced more stems than Kenya, but Kenya's stems commanded a slightly higher price. Uganda was at a disadvantage in terms of the costs for air-freight, fertilizers and chemicals, and electricity and fuel. Net profit in Kenya was lower than in Uganda, despite Kenya's premium of half a cent per stem.

Understanding Uganda's cost position relative to its competitors allows for the development of a strategy. For example, in this case, if Uganda could lower its air-freight costs, and the costs of inputs, such as fertilizers, chemicals, and electricity through efficiency gains, it could sell more flowers at a slightly lower price than Kenya.

Description	Uganda	Kenya	Difference (Uganda minus Kenya)
Production	30,000,000	27,500,000	2,500,000
Average price	0.017	0.018	-0.001
Gross sales	495.000	484.000	11,000
Air-freight costs	79,750	74,250	5,500
Handling costs NL	11,000	11,000	0
Handling costs Local	1,650	2,750	-1,100
Auction/Agent fees	0	0	0
Total marketing costs	92,400	88,000	4,400
Net sales	402,600	396,000	6,600
Transport Packing			
material	28,000	25,667	2,333
Fertilizers/Chemicals	40,000	34,000	6,000
Electricity Fuel	30,000	22,500	7,500
Labor	38,325	49,275	-10,950
Staff/management	60,000	60,000	0
General prod. Costs	20,000	20,000	0
Planting material	52,500	52,500	0
Repairs & maintenance	6,000	6,000	0
Operational costs	274,825	269,942	4,883
Gross margin	127,775	126,058	1,717
Depreciation investments	40,000	40,000	0
Interest loans	24,000	24,000	0
Net margin	63,775	62,058	1,717

Table 13: Cost Structure	for Cuttings Production	per Hectare: Ug	ganda vs. Kenya in USD
--------------------------	-------------------------	-----------------	------------------------

Source: Uganda Horti-floriculture Sector Technical Note 2



Figure 21: Components of Total Cost of Cuttings in Uganda and Kenya

Source: J. E. Austin Associates, Inc, Uganda Horti-floriculture Sector Technical Note 2

In benchmarking the cost structure for cuttings, the most apparent conclusion is that Uganda does not have much comparative advantage in relation to Kenya. Again, air-freight costs and input costs are higher in Uganda, but net margin is still slightly greater in Uganda.

Using Benchmarking: Cluster Structure

The assessment also benchmarked the structure of Uganda's floriculture cluster against that of Kenya and the Netherlands. The cluster assessment included important elements that are part of a value chain analysis but also looked at the linked impact on cluster participants of driving forces and critical success factors. In Table 14 below, Uganda is seen to be the weakest of the three countries. The Netherlands was included as a best practice case, against which Kenya performs moderately well. However, both Uganda and Kenya lack the specialized equipment and services needed to reach their potential. Prices for inputs in Uganda are about 10 to 20 percent higher than in Kenya. Also, several flights per day directly link Kenya to Europe and can be used for exporting flowers. Uganda has fewer direct flights to Europe, and exporters must often ship through Nairobi rather than directly to Europe. Both countries sell their flowers to the Dutch auction.

			Uganda	Kenya	Netherlands
		The exciption	4	+	
			т	+	++
s			+	+	++
rivii rce			+/-	+/-	+
<u>d</u> 6	1	The horticultural sector	+/-	+	++
		Land and climate	+	+	+
	set	Labor	+	+	
	E	Local infrastructure	+	+	+
	nm	Agricultural input supply	-	+	++
	lini	Water	++	+/-	+/-
	2	Access to the market	+/-	+	++
ors.		Loans and credits	-	+/-	++
ncto		Producer associations	+	+/-	++
S fé	set	Skilled management	-	+	++
Sec	al	Cargo handling facilities	++	+	++
ncc	ion	Promotion organizations	-	+/-	+
s le	ldit	Horticultural education	-	+/-	+
tice	Ac	Research and training	-	+/-	+
Ciri		Extension service	-	-	+
		Seeds & plant material		+/-	++
		Soil & water testing facilities	-	+/-	++
	<i>4</i>	Growing medium	-	-	++
	s s	Packing material	-	+	++
	ctic	Consultancy services		-	++
	ate	Bookkeeping & accountancy		-	++
	brd H m	Certification Institute		-	+
	nd anc	Selection and breeding		-	++
	s a Its i	Greenhouse construction		-	++
ork	ice Ipu	Greenhouse equipment		-	++
etw	erv. ir	Greenhouse covering material		-	++
ž	Ś	Fertilizers & chemicals		-	+
stor		Specialized transport		-	++
Sec		Biological crop protection			++

Table 14: Driving Forces and a Comparison between Uganda, Kenya, and the Netherlands

Source: Uganda Horti-floriculture Sector Technical Note 2

Outcomes

Based on the results and findings of the benchmarking exercise, floriculture industry leaders worked together to develop the *Ugandan Floriculture Competitiveness Plan: 2005–2010*. Elements of the strategies presented in the Plan build upon the benchmarking exercise. The benchmarking exercise highlighted shortcomings in several areas, including product diversification, research and development, and transport costs. The Plan looks to improve the Ugandan floriculture industry's performance through the accomplishment of five key targets and associated sub-strategies:⁵⁶

- 1. Increase export volumes and values
 - o Expand in existing markets with existing products
 - o Open new markets with existing products
 - o Establish a Europe-based unpacking distribution facility
 - Diversify and expand product offerings
- 2. Increase internal and Foreign Direct Investment

⁵⁶ Building Uganda's Global Competitiveness in Agribusiness—The Uganda Floriculture Competitiveness Plan: 2005– 2010. USAID.

- o Clarify strategic industry status
- o Develop and implement a floriculture investment incentives package
- Develop an upland expansion plan
- o Develop financial support mechanisms for the industry
- o Develop and carry out investment promotion program
- 3. Improve technology and practices
 - o Set up and maintain an effective cold chain management system
 - o Facilitate expansion of fresh management services
 - Establish the Floriculture Training Development Center
 - Develop and Install a product expansion program
- 4. Develop and launch a quality, branding, and marketing campaign
 - o Strengthen Uganda's position as a quality service provider
 - o Standardize export packaging
 - o Develop and launch a floricultural branding program
 - Develop and Implement a marketing program
- 5. Position the Uganda Flower Exporters Association (UFEA) for sustainability
 - Prepare and implement industry development support program
 - o Produce revenue generation and sustainability plan

Today, the volume and value of exported cut flowers continues to grow, reaching 7,596 mt and USD 34,720,000 in 2006. UFEA supports the industry through research, training, and market promotion. Producers are also focusing on quality assurance, standards, and certification, with several farms already lodging applications for EurepGAP certification. A majority of the Ugandan growers have received MPS ABC certification, indicating that they meet standards for EurepGAP's international environmental program focusing on pesticide use, recycling, and energy and water use. Therefore, Uganda is not only competing on the basis of lower costs, it has also moved towards competing on value, as seen by

Figure 22: Uganda's Flower Exports 1994-2006



improvements made in quality and standards certifications.

Most flowers are now shipped directly to Europe, rather than via Nairobi. Approximately 50 percent of the flowers are sold at the Dutch auction, with the other 50 percent sold directly to wholesalers and retail outlets in Europe.⁵⁷ Uganda has also begun shipping small quantities of cut flowers to the United States, which represents a new market opportunity African flowers. In terms of product for diversification, Uganda's Competitiveness Plan focuses on continuing to expand exports of sweetheart roses. However, one grower has established a farm at higher elevations to attempt growing larger-headed, higher value roses.

Source: USAID APEP Program

⁵⁷ Author interview with Christine Kiwanuka, USAID APEP Program.

Tool 4: Upgrading and Deepening the Value Chain

Increasing the competitiveness of the value chain by moving it in a new direction – towards a new market, market segment, or customer; towards increased efficiency within the value chain; or towards adding operations within the value chain, for example – is referred to as upgrading. Actions that upgrade or increase the competitiveness of a value chain can take many forms and include improving product quality, adding more operations to the value chain, bringing value chain operations into a country from overseas, capturing a new market channel, and entering a separate value chain (new market) with a similar product.⁵⁸

In deepening the value chain, firms address gaps including unmet market demand and value, opportunities for vertical or horizontal integration, greater specialization, and the expansion of services to other value chain members.

The gap analysis (described in Tool 3) methodology provides a means to identify additional operations that may be added to the value chain. It also provides guidance on how best to prioritize opportunities and adjust or expand operations.

Adding value is often incremental, but it can often be accelerated by Foreign Direct Investment (FDI) and joint ventures—especially with previous buyers. As well as increasing specialization, adding value also frequently means bringing operations in-country that were previously conducted abroad.

Deepening Through Adding Operations

Efforts taken to deepen the value chain must be driven by market opportunities and demands. The addition of operations requires sound market analysis, strategic planning, and good communication among value chain participants. The value chain must prioritize the possible opportunities that it identifies and then act as a unit to add them. In Mongolia, a competitiveness initiative was able to work with the meat industry in order to identify possible operations to incorporate (see the box below).

⁵⁸ "AMAP BDS Knowledge and Practice Task Order Lexicon," USAID.

Box 4: Mongolian Meat Industry—Upgrading the Value Chain

The Mongolian meat industry has traditionally been oriented towards low-end exports of animal carcasses to Siberia. Through work with the Mongolian Competitiveness Initiative (MCI)^{59,} plans were made to integrate value-added operations such as quality checks, packaging, and marketing into the meat industry value chain. These upgrades were intended to reorient firms toward more demanding and lucrative export markets.

With USAID and USDA assistance, the value chain solicited a former USDA expert in meat regulatory standards to help facilitate improvements in health and sanitary standards. MCI also identified transport options and completed cost studies to confirm the feasibility of export to five Asian and two Middle Eastern markets. Lobbying various associations and Government agencies, the project worked with industry to streamline government policies and standards related to agricultural export.

In Figure 23, both the traditional and a new "processed" meat export value chain are detailed. In this figure, the

Figure 24: Deepening the Value Chain Under Two Scenarios



Source: Nathan Associates, Inc. and J.E. Austin Associates, Inc. 40 percent in earnings from the meat industry.

Author: Michael Gorman – J.E. Austin Associates



Figure 23: Mongolian Meat Export Value Chains

Source: J.E. Austin Associates, Inc. processed meat export channel represents the opportunity to add value by incorporating additional operations within the value chain. The "VC Intervention" arrows represent opportunities identified for intervening in the Mongolian meat industry in order to deepen the value chain.

"Figure 24: Deepening the Value Chain Under Two Scenarios" quantifies the value that can be added by deepening the value chain. In this instance, the addition of veterinary services, meat inspection, processing, packaging, labeling, and marketing operations to the Mongolian meat value chain provides gains of nearly

⁵⁹ The Mongolian Competitiveness Initiative was implemented by the consulting firms Nathan Associates and J.E. Austin Associates.

Deepening is Specialization

Opportunities for specialization rest on the size of the market for the specialized operation or service, and on the confidence that the customers of the specialized business or operation have that the work will be carried out to an appropriate quality level with needed degrees of customer service, requisite confidentiality, and on a sustained basis (i.e. the specialized business will not fail and disappear). These criteria are, in many respects, the entry criteria for value chain deepening through specialized operations.

Specialization may offer the value chain the opportunity to accomplish otherwise unattainable goals. Risk and investment costs may now be shared and offset by the cost savings that result from cooperation and information sharing.

Methods for Deepening the Value Chain

New Entrants/Entrepreneurs

Entrepreneurship generates many new entrants, added operations, and captured value in every value chain. In each market, entrepreneurs arise to take advantage of market opportunities and create links with value chains. Such new business formation can be encouraged by access to business services (such as availability of finance), and by good relationships and familiarity with the needs of the other actors in the value chain.

New Investment (Domestic and Foreign Direct)

Foreign direct investment by multinational corporations is one of the most common ways that technologies are transferred to value chains in developing and emerging economies. Also, knowledgeable domestic businesses can purchase or license new technologies. Bringing quality-sorting or product-packaging operations into the developing country, for example, can increase cost efficiency and product competitiveness while also adding value to product value chain and the local economy.

Box 5: Deepening the Value Chain: Glass Jar Production in Armenia

Armenian agricultural products include grain crops, vegetables, and fruits. The latter two are mainly exported in processed form; vegetables are most often canned, while the fruits are processed into juices, jellies, jams, compotes, leathers, and fruit fillers, which make up the majority of the produce packed in cans and jars. The lack of availability of proper quality screw-top jars and the need to import them from Europe raises costs for this type of processor/exporter. (Armenia also has important wine and brandy industries, which require glass bottles.)

In 2003, Armenia had very restricted glass jar production capabilities. Existing production capacities had few molds, and, therefore, were able to produce only a limited number of standard designs. Armenia's landlocked situation and high transportation costs make the import of glass bottles and jars very expensive, which restricts opportunities for export of Armenian produce.

The fact that quality, locally manufactured screw top jars were not available was a constraint to adding value. There are four big canners in Armenia and several smaller ones. The quality of Armenian-made glass containers and the ability to customize them (jars for juice or preserves, for example) are not satisfactory for quality- or image-conscious processors. While critical of the quality of locally made bottles, processors cited price pressure as requiring them to continue to purchase locally. This greatly hampered the ambitions of producers of food and beverage products, who were unable to use Armenian-produced packaging as a differentiating factor.

Recognizing a gap in the glass container market, two companies made major investments in 2004 in glass jar manufacturing to serve the Armenian market.

The bigger investment was by Saranist, which is also trying to penetrate the Georgian market. Saranist established a new, modern glass manufacturing factory, Arm-Glass Company. The second manufacturing investment was by Glass World Company. GWC has modernized its production line through a EUR 30 million investment, importing advanced, high-technology equipment and installations from Western Europe. Both Saranist and GWC have long been established in Armenia - Saranist in 1990 and GlassWorld (formerly Armkhrustal) in 1964.

Source: J.E. Austin Associates Interviews, websites. Author: Michael Gorman - J.E. Austin Associates

Vertical Integration (see also Tool 6)

Existing firms also frequently identify opportunities to incorporate new technologies or operations into their structures. Recognizing opportunities to add value, achieve efficiencies, or improve quality—for example, by adding operations through vertical integration (by a value chain or by a firm with the value chain)—enables value chain deepening. In many parts of the world, vertical integration is achieved through acquisition of another firm along the value chain.

The formation of new organizations, such as farmers associations, service-provider associations, and marketing organizations also provides opportunities for otherwise fragmented producers or other businesses to combine their resources to add operations to a value chain.

Commercial Joint Ventures

Sharing the risk of incorporating new operations into the value chain or investing in a new technology is an appropriate measure in some environments. Fresh produce preparation, packaging, and logistics centers are frequently good candidates for a joint venture, for example. In another example, the tea cluster of Sri Lanka worked closely with the University of Moratuwa to jointly develop and commercialize a local color separator technology, which could be used to upgrade the quality of tea supplied to the local Sri Lankan auction. The cluster was able to leverage University resources in order to develop a new separator at one-fifth the cost of the old separators which were imported from abroad.

Conclusion

Many actions can be taken to increase the competitiveness of and to add value to a product's value chain. These upgrading activities are often achieved in part by deepening the value chain through means such as adding operations to the value chain, vertical integration, bringing operations into a country's value chain, or investing in new technologies. These actions are often facilitated by the opportunities created by Foreign Direct Investment, association-provided services, new investment, vertical integration, and public-private partnerships.

Successful value chains add value to and deepen their operations while responding to market conditions in order to achieve growth and increase profitability.

Case Study: Kenyan Green Beans and Other Fresh Vegetable Exports⁶⁰

Introduction

Possessing a perfect agricultural climate for off-season vegetable production and export to Europe, Kenya has been providing European tables with vegetables for nearly fifty years. Kenya's success has been due to market segmentation, servicing niche markets, and investing in marketing. The industry has constantly refocused its efforts on exporting higher unit-priced products. Products not fitting the profile have been dropped, and the industry has also expanded into products of greater value such as pre-packed and mixed vegetable packs. For example, export mixed vegetable packs command a price of USD 8.90 per kg vs. extra fine beans at USD 4.14, fine beans at USD 3.30, and Asian vegetables at around USD 2.00 per kg.

Tool: Adding Value through Added Operations

"Deepening the Value Chain" refers to opportunities to add or capture more value within a particular product or industry's value chain (VC) by adding processes. Deepening can be achieved by recognizing "gaps" in the value chain or facilitating new linkages between value chain actors. These opportunities can be achieved through by various actions,

Figure 25: Map of Kenya



Source: CIA World Factbook 2007

such as adding operations to capture market demand and value, upgrading the value chain, re-orienting the chain to new market opportunities, integrating different aspects of the chain, promoting specialization among chain participants, and increasing chain cost efficiencies.

Background: Kenyan Green Beans⁶¹

Kenya has been exporting vegetables to the UK since the 1950s. Reasons for Kenyan success have varied with the changing market forces of the highly competitive UK and European markets. Kenya's original success in exporting vegetables was based on its climatic and geographic competitive advantage. Producing temperate products all year round and being well served by northbound air-freight (thanks to the Kenyan tourism market) proved lucrative for Kenyan vegetable exporters. Increasingly, Kenyan success has been due to market segmentation, investing in certification schemes, adding value to products through sophisticated packaging, servicing niche markets, and investing in marketing.

Over the years, due to effective public/private dialogue, the Kenyan Government has been receptive to implementing regulatory changes, investing in education, and improving infrastructure which have increased the competitiveness of the industry. For example, the Horticulture Crops Development Agency (HCDA) of Kenya was initially directly involved in the trading of vegetables but eventually switched to a more facilitative function; it now focuses solely on certification schemes.

Similarly, restrictive policies regarding the sales of fresh pineapples in the 1970s and importation of planting materials in the 1980s have now been lifted as a result of close consultation with the private sector. Throughout the 1970s and 1980s, the majority of Kenyan vegetables imported into the European Community were handled by firms that serviced wholesale markets and small or medium retail outlets. In

⁶⁰ Authors: Carlton Jones, Michael Gorman, and Martin Webber - J.E. Austin Associates

⁶¹ NEPAD Technoserve case study.

Case Study: Kenyan Green Beans and Other Fresh Vegetable Exports

the 1980s, Kenyan exports doubled in five years due a differential foreign exchange rate for horticulture exports, which the Government set below average prices, providing further incentive for exporters to invest in the industry.

By the late 1990s, due to lobbying efforts of the Fresh Producers Exporters Association of Kenya (FPEAK), the Kenyan Government partnered with the private sector to expand the Fresh Produce Terminal at the Nairobi airport, thus improving the competitiveness of fresh vegetable exports. Then, throughout the 1990s, large supermarkets began to dominate the European grocery sector, in part, by featuring signature "fresh produce" sections. As they did so, these firms increased the market demand for higher quality, more variety, and price-competitive fresh produce. To meet demand, many firms have decided to vertically integrate their retail and wholesale operations, thus concentrating their power in the market and driving price competition and product diversification as major market forces.

Timeline of Horticulture Development in Kenya

- 1957 First fresh produce to UK by air
- 1960s Duty-free access to UK market
- 1960s Investment in private farms around Lake Naivasha.
- 1967 Horticulture Crops Development Agency (HCDA)
- 1970s Egerton College starts offering degrees in horticulture
- 1975 Fresh Produce Exporters Association of Kenya
- 1980s HCDA passes EU trade enquiries to exporters
- 1987 Exports double in 5 years
- 1999 New fresh produce terminal built at Nairobi Airport



Figure 26: Kenyan Green Bean Value Chain

Source: J.E. Austin Associates, Inc..

In the 2000s, as the power of the supermarkets continued to drive the market, many supermarkets began to pursue market segmentation and branding strategies which increased the demand for higher quality standards, different varieties, and organic or "safer"⁶² produce. A number of exporters have invested heavily in growing their own high-quality certified vegetables to take advantage of the increased market opportunities for high-quality produce. The effect of these trends has been a much shorter supply chain, a greater degree of vertical integration, fewer active players, and production and exporting on a much grander scale. By the early 2000s, seven of the largest food retailing chains accounted for 76 percent of fresh fruit and vegetable sales and 70 to 90 percent of fresh produce imports from Africa.⁶³

As of 2004, the total Kenyan vegetable export trade was worth USD 139m, and the country ranked second in Africa in fresh exports vegetables. The

industry employs 45,000 to 60,000 people, of whom an estimated 60 percent are women, in commercial farms, processing, and logistics operations; another 7,000 are smallholders. Employees typically earn just under USD 2 per day, while smallholders are reportedly able to earn the equivalent of USD 7 per day.

⁶² "Safer" refers to produce with limited levels of chemical residue.

⁶³ "The Market for Non-Traditional Agricultural Exports," FAO.

The Value Chain

In Kenya, green beans have traditionally been the most popular cash crop among smallholders due to their short growing period which facilitates a more regular cash income.⁶⁴ Farmers will typically plant as much as they can sell, and those with contracts or a firm commitment from an exporter may devote 100 percent of their land to the cultivation of green beans. Green bean monoculture cultivation can yield up to four harvests per year if accompanied by application of chemical fertilizers. The main challenges faced by smallholders are both brought by the rainy season: a higher disease incidence in the crop and poor accessibility to areas with bad roads which prevents regular collection by exporters.

The value chain graphic (Figure 27) illustrates the smallholder "broker" and "small to medium exporter" market channels. The broker channel is composed of approximately 20 to 50,000 micro-enterprises



Figure 27: Integrated Large Exporters Value Chain

Source: Adapted from Irwin, Grant, Parker, and Morgan.

certification, and product diversification.

sed of approximately 20 to 50,000 micro-enterprises (mostly households) who sell to SME exporters (SMEX) or brokers. The value chain can be characterized by its low levels of information sharing, with inaccurate records of chemical usage during cultivation that denies it access to the European market. In the SMEX value chain, approximately 15– 20 exporters may contract or have close working relations with their green bean suppliers (nearly 4,000 SME farmers, small outgrowers, and farmer associations). The exporters typically provide inputs to ensure the quality and quantity of products. Smallholders and small and medium producers have been increasingly pushed out of the cultivation of green beans due to market requirements and conditions.

The present market conditions in the EU supermarket sector have influenced a shift in the Kenyan green bean industry to more integrated value chains best represented by larger integrated exporters. The value chain for large integrated exporters (Fig 26) is characterized by exporters having strong links to end markets through and producers contractual ownership. agreements and Supply chain management is more efficient due to information sharing within the integrated value chain that eliminates costly demand shortages or oversupply. Products are traced from their origin and production practices are controlled to ensure quality and certification schemes. Increasingly, the value in these relationships is garnered from investments made in value-added operations such as packaging, labeling,

These value-added results are clearly represented in the labor statistics for the integrated large exporter value chain. Approximately 7,000 smallholders are involved in fresh vegetable export, compared to 40,000–60,000 in the processing industry, at packhouses, or as farm laborers. For example, Homegrown is Kenya's largest horticulture exporter and is an example of a highly integrated company. Ninety percent of Homegrown's crops are grown on its own farms, where it controls the storage, cooling, and logistics from the field to the packing station; it has a joint venture with an airfreight company and a dedicated importer in the UK. Homegrown works with about 600 smallholders and employs nearly 8,000 seasonal employees for its processing operations. Homegrown has also recently completed a factory for prepared

⁶⁴ Okado, Mark, UN Conference on Trade and Development, "Background Paper on Kenya Off-season and Specialty Fresh Vegetables and Fruits: Lessons of Experience from the Kenya Horticulture Industry," 1999.

salads, providing the capability to pick, prepare, fully label, and transport the salads to the supermarket shelves within 48 hours.

Homegrown's recent investments in product development are indicative of the value drivers for the entire fresh vegetable export industry. Driving this accelerated value growth in fresh vegetables has been the emergence of semi-processed products that meet stringent European standards and certifications. This growth in exports consists of a broad range of products produced under very strict hygienic conditions. In Kenya, the increase in value-added processing to produce "high care" products such as salads, prepared vegetables, and stir-fry mixes has increased export values for fresh vegetables by 250 percent.⁶⁵

Conclusion

The Kenya fresh vegetable export industry has grown enormously in size and value added, in large part by implementing new processes and operations. These have been initiated by private business in response to evolving market trends, recognized opportunities, and value chain pressures. The public sector has been an active partner in this growth. Further opportunities exist to increase the competitiveness of the Kenyan fresh vegetable export industry through value chain deepening, as well as through other approaches (e.g., increasing the technical capacities and market understanding of serving growing markets beyond Europe, extending the exporting season, and reducing costs and losses through infrastructure).⁶⁶ The realization of each enhanced process will, in turn, provide opportunity for added services within the value chain.

⁶⁵ Jaffee, Steven. 2003. "From Challenge to Opportunity: Transforming Kenya's Fresh Vegetable Trade in the Context of Emerging Food Safety and Other Standards in Europe." Agriculture and Rural Development Discussion Paper. World Bank.

⁶⁶ TNS. 2004. "Nepad: Ag Commodity Case."

Tool 5: Identifying Business Models for Replication

By analyzing value chains, participants can often identify intermediation opportunities that offer increased efficiency, economies of scale, reduction of transaction cost, or more value added in the chain. Entrepreneurs and businesses may be the first to identify and act on them; or government and development partners (and their consultants/practitioners) may be the ones to promote such opportunities. The defining characteristic in the context of this tool is that the opportunities offer the possibility of replication within the value chain.

The benefits of identifying and promoting intermediation opportunities go beyond adding value and upgrading the value chain (with the resulting net economic benefit in jobs, wages, and incomes). More than these, however, use of this tool in countries with a particularly weak private sector and lack of an entrepreneurial culture amounts to investing in the promotion of entrepreneurship. Additionally, as individual entrepreneurs improve on the basic business model, they often generate their own innovations.

Replicable business models can be recognized through a variety of mechanisms and experiences. For example, the opportunity can be identified through the simple need to upgrade quality (of both raw materials and processed product); through learning of business models that have worked in other regions or countries in related types of value chains; or through the application of analytical tools described elsewhere in this Guide—such as identifying needed services, benchmarking and gap analysis, and market analysis.

Box 6: Replicable Business Models - Rwanda Coffee Washing Stations

Rwandan coffee was a principal source of foreign exchange for the country until the 1990s. But coffee's contribution to Rwanda's foreign exchange earnings declined through the mid- and late-1990s. Rwandan coffee production never recovered to 1992 production volumes (39,000 mt), because of inefficiencies in the coffee value chain. The country's disparate nature of coffee farming, the poor health of its coffee trees, the lack of wet-milling stations, and low incentives for reinvestment all contributed to inefficiency. Growers were not offered higher prices for better quality beans, so they spent little time grading and separating their bean harvests. Low coffee yields and poor price points influenced farmers to focus on other crops with higher margins, further diminishing coffee's competitiveness in world markets.

Despite the constraints that led to low-quality and low-quantity commodity-grade coffee, the GoR and donor partners believed that Rwanda possessed the capacity, environmental conditions, and political will to improve its coffee position in world markets. What Rwanda lacked was technical capacity, market information, and a coherent strategy. Two USAID-funded projects, Partnership for Enhancing Agriculture in Rwanda through Linkages (PEARL), and Assistance a la Dynamisation de Agribusiness au Rwanda (ADAR), helped to provide the strategy and technical capacity that assisted in Rwanda's coffee quality and quantity improvements. The projects sought to improve Rwandan coffee by, among other actions, facilitating the opening and equipping of coffee-washing stations in Rwanda's top 50 producing districts. These washing stations filled a crucial gap in Rwanda's production cycle and allowed the coffee's quality to improve.

With an annual budget of almost USD 700,000 dedicated to supporting the coffee sector, ADAR worked directly with private investors to open 16 washing stations in 2005. Direct assistance included feasibility studies, business plans, construction planning and supervision, and training in coffee processing. Similarly, PEARL worked with rural cooperatives to assist in cooperative formation, business planning, washing station construction, processing, cupping, marketing, and Fair Trade certification. In both projects, the assistance provided the platform for a replicable business model to be adopted for numerous future washing station openings. In the model, investment opportunities were created via a loan guarantee program that allowed the private investors to construct collection/washing stations and process coffee beans for improved quality. The model was replicated each time a

private sector investor sought to open a washing station. The investor took out a loan from the guarantee fund and, along with technical assistance from the projects, began processing coffee for export. By November 2005, 10 of the 11 loans provided by the program, totaling USD 1.6 million, went to private sector operators.

As of January 2007, the private sector was continuing to invest in coffee washing stations in Rwanda. The replicable business model provided by the projects has helped to establish 80 functioning stations throughout the country, with the goal of 120 washing stations by the end of 2008. The washing stations provide an important intermediate role in the coffee value chain and have also proven to be platforms for entrepreneurship and entrepreneurial innovation. Ultimately, Rwanda hopes to have its entire coffee production fully washed by 2010.

Author: Carlton Jones – J.E. Austin Associates

The Pakistan Dairy industry provides a useful illustration (see Box 7) of a replicable model. Value chain analysis in Pakistan revealed that 45 percent of milk produced in the country never reaches market because, in most areas, only the morning milk (55 percent of potential output) is collected. The rest (evening milk) goes to waste. Further, domestic demand is growing at twice the rate of supply. This analysis led to the identification of an intermediation opportunity that could be exploited by entrepreneurs – namely, investments in simple refrigeration centers for the purchase and collection of milk which allowed an increase in the quality and supply of milk, as well as providing farmers an outlet for selling it. This business model is being widely replicated and it is estimated that, just two years after efforts began, at least 2,000 collection centers will be operating by 2008.

Box 7: Dairy Pakistan—Identifying and Replicating Business Models within the Value Chain

Background

An estimated 30-35 million farmers in Pakistan are engaged in raising livestock, which generates 30–40 percent of their income. Ninety-seven percent of fresh milk is either consumed locally or distributed through informal trading routes. The farmers' dairy production has often not reached domestic markets, in spite of the fact that the market for dairy is growing twice as fast as the supply.

Dairy farmers want to invest – but need to manage risk

The dairy value chains in Punjab Province and in the area around Karachi recognized an opportunity to improve the quality and increase the quantity of the milk that they were producing and marketing. The chain participants knew that they had unmet demand because they were able to sell everything they could supply. The industry's Strategic Working Group (SWOG) meetings identified a huge market opportunity that could entice entrepreneurs to invest time and money into meeting this growing need. In fact, the dairy sector was missing out on selling its second (evening) milking, which is 40 percent of the milk produced on the farm.

The most significant issues for the value chain were that 97 percent of the trading was done on an informal basis, and there was no cold chain to handle the storage of a second milking. Of the milk that farmers do sell, 15–19 percent is wasted en route to market, again due to spoilage because of lack of chilling.

Although these facts appear to present investment opportunity, it can be difficult to create the conditions within the value chain that motivate businesses to invest and that encourage the value chain to upgrade its practices. Buying new equipment and changing their practices means both costs and risks for farmers and intermediaries. These barriers were removed by a facilitated strategic planning process that identified a workable business model. The planning process developed sufficient trust among the participants to create a strong win-win solution and to generate commitment to the model, with risk-sharing and up-front financing from Dairy Pakistan, the public-private institution that was designed by the SWOG project.

The replicable business model (Collection Centers/Cooling Stations)

The core of the business model was to encourage entrepreneurs to invest in and to manage collection centers/cooling stations. The entrepreneur puts up 20 percent of the investment and receives a no-interest loan for the remainder, which was subsidized by the government, managed by Dairy Pakistan, and distributed by a consortium of banks. Dairy Pakistan then provided technical training on how to operate the cooling station. The large dairy processors/distributors committed to regular, predictable milk collection.

Communicating the case for investment

The SWOG presented the market opportunity to potential entrepreneurs. Dairy Pakistan and the several banks sent out formal invitations for entrepreneurs to submit applications to Dairy Pakistan for investment. The project and Dairy Pakistan supported entrepreneurs with technical assistance in learning to operate the tanks, again reducing the risk of investment. As of mid-2006, the project was facilitating 2,150 collection tanks and had received 3,050 applications.

Summary and Expected Results

It is expected that at least 2,200 people will be employed at the centers and that the entrepreneurs who own and operate the centers will earn USD 63M (net present value). Additional benefits will result from increased availability of milk on the market, additional household (farm) incomes, and profits further down the market chain.

Author: Mike Ducker – J.E. Austin Associates

Once a business model suitable for replication is identified, value chain participants and development practitioners can support and facilitate value chain stakeholders in successfully implementing a pilot enterprise. Once proven, stakeholders can encourage the replication through promotional campaigns, business associations, technical assistance, and other available means. In the Pakistan Dairy example, a public-private institution to promote the dairy sector is facilitating the replication of the collection centers by promoting the business opportunity, soliciting applications for discounted farm cooling tanks (which the institution negotiated with the provider), and identifying commercially appropriate sites for centers.

A Step-by-Step Summary of Tool 5: Identifying Business Models for Replication

- Identify intermediation opportunities (business models) to increase efficiency and value added suitable for replication. Many of the tools in this Guide are helpful in this respect.
- Support and facilitate value chain stakeholders to implement a pilot enterprise.
- When an approach is successfully proven, encourage its replication through promotional campaigns, business associations, technical assistance, and other available means.

Case Study: Identifying and Implementing Replicable Business Models-Mozambique Cashews⁶⁷

Introduction

Rejuvenating the Mozambican cashew sector requires innovative approaches to bring value back to the actors in the value chain. This case demonstrates how small and medium hand-processing plants were identified as replicable businesses supporting that rejuvenation. Mozambique has not yet returned to its former cashew dominance, but those in the sector have learned that, through replicable business models, value chain actions can rebuild the private firms that bolster the sector.

Points to Consider

When reviewing this case consider the following questions:

- What led to the need to identify replicable business models?
- When is it appropriate to consider implementing such a model?
- What role do local entrepreneurs play in implementation?

Background

From the 1920s until mid-1970s Mozambique was considered the world's leading cashew producer (240,000 mt at its peak in 1973)⁶⁸, with considerable domestic capacity in processing quality cashews. By the early 1980s, Mozambique possessed over a dozen processing factories and was the first African country to process cashews on an industrial scale, rather than through the traditional method of hand-processing by SMEs.

However, by the end of the 1970s, other global producers (India and Brazil) began to threaten Mozambique's cashew dominance.⁶⁹ In response, in 1978, the GoM banned the export of raw nuts. The rationale at the time was to stimulate domestic processing and maintain global processing dominance. But after a variety of events, including a civil war from 1982–1992 and the adoption of policies that fixed raw nut prices, production continued to decline (down to 22,106 mt in 1990).⁷⁰

Figure 28: Mozambican Cashew Nut Exports Since the 1970s (In thousand metric tons)



Source: FAOSTAT

In 1992, the GoM implemented many reforms in the hope of regaining past cashew processing performance. The raw nut export ban was lifted, previously state-owned factories were privatized and

⁶⁷ Authors: Carlton Jones and Martin Webber - J.E. Austin Associates

⁶⁸ Technoserve.

⁶⁹ In 1978 India was the top world producer, at 165,323 mt. Brazil was second with 77,000 mt, and Mozambique was fourth with 61,000 mt produced. (Source: FAOSTAT.)

⁷⁰ Deloitte and Touche (1997).

sold to local entrepreneurs, and there was a push to invest heavily in new large-scale mechanical processing facilities. However, the Government also introduced high taxes on raw nut exports (18–22 percent), again with the goal of pushing the entire industry toward domestic processing. The reforms failed, and the sector essentially collapsed by 1994.

In 1995, the Mozambican Government liberalized the cashew sector, in part as a condition for World Bank loans. The reduction of export tariffs (from 18–22 percent to 20 percent, then again to 14 percent) did slightly increase prices at the farm gates, but, some argue,⁷¹ also led to the closing of most proces-

Figure 29: Weaknesses in the Mozambique Cashew Domestic Value Chain



sing factories. Mozambican processors found it too difficult to compete with traders selling raw nuts to India (where the industry is reportedly subsidized). Additionally, because of low margins, processors did not reinvest or maintain their plants; equipment began to break down, and improper and improperly maintained equipment caused a high level of cashew kernel breakage.

With most processing factories closed, and 7,000 employees out of work, Mozambique's cashew sector suffered another relapse. By the late 1990s, instead of having a vibrant value-added cashew sector, Mozambique exported most of its nuts raw to India for processing and value-added. In 2001, local businesses pressured the Government to reinstate the export tax on raw nuts.

Concerned about the sector, USAID commissioned a cashew sub-sector analysis⁷² seeking innovative means to

industry revitalization while maximizing benefits to small growers. The study results suggested that small and medium processors could bring value back to Mozambique by processing some of the raw nuts that otherwise would be exported to India. Thus, a program was implemented with the goal of establishing a profitable SME processor in the northern province of Nampula that would serve as the pilot. If successful, the processor could serve as the model for other, similar enterprises.

The Lead Firm Model

Source: J.E. Austin Associates. Inc.

A local businessman, Antonio Miranda, possessed many of the characteristics hoped for in an entrepreneur. He was innovative, thrifty, socially conscious, and had vision. With the support of the technical assistance provider,⁷³ a small-medium-scale hand processor, Miranda Caju, Ltd., was established in 2001 on the grounds of a previously closed facility. The building was reconstructed using local labor, which provided jobs to the surrounding community. Mr. Miranda was able to raise funds for seed and working capital (USD 47,000) in the form of a Guarantee Fund from INCAJU, Mozambique's National Cashew Institute, and he installed new equipment for whole nut production.

In hand processing, each nut is steamed, shelled, dried, and peeled by hand, then pre-graded to ensure a higher percentage of whole kernels. After kernels are vacuum-packed, they are trucked to exporters in the port of Nacala. The plant had the capacity to employ 460 workers, purchase 12,500 mt of raw cashews from small growers, and process 1,250 mt of cashews that rivaled India and Brazil in quality.

⁷¹ McMillan, Margaret. 2003. ["]When Economic Reform Goes Wrong: Cashew in Mozambique."

⁷² The sub-sector analysis was conducted by U.S.-based NGO Technoserve.

⁷³ Technoserve.

Case Study: Identifying and Implementing Replicable Business Models – Mozambique Cashews

When Mr. Miranda announced his intention to hire locally, 1,000 candidates applied to fill the 70 original positions.

Miranda Caju experienced and overcame many challenges. Within months, the processing plant was selling to a major buyer in Holland. Not only was the plant making a profit, but Miranda Caju workers earned, on average, USD 300 per year vs. the average of USD 8 per year in cash income earned by subsistence farmers.

Miranda Caju purchases raw nuts from several sources, including recently formed farmer associations and small growers. Before the plant's entrance into the market, most small growers sold to traders, who then sourced to wholesalers. Miranda Caju was able to locate small growers with the capacity to improve

their growing methods to provide better yields and, thus, higher quality raw nuts to the Miranda plant. Miranda Caju's entrance into the market benefited small growers by growing their incomes an average of 20 percent.

Replicating the Model

Recognizing the potential for additional smallmedium processors to enter the cashew sector, TechnoServe arranged for other entrepreneurs to be trained on the Miranda Caju premises to learn first-hand how to run a cashew plant. Mr. Miranda was an instrumental trainer, providing dav-to-dav experiences and lessons to the entrepreneurs learning the business. His ability to share lessons learned on a variety of topics like operations, cost savings, plant location, financing, hiring practices, and sourcing raw nuts from growers provided a sound foundation for other entrepreneurs to launch their own processing facilities in 2004. Also in 2004, Miranda Caju opened a second operating plant to meet the growing demand for processed cashews.



Source: J.E. Austin Associates, Inc.

Miranda Caju served as the model for five additional processing plants to open between 2002 and 2004, with a total processing capacity of 8,850 mt.

Figure 30: Mozambique Cashew Domestic Value Chain with Small Processors

Name	Location	Started	Processing Capacity, (mt)	Employees	Small Grower Purchases
Miranda	Namige,	2001-2002	1,250	460	12,500
Caju	Mogincual				
Africaju	Namialo,	2002-2003	1,500	160	10,000
Lda.	Meconta				
IPCCM	Murrupula	2002-2003	1,000	84	10,000
	Sede				
Miranda	Angoche	2003-2004	1,500	230	5,000
Caju	Sede				
Alexim Lda.	Iuluti,	2003-2004	1,000	63	1,200
	Mogovolas				
Moma Caju	Mecone,	2003-2004	1,000	63	1,200
Lda	Moma				
Macia Caju	Macia,	2003-2004	1,500	70	1,500
	Macia				

Table 15:	Mozambican	Cashew	Processing	Operations
Table 15.	102amorcan	Cushew	1 I UCCooming	operations

Source: TechnoServe

Conclusion: Rebirth of a Sector

Since 2001, twelve processors have now opened under the profitable, replicable business model demonstrated by Miranda Caju. Each is in varying stages of development and success, with Miranda Caju continuing to lead the small-medium cashew processor market. Of those businesses, Miranda Caju continues to grow and innovate. The firm also hopes to increase the percentage of raw nuts it purchases from farmer associations (currently 40 percent), and it now provides on-farm technical assistance to its small growers in the form of help with seedling re-planting, quality control, and improved yields.

The entire value chain has benefited from these interventions, not only in increased volume of quality cashew processing and exports, but also from realizing that profitable manual processing businesses can be created and can improve quality, create jobs, and rebuild the Mozambican cashew brand. Access to investment and working capital has improved, initially through INCAJU's Guarantee fund, and later by other guarantee funds managed by the Ministry of Industry and Commerce. Since 2001, these processors have annual sales over USD 5.1 million and employ over 3,000 workers. Smallholders are also benefiting through the increased prices they receive at the farm gate.

Mozambique's cashew industry is examined further as a case accompanying Tool 7: Horizontal Collaboration - Creating and Taking Advantage of Economies of Scale.

Tool 6: Capturing Value Through Forward and Backward Integration

At the firm level, vertical integration means creating forward and/or backward linkages. A firm becomes more vertically integrated when it takes on more of the activities that take place within its value chain. Vertical integration makes sense if, for example, the business is seeking to ensure supply or otherwise control inputs, capture more value, achieve economies of scale, or ensure access to information. A value chain, by consequence, becomes more integrated through the decisions of firm-level actors.

Vertical integration also takes place at the value chain level when more stages are brought into the country's value chains. This means that a nation's businesses are taking on more of the activities within the global value chain, which adds value, provides more market contact and information, creates employment, etc. Value chain members must decide whether it makes sense for them to integrate – a calculation that includes profit, risk, investor wishes, and other factors. Integration can also mean adding more functions to the value chain and not necessarily incorporating them from other countries. This integration may involve joint commitments or even joint investments.

Companies and enterprises have always made conscious decisions about whether it makes sense to vertically integrate. At the value chain level, one can even think that, before there was international trade, every industry was vertically integrated with a region or nation. This is a tool that examines whether there are gains to taking on more of the value chain's functions.

Figure 31: Firm Value Chain



The value chain understood at the firm level can be very simply illustrated (and, of course, can also be shown with much greater detail). In the accompanying figure, each stage represents a link in the value chain. In terms of integration, a firm operating in the production stage can assess its distribution channels and decide that it can transport its own goods more cheaply and efficiently than by using another firm. This firm has then made the decision to integrate forward and take on its value chain's distribution function.

The firm could also make decisions regarding its supply of inputs. At the production stage, a firm might recognize that its suppliers are not producing inputs that are adequate for use at the production stage. Rather than share information and work with one or a group of suppliers, the firm might simply integrate backward and develop its own ability to provide the inputs it needs.

Vertical Integration from the National Value Chain Perspective

For most products in most developing countries, relatively simple value chains focus on getting a product (typically a largely undifferentiated commodity) to market (whether that market is

domestic or international). From this perspective, the challenges are to achieve low transaction costs, push volumes of product to the market (and sell them), and reduce losses from spoilage, waste or theft. From the perspective of the in-country value chain participants, such value chains tend to be supply-driven and production-driven (a situation that poses its own challenges, if the producer's interests are not in line with market requirements). Also, the recipe for business development from this perspective involves reducing production and logistic barriers and removing taxes and levies (or imposing them to protect against cheaper competition or foreign competitors who are willing to buy raw materials at slightly higher prices). A "market information gate" that prevents domestic producers from understanding the export market is often present, thereby presenting both a challenge and an opportunity for forward integration (See accompanying figure).⁷⁴



Figure 32: Improving Value Chains (Before Value Added)

Source: J.E. Austin Associates, Inc.

⁷⁴ Forward integration by domestic producers often means that they need to access more information about the market in order to produce products that will sell. There are many ways of doing this. In order to gain a foothold in the export market, the exporter should learn as much about the market as possible and communicate that information to suppliers. Exporters could also pay the producers (and intermediaries, e.g., transporters) a premium to produce to certain quality and delivery standards. From the perspective of producers and intermediaries, who also want to sell more, they should do everything possible to learn about the market's (and the exporters') requirements. This sharing of information will add value and help value chain participants produce the right products in sufficient quantity.



Figure 33: Improving Value Chains (with Value Added)

One typical competitiveness goal is to increase quality and service within the value chain and to offer customers a desirable, higher-value product or service (shown in the accompanying figure as X+Y). This, if achieved, can offer huge increases in productivity and make the value chain and the country as a whole more competitive. (See Tool 4, Upgrading and Deepening the Value Chain.)

An even more rewarding goal is to add value along the value chain (shown as X+Z).⁷⁵ This can be fairly simple, e.g., coffee washing, packaging versus bulk tea, filleted fish, cut and packaged flowers delivered directly to retail outlets, or furniture or furniture components. It is analogous to the Mongolia meat production example described earlier in Tool 4, Figures 23 and 24. These value additions contrast with, for example, selling unprocessed logs or lumber, or collecting and grading vegetables for delivery to supermarkets. Value addition can be accomplished by introducing new enterprises into the value chain (thereby deepening it), or by having existing firms take on new activities (vertical integration). Adding value might also mean creating domestic operations that had previously been performed abroad. With cut flowers (as in the Uganda example), this might include assembling bouquets and packaging for display. In many value chains, it might take the form of producing intermediate or final products rather than raw material (e.g. rubber gloves rather than crepe rubber, processed or canned fruits rather than fresh or frozen exports, or filleted and packaged meats or fish rather than frozen products).

Source: J.E. Austin Associates, Inc.

⁷⁵ In reality, different businesses in competitive, market-driven economies generally participate in a variety of value chains – some simple, some offering higher quality, and some offering considerable value addition. Within an economy, businesses make their own choices as to their business model.

Considerations for Integration

There are many reasons to think about integration and several considerations that must be kept in mind when thinking about it.

First is strategy. It is important to understand the context in which the firm or value chain operates and whether it can become more competitive through integration. Once a firm understands its position in the value chain, it can determine whether there are profitable opportunities for expansion into adjacent links. A similar calculation must be done when a value chain seeks to vertically integrate through collaborative approaches. There are many strategic questions to ask about integration, and these vary depending on which stage the firm occupies in the value chain. Jim Austin's book, *Agroindustrial Project Analysis*,⁷⁶ long a touchstone for investors and practitioners, lists important questions to ask in analyzing agricultural projects. Some are related to vertical integration and are relevant mainly to the production stage of the value chain, including:

- Is there competition in procurement among similar agro-industries?
- How much quality control would be gained if the processor integrated backward to assume the production, storage, transport, and handling functions?
- How do improvements in quality control compare with the cost and with the alternatives for quality control?
- Should the producers integrate backward and undertake transport or production, or both?
- Would backward integration lower the costs of raw materials?
- How much will control of quantity, quality, and timing improve with integration?
- How far back should the producers integrate?
- How much additional fixed investment will be required to integrate?
- How much additional working capital is required?
- How might integration reduce the project's flexibility in obtaining sources of raw materials?
- What are the economic and operational risks of a decrease in this flexibility?
- How will integration affect variable and fixed costs?
- How will integration affect the plant's break-even point?
- Is integration politically feasible or socially desirable?

Second are operational considerations. Vertical integration may be necessary if a firm needs to be able to control its supply of inputs. This need for control is especially relevant if the firm needs highly specialized inputs for its products. Also, by controlling its supply, a vertically integrated firm can more easily ensure availability. Although suppliers can often provide inputs at a lower cost, especially if there is competition within that stage, a vertically integrated firm may actually lower its costs by providing its own supply.

⁷⁶ James Austin Agroindustrial Project Analysis. The Johns Hopkins University Press. 1981

Box 8: Benefits of Vertical Integration—ZEGA and Zambia's Horticultural Value Chain

Created in the 1980s to Support Diversification into Higher-Value Horticultural Products In the 1980s, Zambia's three largest horticulture exporters created the Zambian Export Growers Association (ZEGA). Zambian exporters saw opportunity to use the country's considerable natural advantages to produce and export higher-value horticultural products. ZEGA's founders recognized that they needed critical mass to purchase inputs from South Africa and to negotiate duty-free incentives with the Zambian government. ZEGA was established without any donor support, but, as it evolved, it became an important vehicle for donor support.

Grew in the 1990s by Developing Competence in the Freight Business

Perhaps the main benefit of grower cooperation was to secure air freight. Zambia's air-freight export tonnage was always less than in competing countries such as Kenya and Zimbabwe, so achieving critical mass to secure competitive rates and capacity was difficult.

In the 1990s, some of Zambia's big horticulture growers started to create linkages with flower markets in Holland and large food retailers in the UK. With these linkages to sophisticated markets, and with perishable products at stake, the Zambian growers realized that getting reliable, affordable air freight service to Northern Europe would be key to its long-term competitiveness. During the 1980s, Zambia Airways offered subsidized air-freight rates to Europe for the horticulture exporters, but the airline became insolvent in the late 1980s. ZEGA instead negotiated its first freight contract with British Airways, which nonetheless put Zambia at a cost disadvantage compared to other horticultural exporters such as Kenya and South Africa. However, ZEGA members stood firm; even though they could



Figure 34: ESA Rose Exports to the European Union (in tons)

Source: UNCTAD: United Nations Conference on Trade and Development. www.unctad.org

sometimes get cheaper freight rates on passenger planes, they continued to use ZEGA for freight services so it could amass tonnage. Once producers reached a sufficiently large tonnage of horticultural produce, ZEGA was able to negotiate with African freight carriers to achieve cost-competitiveness. Over time, ZEGA developed a competence for managing freight firms, and now has at least one air-freight shipment of horticulture products going out a day. ZEGA even sought to break off the freight business separate into a company, although it

continued to grow its expertise and service its clients.

Began Cold Storage at the Lusaka International Airport

Because of EurepGAP standards, the horticulture value chain in Zambia needed to have an integrated cold chain from the farmer to the European importer. The large growers invested in refrigerated trucks while ZEGA handled cold storage and logistics at the airport. This created a full, integrated cold chain for floriculture and horticulture export products.

Backward Integration into Procurement of Inputs

ZEGA supported backward integration by buying farm inputs, such as fertilizer, which has provided the industry with two advantages: bulk purchasing power, and thus lower input costs, for the sector and important working capital because ZEGA sells on account. Because ZEGA controls the important link to air transport, growers have a powerful incentive to influence other growers to pay their debts.

Results

The vertical integration within Zambia's value chain, led by ZEGA, enabled the sector to increase exports of fresh vegetables and cut flowers from USD 6,000,000 in 1994 to over USD 33,000,000 in 2001 and USD 43,000,000 in 2005. This increase in exports was enabled by ZEGA's forward integration into air freight and backward integration into procurement. ZEGA has filled important gaps within the Zambian value chain and has enabled all value chain participants to benefit from economies of scale. ZEGA has also been a recognized "face" of the industry in dialogues with the Zambian government and the donor community in receiving technical assistance to support Zambia's export growth.

Author: Carlton Jones – J.E. Austin Associates

Box 9: Bulgaria Wine—Integrating Operations to Secure Sourcing of Raw Material

The Bulgarian wine industry is closely linked to Bulgaria's grape-production subsector. The traditional value chain (as represented in the accompanying figure, "Bulgarian Wine Value Chain: Traditional") illustrates how winemakers rely on the grape auction and imports for their grape supply. However, a severe frost and low temperatures in 1997/98 significantly restricted grape production at Bulgarian vineyards. More than 50 percent of vineyard production in Northern Bulgaria and 20 percent in Southern Bulgaria were lost. Wineries in the north were forced to turn to suppliers from Southern Bulgaria and Romania.

This increased competition for grapes throughout Bulgaria and caused prices to rise quickly. Consequently, Bulgarian winemakers were forced to pay more for their Bulgarian grapes or to import grapes from neighboring countries with little control over the quality and variety of grapes. The poorly controlled imports





Source: J.E. Austin Associates, Inc.

of grape production and has made plans to expand to 350 hectares in order to secure a reliable and costeffective supply chain.

As shown in the accompanying figure, VA also now secures a portion of its grape supply by offering small vine-growers short-term contracts for their grapes and preferential pricing on wine. In the contract's





Source: J.E. Austin Associates, Inc.

negatively affected wine production.

The firm Vinzavod-Assenovgrad $(VA)^{77}$ is located in a region famous for growing grapes. Prior to 1997/98, VA had never had problems securing grapes from local suppliers for its production cycle.

The grape shortages made the company aware of the need for measures to ensure local supply to maintain quality and varietal differentiation. VA management decided to develop its own vineyards and to create new contract mechanisms to secure grapes from local producers. The company invested in 200 hectares

⁷⁷ VA's corporate website: <u>http://www.mavrud.com/en/index.htm</u>
inaugural year VA sourced 40,000 Kg of raw grapes from local small holders. However, in subsequent years VA allowed purchases under the new mechanism to decline, once again relying on the grape auction as the primary avenue for domestic purchases. While the new contract mechanism did not radically restructure VA's supply chain, it did institutionalize a new fail-safe for securing local production in years of scarcity.

Authors: Michael Gorman and Martin Webber – J.E. Austin Associates

Integration must make basic financial sense: a firm or country must evaluate whether value is being generated in adjacent links and whether it can capture enough of that value to make integration profitable. The firm or country should perform a cost-benefit analysis to identify whether the benefits outweigh the costs.

Why Integrate?	Considerations
Control Reasons	• Insufficient supply
	• Inadequate supply
	• Need for highly specialized inputs
	Cost savings
Lack of Intermediation	• Ineffective communication and flow of information
	between members
	 Nonexistent technical/embedded services
Establishment and	• Solidify position in value chain
Expansion	• Lower costs through economies of scale
	Alter competitive landscape
Capture More Value	Opportunities to increase revenues without
	overstretching resources
	• Opportunities to undertake more functions without
	overstretching resources
	Opportunities to create value

Table 16: Some Reasons to Consider Vertical Integration

Case Study: Capturing Value Through Integration—The Ghanaian Pineapple Industry and Blue Skies Holdings⁷⁸

Introduction

Vertical integration generally refers to a firm's ownership or control of vertically related activities. The greater the firm's ownership and control over successive stages of the value chain for its products, the greater its degree of vertical integration. In a value chain context, vertical integration can also be achieved between upstream and downstream firms when there is a high level of integrated systems and information sharing.

One case in point is Blue Skies Holdings Ltd., an example of the many successful, vertically integrated exporters of pineapple and other processed fruit from Ghana.

The Ghanaian Pineapple Industry

According to the Food and Agriculture Organization (FAO), there are more than 80 countries producing approximately 17 million tons of pineapples. More than 11 million (65 percent) of the 17 million tons grown are destined for export.⁷⁹ Pineapples are exported in various forms, and nearly 80 percent of pineapples are found on the market in processed form: 48 percent as juice and 30 percent as canned fruits. Thailand, Brazil, the Philippines, and India are the main producing countries. Thailand, the Philippines, and Indonesia account for 80 percent of the canned pineapple industry. Brazil's production is essentially consumed domestically, as is India's.⁸⁰

The fresh pineapple market has traditionally been dominated by Côte d'Ivoire, Costa Rica, and the Philippines. Growth in the fresh pineapple industry has averaged 6 percent per year since 2000. Political instability in Côte d'Ivoire has caused its pineapple export growth to fall to 2 percent over the past five years, and it has lost significant market share to Costa Rica and Ghana. In fact, during this same period, Ghanaian exporters have achieved an average growth rate of 45 percent while Cost Rica has nearly

doubled its exports to Europe. Costa Rica's success has centered on its usage of the preferred MD2 varietal, as well as the logistical competence and marketing prowess that the multinational Del Monte Foods employs to service the European market.

Ghana's West African location provides a beneficial climate for growing fruits and vegetables, as well as advantages for servicing the European market. Ghanaian fresh pineapple exporters face fierce competition from Costa Rica, Côte d'Ivoire, Honduras, Mexico, Ecuador, and others. Costa Rica is the leading exporter of pineapple to the

Figure 37: Pineapple Exports



Source: FAO STAT

⁷⁶Authors: Michael Gorman and Martin Webber - J.E. Austin Associates

⁷⁹ Food and Agriculture Organization of the United Nations, FAOSTAT located at: http://faostat.fao.org/default.aspx

⁸⁰ "The World Pineapple Market: when growth goes hand in hand with diversity", FruitTrop March 2003 – No. 100.

European markets with an annual export of about 300,000 tons, followed by Côte d'Ivoire with exports of 150,000 tons, while Ghana has just reached the third position with 71,000 tons.⁸¹ Ghana's main competitive advantage over Costa Rica is its location, while its political stability and business environment compare favorably with Côte d'Ivoire.

Overall, Ghanaian pineapple exports to European markets have increased in recent years thanks to greater investment in the fruit-export industry from both the private and public sectors. Regulatory reform, tax incentives, market linkages, investments into new varietals, public-private partnerships, and new economies of scale within the value chain have all helped to expand the Ghanaian pineapple industry.

Background

European Market

The European market for fresh produce has been expanding with the rising incomes of European consumers. In 2005, the European market imported approximately 600,000 tons of pineapples of which one-quarter are estimated to be fresh pineapples.⁸² However, several changing factors are driving the transformation of the produce sector including supermarket strategies, food safety legislation, supply chain integrity, rationalization of the supply base, and innovation. Consolidation of European supermarkets has, in part, been driving these changes. By the early 2000s, seven of the largest food-retailing chains in Europe accounted for 76 percent of fresh fruit and vegetable sales and 70–90 percent of fresh produce imports from Africa.⁸³ In the UK, supermarkets are even more concentrated, with just four big chains accounting for 73 percent of sales at supermarkets and convenience stores.⁸⁴ Thus, the EU-SSA pineapple-export supply chains are characteristic of buyer-driven global commodity chains; the European supermarkets increasingly demand products that are low-cost and quality-certified (resulting in higher profits via the use of branding), as well as new methods of marketing differentiation.





⁸¹ GEPC News, http://www.gepcghana.com/news.php?item=4&n=.

⁸² Vagneron, Faure, Loeillet, "Is there a pilot in the chain? Identifying the key drivers of change in the fresh pineapple sector" CIRAD, 2005.

⁸³ Hallam, Liu, Lavers, Pilkauskas, Rapsomanikis, Claro, "The market for non-traditional agricultural exports," Commodities and Trade Division, FAO, 2005.

⁸⁴ "Big chains enjoy a buyer's market," *The Economist*, February 15, 2007.

European populations with higher disposable incomes have increasingly been demanding high-standard, certified, and fresh fruits and vegetables on a year-round basis. The European market is characterized by several relevant market factors and trends:

- Increases in wealth and demand for high-value products (many of which are imported)
- Exotic produce market growth (see accompanying figure)
- Diversified preferences: national and ethnic
- Revolution in market structure: retail-outlet dominance, foodservice, and catering
- New marketing formats: pre-packaging and pre-cut vegetables
- Consolidation of sources: direct linkages and faster market response
- Regulatory environment change; lower tariffs but increased safety constraints
- Value chain integration and just-in-time inventory
- Logistics networks: sea-freight and air-freight capacity
- Niche markets; bio-certification and fair trade

In Ghana, fresh pineapple is exported by 60 companies, although more than 50 percent of the total export volume is produced by the larger companies, such as Jei River Farm, Farmapines, Koranco Farms, and Prudent Farms. Exporter association organizations like the Seafreight Pineapple Exporters of Ghana (SPEG), the Horticultural Association Ghana (HAG), and the Exotic Fruit Exporters Association (EFEG) work to help Ghanaian exporters service the growing European fresh produce market. Processing companies such as Blue Skies, Tonggu Fruits, and First Catering export fresh-cut pineapples to high-quality retailers such as Marks & Spencer and Sainsbury's.

Ghanaian fruit exporters source their products from both commercial farmers (70 percent) and small-scale farmers (30 percent), and the large commercial farming enterprises are typically operated as outgrower schemes. While recent market conditions have led export companies to increasingly invest in vertically integrated outgrower operations, nearly all exporters mitigate risk by buying fruit from small-scale farmers to supplement commercial farms' production. By sourcing from small-scale producers throughout the country, exporters ensure a diverse supply chain that is not as susceptible to weather, crop sickness, or other unpredictable risks.

Ghanaian Exports to Europe

During the late 1980s and early 1990s, Ghana carved a niche significant in the European Union (EU) market as a primary supplier of topquality. air-freighted pineapples, supplying about 60 percent of the estimated annual 20,000 mt of pineapples air-freighted to Europe.⁸⁵ Throughout this period, only a small amount of

Figure 39: Ghanaian Fresh Pineapple Exports



Source: Voisard and Jaeger, Ghana Horticulture Development Study, and SPEG (2004 figures)

pineapples were sea-freighted (2,710 tons; see accompanying figure). The air-freight market's success was due to Ghana's liberal market regulation, efficient handling services, and the diversity of available and relatively inexpensive flights to all parts of Europe. However, during this period, pineapple export operations remained segmented and spread among many participants, and little integration was accomplished. Ghanaian export volumes were eventually restricted by the limited air-freight capacity of commercial passenger aircraft.

⁸⁵ Partnerships for Agribusiness Development, Agricultural Trade, and Market Access by TechnoServe for NEPAD, November 2004

The Blue Skies Holdings Ltd.⁸⁶ Value Chain

Established in 1998, the Blue Skies Holdings processes fresh, chilled pineapple, mango, watermelon, passion fruit, and papaya for export (see accompanying figure). Blue Skies' products are certified to meet the EurepGAP protocols for quality compliance. Fruits processed by Blue Skies are sourced mainly from Ghana's eastern and central regions, with supply gaps being filled by imports. The Blue Skies value chain is highly integrated, having incorporated operations from buying and transporting raw pineapples to delivering certified, cleaned, cut, packaged, and branded pineapple pieces to UK distribution networks.

The company has made excellent use of its owner's networks within the EU supermarket industry to achieve market access for its products.

Since 2000, the company has grown tremendously, expanding its value chain by incorporating additional operations into its processing facilities. It has also expanded its local employment. Blue Skies started operations with 38 workers and has since increased its workforce to 1,000, over 60 percent of whom are permanent staff. It has also replicated its business model with similar ventures in South Africa and Egypt. By providing good extension services and farmer training, and by offering higher prices, the company has rapidly increased its production from one to about 35 tons per week. Blue Skies is also known to pay its farmers promptly and at a higher price per kilogram than other pineapple buyers in the Nsawam area of Ghana.

Blue Skies does not provide credit to farmers, nor does it link them to any financial agents; its prompt payment and higher prices are sufficient incentives for farmers to invest in their farms. In addition, the company has assumed technical and financial responsibility for certifying all its suppliers. It also provides inputs and equipment to its producers for purchase. Because of the investments needed to get certification, those suppliers whom Blue Skies has helped obtain EurepGAP accreditations are obliged to sell to the company. Recently, 57 percent (77 of 135) of Blue Skies' small-scale pineapple producers were certified as organic, Fair Trade producers. Blue Skies operates with individual farmers and not cooperatives.

Several factors contributing to Blue Skies' success reflect concepts discussed throughout this Guide, such as high levels of trust, sharing of information, innovation, value addition, positioning collaboration, and risk mitigation. Blue Skies emphasizes prompt payment to farmers, the provision of training and education on EurepGAP standards, certification of farmers, interest-free loans for dedicated farmers, and willingness to improve local road infrastructure to improve access to farms by company trucks.

Conclusion

Blue Skies has been able to take advantage of strong market linkages with the European grocery sector, access to information on innovative practices, and foreign capital and expertise. It has implemented a highly sophisticated production and quality-control model and provided appropriate incentives and support to its suppliers. Blue Skies services a value-added portion of a large, highly competitive commodity market. Starting with the same raw material, the value chain and operating model of Blue Skies is very different from that which the Sea Freight Pineapple Exporters of Ghana has successfully implemented, which we will discuss in Tool 7.

Blue Skies Operating Schedule

In less than 48 Hours, Fresh Organic Pineapples Are Harvested in the Mountains of Ghana and Delivered to UK Consumers

Thursday

9:00 AM: Pineapples are harvested in Akwapim Mountains (62 miles west of Accra)

⁸⁶ http://www.bsholdings.com/

Case Study: Capturing Value Through Integration – The Ghanaian Pineapple Industry

10:00 AM:	Men cut pineapples and women deliver to collection point
10:30 AM:	Pineapples are sorted by class at collection point
12:00 PM:	Fruit is loaded onto trucks and heads to Blue Skies factory 100 km away (2,000 pineapples per truck)
2:00 PM:	Pineapples arrive at factory and are processed
2:45 PM:	Pineapples roll off assembly line and are cleaned; "topped and tailed"; have their skin trimmed; weighed; sealed in Sainsbury-labeled (UK supermarket) tubs; put in holding chill and packed into cardboard boxes
7:00 PM:	Refrigerated load of pineapples leaves factory for 100 km journey to Accra
10:00 PM:	Boxes of pineapples are packed onto BA flight and take off for the UK
Friday	
5:45 AM:	Pineapples arrive in London and transit customs
8:45 AM:	Pineapples are taken to BA perishables-handling center outside of London
9:30 AM:	Pineapples are taken out of cold storage and quality is inspected again
11:00 AM:	Sainsbury truck picks up pineapples and takes them to the supermarket's distribution center 36 miles away
12:00 PM:	Fruit is sorted according to Sainsbury store orders
Saturday	
4:00 AM:	Delivery to Sainsbury stores made
5:00 AM:	Fruit goes on sale in organic section
7:00 AM:	European shoppers begin to purchase pineapples
9:00 AM:	European consumers eat pineapples for breakfast

Source: Blue Skies corporate website. www.bsholdings.com.

Tool 7: Horizontal Collaboration - Creating and Taking Advantage of Economies of Scale

Tool 6 focused on approaches to vertical integration within a value chain. Horizontal linkages also offer excellent opportunities for value chain participants to obtain scale advantages through interfirm coordination. By combining resources and sharing information, horizontal cooperation allows participating companies and producers to achieve improved quality, service, and savings through increased access to inputs, more leverage in sales negotiations, and greater ability to design initiatives that emphasize upgrading the value chain.

This tool focuses on the implementation of horizontal collaboration mechanisms and linkages among businesses to overcome problems stemming from small-scale activity by some individual producers or enterprises. This small scale prevents them, for example, from accessing inputs at optimum price, from reducing the uneconomical unit costs of extension services, or from enjoying enhanced market power stemming from increased volumes of production and nonfragmented marketing.

Typical benefits of collaboration to achieve scale in transactions or operations include reduced costs from activities carried out jointly and increased access to more and better inputs and services, which enables more professional management and marketing due to greater affordability. Looking further, though, horizontal collaboration among enterprises creates a collaboration platform that could later allow the chain to move toward forward or backward integration (discussed in Tool 6), or to achieve improved quality (Tool 9). Business-to-business collaboration is certainly an approach that enterprises can and should explore independently. There is also potential for development partners and other facilitators to assist the value chain participants in recognizing and creating collaborations.

The first step for enterprises looking to implement horizontal linkages is to identify areas in which they can operate jointly with increased efficiency and effectiveness. Typically, there will be many opportunities – in marketing, procurement of inputs, management, or logistics.

When enough enterprises have expressed interest in the idea of horizontal cooperation, some type of operational form, such as a simple agreement or a set of actions facilitated by a producers' association, formalizes the arrangement. The benefits of horizontal collaboration can lead to the creation of cooperatives, associations, or new companies, or toward the reorientation of existing organizations to perform the joint activity. For example, joint purchasing of inputs could form a basis for an inputs depot to be run by an association or as a jointly-owned commercial venture in which many of the enterprises are represented. The need to obtain quality extension services can similarly be satisfied through a more developed member services association or cooperative. Joint marketing and logistics can lead to the creation of a collection center, which could be responsible for increasing product volume sales and thus obtaining better prices while reducing transportation costs when consolidating shipments. The center could be operated as a company or within a cooperative.

Tool 7: Horizontal Collaboration - Creating and Taking Advantage of Economies of Scale

Economies of scale do not necessarily need to be developed through producer associations or cooperatives. Horizontal linkages are in many cases recognized, instigated, and organized by lead firms in the value chain.

A Step-by-Step Summary for Tool 7: Horizontal Collaboration - Creating and Taking Advantage of Economies of Scale

- Identify areas in which enterprises performing similar activities in the chain can operate jointly with increased efficiency and effectiveness. Typically, many will lie in marketing, procuring inputs, management, and logistics.
- Promote the idea to generate enterprise buy-in.
- Give operational form to the collaboration, such as through simple agreements, cooperative creation, new associations, or companies; reorient existing organizations to perform the joint activity.

Case Study: Creating and Taking Advantage of Economies of Scale— The Ghana and Côte d'Ivoire Experiences⁸⁷

Introduction

The case accompanying Tool 6 described a successful story of vertical integration within the Ghanaian pineapple value chain. This case now looks at a story of achieving economies of scale within the same industry. Economies of scale characterize a production process in which an increase in the number of units produced or managed generates a decrease in the average cost per unit. Achieving economies of scale is important when the minimum units required to access desired inputs, services, technologies, or other capacities are quite large. Such minimums are required to dissipate the high usage or acquisition costs of a service or facility over a larger number of inputs in order to increase efficiency.

The term "reaching scale" refers to attaining a level of production that allows the addition of further investment or the incorporation of additional operations. Where individual actors lack such scale, collaborative mechanisms can substitute. In Ghana, pineapple exporters were required to reach substantial volumes (scale) before they were able to access sea-freight transportation. This case describes their success.

Côte d'Ivoire: Office Centrale des Producteurs-Exportateurs d'Ananas et de Bananes (OCAB)⁸⁸

In 1999, Côte d'Ivoire was the largest horticultural exporter in West Africa, exporting USD 140, million of fruits and vegetables, primarily bananas and pineapples. A brief look at Côte d'Ivoire in the 1980s and 1990s helps to put Ghana's actions in perspective.

As a result of increased competition from Thailand, the collapse of state enterprises, and economic reforms that reduced subsidies, Côte d'Ivoire exports of canned pineapple and pineapple juice had practically disappeared by the late 1980s. In response, much of Ivorian pineapple production switched over to the export of fresh pineapples to Europe by sea-freight, using the same refrigerated freighters ("reefers") that are used to transport bananas. This move took advantage of Côte d'Ivoire's proximity to Europe, a factor much more important in the fresh-pineapple trade than in the market for canned pineapple given spoilage considerations.

However, by the late 1980s, Côte d'Ivoire began to lose market share to Central American and Caribbean fresh-pineapple exporters. After supplying close to 90 percent of the European market for fresh pineapple in the mid-1980s, Côte d'Ivoire's market share fell to two-thirds in 1990.⁸⁹ But the 1990s also brought several changes favorable to the Ivorian fruit and vegetable export industry.

First, in the 1990s, the Office Centrale des Producteurs-Exportateurs d'Ananas et de Bananes (OCAB) was formed to represent the interests of exporters, set quality standards, and facilitate communication. OCAB has reduced the number of "approved" exporters of fruit in an attempt to maintain quality standards. It also organizes the charter of refrigerated ships to transport bananas and pineapples to Europe. Second, in 1993, after much debate, Europe harmonized its banana-import policies to make way for the single European market and continued granting former colonies preferred access to its markets. Third, the 50 percent devaluation of the CFA franc in January 1994 helped stimulate the economy, particularly the export sectors.

The net impact of these three factors was that banana exports grew from 95,000 mt in 1990 to 215,000 mt in 1999, while fresh-pineapple exports expanded from 135,000 mt to 183,000 mt over the same period

⁸⁷ Authors: Michael Gorman and Martin Webber - J.E. Austin Associates

⁸⁸ Minot, N & Ngigi, M. "Are horticultural exports a replicable success story? Evidence from Kenya and Côte d'Ivoire," International Food Policy Research Institute and Egerton University, paper presented at the InWEnt, IFPRI, NEPAD, CTA conference "Successes in African Agriculture," Pretoria 1–3 December, 2003.

⁸⁹ Rougé, B. and M. N'Goan. 1997. "L'anana en Afrique de l'ouest et du centre." Acta Horticulturae 425: 75-82.

(FAOStat, 2002). Côte d'Ivoire became the second-largest fresh-pineapple exporter in the world after Costa Rica (Ti, 2000). It is estimated that approximately 35,000 people are employed by the banana and pineapple plantations.

In pineapple production, smallholders continue to dominate. 70 percent of Ivorian pineapple exports are produced by smallholders on farms of 0.5–10 hectares. The remaining 30 percent is produced by large plantations, including some owned by vertically integrated banana companies such as Compagnie Fruitière and Chiquita. One reason for the greater involvement of smallholders in pineapple production compared with banana production is that the initial investment cost of establishing a plot is estimated to be 3-4 times as great for bananas.⁹⁰

Figure 40: Ghana Pineapple Value Chain



SPEG Exporter Channel

Source: J.E. Austin Associates, Inc.

Several factors lay behind the past success of fruit and vegetable exports from Côte d'Ivoire. First, Côte d'Ivoire had long been known for its political stability. Second, President Houphouet-Boigny had, for the most part, supported agriculture-led growth. Third, Côte d'Ivoire had benefited from its proximity to European markets since it is just 8–10 days by sea freighter from Marseilles. Although it also benefits from frequent air connections with Paris, this factor was less important since most Ivorian fruit and vegetable exports have been via sea-freight. Fourth, the government had had relatively limited involvement in production and marketing, particularly in the horticultural sector.

However, with Côte d'Ivoire's current political instability, much of the fresh-pineapple exporting industry has shifted to more stable locations such as Ghana.

The Sea-Freight Pineapple Export Value Chain

The Sea Freight Pineapple Exporters of Ghana (SPEG) Association was formed in 1995 by Integral Ghana Ltd., Jei River Farms and John Lawrence Farms⁹¹ to develop sea-freight shipments of fresh pineapples from Ghana. SPEG identified Union

Bananière Africaine (UBA/Dole) of France that same year to provide freight services to the Ghanaian industry. The UBA boats are refrigerated vessels transporting bananas from Cameroon, and operators allocate space for Ghana's pineapples based on available free space after the banana loads. Travel times to southern and northern EU destinations are 9 and 13 days, respectively.

Since its formation, and the introduction of sea-freighting, SPEG has become a driving force in the Ghanaian pineapple industry and has been profitable from its inception. As a result, its membership increased from three in 1995 to 22 by mid-2005.⁹² Pineapple exports from Ghana have increased from 15,764 tons valued at USD 5,600,000 in 1995 to 57,392 tons valued at USD 18,300,000 in 2003. The percentage of sea-freighted exports to total pineapple exports increased from 17 percent in 1995 to 68

⁹⁰ Rougé, B. and M. N'Goan. 1997. "L'anana en Afrique de l'ouest et du centre." Acta Horticulturae 425: 75-82.

⁹¹ SPEG was formed with support from the Ghanaian government and USAID under the Trade and Investment Program (TIP). The TIP provided technical staff during SPEG's three formative years to oversee the coordination of sea shipments. Ghana's government leased a shed at the Tema port to SPEG solely for the consolidation of fruits before shipment.

⁹² Danielou, M. & Ravry, C. "The Rise of Ghana's Pineapple Industry," Africa Region Working Paper Series No. 93, 11/2005.

percent in 2003. Ghanaian pineapple exporters now have access to two vessels on a regular weekly basis to the European ports of Vado and Vendres in the south and to Port Antwerp in the north.

The availability of regular vessel services since 1995 has benefited all producers. Large-, medium- and small-scale producers have expanded production and generated increases in farm-level incomes and employment. The 10 largest exporters now control about 71 percent of total exports (more current information is being verified).93

Although initial pineapple exports from Ghana were a private-sector initiative, the Ghanaian government

supported the seaalso freighting by dedicating a facility at the Tema port for the consolidation of shipments. An process intensive planning involving Ghana's government and industry players is ongoing improve post-harvest to handling facilities at the Tema Port and the airport at the cost of USD 2,500,000 under the World Bank's program, AgSSIP.

Benefits from Business Partnerships

The most important partnership developed over the past nine

Bananière Africaine (now Africa

Figure 41: Ghana European Pineapple Exports



Express Line) arrangement for the sea-freighting of pineapples. This strong partnership has ensured the availability of regular vessel services since 1995.

Polycraft, a local carton manufacturer, is also working with SPEG to provide its members 35 percent of packaging requirements. SPEG also undertakes bulk procurement of agro-chemicals from suppliers such as Wienco, Chemico, and Dizengoff for distribution to its members. On average, SPEG budgets about USD 50,000 towards the purchase of agro-chemicals for its members. The prices that members pay are about 10 percent below market price but include a small margin to cover administrative costs.

SPEG's success in increasing Ghana's primary pineapple exports was at least partially due to the development of sea-freighting capacity. Another success factor in the industry was Ghana's proximity to the market.

The early air-freighting of Ghana's pineapple exports was attractive because the fruit could arrive at the EU market in six hours. Sea-freighting extended shipping periods to between 9 and 13 days to southern and northern EU ports, respectively. Ghana's logistical advantages over Latin American exporters (who have longer shipment periods of 16–20 days to the EU) were an advantage until recently, given European preferences for the MD2 pineapple, which Ghana was not producing.

In 2005, Ghana lost considerable market share due to its decision not to invest in the MD2 varietal. However, since the adoption of MD2, Ghana's pineapple quality and presentation are now up to EU standards and its export revenues have increased in per-unit value (see accompanying figure; this topic is further discussed in Tool 6).

⁹³ "Partnerships for Agribusiness Development, Agricultural Trade, and Market Access: A concept note for NEPAD" by TechnoServe, November 2004.

Conclusion

SPEG has been able to increase its share of the expanding European pineapple market by building scale through leveraging its relative logistical competitive advantages over its Latin American counterparts. SPEG is also working with its members to meet EurepGAP standards and implement new traceability, certifications schemes and other standards, and to provide other services.

Case Study: Creating and Taking Advantage of Economies of Scale Within the Mozambican Cashew Value Chain⁹⁴

Introduction

This case demonstrates how firms identified common areas for collaboration and created an association in the Mozambican cashew value chain that enabled the companies to achieve economies of scale in operations. Pointing back to the case in Tool 5, *Identifying Business Models for Replication*, interventions in the Mozambican value chain were required to revitalize the country's processing capacity. To that end, lead firms were identified to establish new processing centers, and once operational, this model was replicated and implemented by other processors. Those same processors experienced bottlenecks in the value chain and came together to address those constraints through economies of scale. These economies are realized when horizontal linkages occur, and the linkages are most likely to be successful when there is a common platform from which firms can operate. In this instance, Mozambican processors were willing to organize and identified areas in which to jointly operate, creating an association that carried out those shared functions.



Source: SA PLACE

Figure 42: Regions of Mozambique

Points to Consider

While reading this case, consider the environment and actors that drove the processors to consider organizing, the steps they took, and the benefits they realized once the association was established. There are three principal benefits to horizontal linkages:

- Lower costs
- Access to services or inputs that are not easily acquired by an individual processor
- Pursuit of initiatives that are difficult for processors to accomplish on their own

Also consider the services offered by the association, as they are good examples of marketdriven (demand-side) services that have a stronger likelihood of being sustainable.

Background

From the 1920s until the mid-1970s, Mozambique was considered the world's leading cashew producer (240,000 mt at its peak in 1973)⁹⁵ with a considerable domestic capacity for processing quality cashews. In 1978, the Government of Mozambique (GoM) banned the export of raw nuts

in an attempt to stimulate domestic processing. Within a few years, Mozambique possessed over a dozen processing factories and was the first African country to process cashews on an industrial scale. But by 1994, after a variety of events – including a civil war, the adoption of policies that fixed raw nut prices, and the continuation of the raw-nut export ban – the cashew sector collapsed (see accompanying figure).⁹⁶ At

⁹⁴Authors: Carlton Jones and Martin Webber - J.E. Austin Associates

⁹⁵ Technoserve.

⁹⁶ Source: FAOSTAT, INCAJU as illustrated in: *When Economic Reform Goes Wrong: Cashews in Mozambique*, McMillan, Margaret 2003.

the end of the civil war, those policies were abandoned, and the GoM hoped to reclaim its past reputation of progress by investing heavily in large-scale mechanical processing facilities. Unfortunately, the government also introduced a high tax on raw nut exports with the goal of pushing the entire industry to domestic processing.

In 1995, the Mozambican government liberalized the cashew sector to meet World Bank conditions for continued loans. (The rationale for this approach was highlighted in the discussion of Tool 5.) The reduction of export tariffs did, in fact, increase prices slightly, but it also led to the closure of Mozambican

factories. processing With those factories Mozambique's closed, cashew sector entered another decline. By the late 1990s, instead of a value-added vibrant cashew sector. Mozambique exported most of its nuts - raw to India for processing and value addition.

Seeing an opportunity USAID assist. to commissioned TechnoServe to conduct a cashew subsector analysis, seeking to identify innovative ways to revitalize the industry and maximize benefits to small growers. As discussed in Tool 5, a model version of a small hand-processing was designed, piloted, and, by 2004, replicated,



Thousands of metric tonnes



Wers. Source: FAOSTAT, Ana Machalela, INCAJU statistician, e-mail communication, July
1 5, a 2001; Raimundo Matule, deputy director of INCAJU, e-mail communications, September
small 2001 and May 2003. As adopted by McMillan, Welch & Rodrik. 2003. "When Economic plant Reform Goes Wrong: Cashew in Mozambique."

The first successful plant, Miranda Caju, provided the replicable business model on which all other plants in Nampula province were based. Afterward, even though processors were established and growing, they still relied heavily on technical assistance. In examining the sector's long-term viability, the processors recognized other extension-service needs that were not being met.

As other small hand-processors entered the market, stakeholders realized that, to ensure long-term sustainability of the entire value chain, these extension services needed to be fee-based rather than subsidized. Unfortunately, if the extension services were provided at full price, only Miranda Caju could pay the fees and still operate profitably. This scenario encouraged the processors to link horizontally to distribute the cost of extension services. Better still, this seemed to be a good business opportunity for a firm to provide these services to all processors as the sector continued to expand.

Mozambique Cashew Processors Link Horizontally

With the assistance of TechnoServe, Miranda Caju's president and founder, Antonio Miranda, considered by his peers to be an industry leader, brought together Nampula province's other processors to discuss jointly creating a private firm to provide services to everyone in the sector. This lead firm would provide valuable services to processors that would otherwise have a difficult time accessing them. These meetings led to the formation of Agro Indistria Associadas (AIA).

Services Provided by AIA

AIA started out in 2004 as a private-sector led services firm comprised of the seven processor firms in Nampula. Each processor contributed USD 500 of seed money to start the firm and was considered an equal owning partner. AIA selected a president, Ali Cherif Deroua, and now the provides services once provided by the consulting firm. Some services are fee-based, while others are non-fee-based. AIA's fee-based services fall into three main categories:

- Processing
- Distribution
- Marketing

Its processing services include importing non-nut inputs like packaging and machinery. Distribution services include



Figure 44: Domestic Value Chain for Mozambican Cashews

Source: J.E. Austin Associates, Inc.

warehousing and load consolidation at the Nacala port and completing and filing paperwork for export administration. In marketing, AIA provides fee-based services such as selling (order-filling) to global buyers and providing pre-shipment quality control. These quality-control measures include ensuring Hazard Analysis and Critical Control Point (HACCP) and EurepGAP compliance by adopting a "three strikes, and you're out" cut rate for suppliers. In this system, suppliers are encouraged to improve their quality measures or risk being excluded from the supply chain if they incur three quality infractions. Adopting this system has reduced instances of poor quality and increased export prices.

AIA provides the following non-fee services:

- Training
- Branding
- Advocacy

AIA provides marketing intelligence to its members through training seminars and advocates on behalf of members by engaging in public/private dialogue. In the past, these discussions have included negotiating with the GoM for targeted infrastructure investments and improving market regulations. With outside technical assistance, AIA is advocating for a gradual reduction of the export tax on raw nuts which, at 18 percent, penalizes farmers and hampers competitiveness. In addition to providing these services to existing processor members, AIA also actively recruits and trains new processors.

Branding is one of the most interesting services AIA provides. Again, with outside technical assistance, AIA created the "ZAMBIQUE" cashew brand. This brand was created to draw international awareness and interest to Mozambican cashews, especially from U.S. buyers.

How Horizontal Linkages Benefit the Mozambique Cashew Value Chain

With AIA now providing much-needed services to small Mozambican processors, the following benefits have been realized.

Improved quality control

Through AIA, quality has improved, and poor-quality claims have been reduced by over 50 percent. In 2005, 26 containers were cited as having quality problems, while in 2006, only 12 were. It is estimated that this reduction saved producers approximately USD 35,000 from 2005 to 2006.

Speedier and cheaper access to inputs

Before AIA, it was nearly impossible to ask for duty exemptions. To qualify for duty exemptions, a processor had to export all imported items within three months. This did not match normal business needs because processing volumes in a year's time were less than a container load, so processors typically could only import their packing needs once a year and not qualify for the exemptions.

Table 17: Cost of Packing Material Before AIA

Item	Cost Before AIA	Cost After AIA
Plastic	Bad quality and	Good quality and
Bags	repacking rate of 12	repacking rate of 5.3
	percent = USD 0.72	percent = USD 0.52
Cartons	3-ply (considered too thin) = USD 1.10	5-ply with 4 colors and printed Zambique logo = USD 0.085

Source: TechnoServe

With AIA, processors were able to combine orders to use temporary import-duty exemptions (IVA - 17 percent + duty 7.5 percent). AIA was also able to reduce shipping freight for its members. Before AIA, the average cost of exports per kg of kernels was USD 0.38, while after AIA, this cost dropped to USD 0.27, and then again in 2006 to USD 0.17.

Table 18: Cost of Shipping Before and After AIA

Item	Cost Before AIA	Cost After AIA
Container to	\$1,850	\$1,450
Rotterdam		
Port and Service	\$920	\$750-800
Costs		

Source: TechnoServe

Improved market linkages and information sharing

With AIA, processors now have a timely and credible supplier in the market. AIA carefully selects sea lines and transit shipments, while keeping daily track of all containers. Not only are these shipping records shared among AIA members, AIA also sends reports containing updates on sales, stocks, and receivables to banks on a weekly basis. This has helped banks feel more comfortable about the perceived risks associated with the sector and has made credit applications a little easier to file. This improved access to information would not have been possible for individual processors, although information is not the only thing shared among AIA members. Today, the improved communication and collaboration among AIA members has also translated into an atmosphere of cooperation between firms. When one member experiences shortage of an input, other AIA members pitch in to provide those supplies, recognizing that as each firm gets stronger, the entire industry benefits.

Improved industry image through "Zambique" brand

Time will tell how successful the introduction of a brand will be and how it will impact the Mozambique cashew sector. Early indications suggest that a unified brand has helped buyers recognize that small

processors in Mozambique have improved their quality. The brand has certainly improved the solidarity of members within the sector.

Lessons Learned

Horizontal collaboration mechanisms, such as those of AIA members, demonstrate the benefits of joint operation. Increasing producers' market power, enhancing their market linkages, and improving quality are all benefits that other industries can consider when seeking to improve their value chains through horizontal linkage.

Tool 8: Positioning Products and Value Chains for Greater Value and Competitiveness

Building on several of the tools already described, a competitiveness positioning assessment provides strategic direction to many of the actions that the value chain implements to improve its competitiveness. This assessment and collaborative decision-making inform the value chain's leadership about the value chain's product offerings, relative to competitors, in key markets. It also forms the foundation for developing a strategic vision and a clear, actionable plan for repositioning the industry in current markets and for penetrating new ones.

Positioning the Value Chain.

In a simple 2 x 2 matrix, the product or service of a company, value chain, or industry can be described in terms of product scope (complexity, value addition) and product differentiation (special qualities, lack of direct competitors). Products and services in the lower left are basic commodities, subject to severe price competition and very restricted profit margins. Products at the upper left have a lot of value added but may still be under severe price competition. Products to the extreme right are highly differentiated and, in that respect, may be able to command higher prices and margins but have little value added. Of course, products and services at the top right are in "competitiveness nirvana," commanding high prices and margins because of their complexity and special qualities in the customer's eye – all assuming, of course, that there is sufficient market demand to interest the value chain participant.

One challenge for value chain participants is to decide where to position the chain's products or services. Where is the product currently positioned in the domestic or global market? Where do we want it to be – and what are the quality, service, and other requirements to be competitive in that positioning? What are the best global performers in any of the positions doing to be competitive? What are the profit margins and likely market demand at each position? Who will our competition be?

Box 10: Ecuadorian Cacao—Positioning the Value Chain for Greater Value and Competitiveness

In some cases, potential value exists in the value chain and needs only to be unlocked. This can take the form either of movements toward higher scope or toward product differentiation. Ecuador's cacao industry illustrates this principle.

Ecuador's cacao industry has made advancements in processing and value added that enabled producers to export higher quality cacao and receive premium prices. The improvements in value enabled a repositioning of the higher quality portion of the Ecuadorian product, and these are being supported by promotional and market channel actions. Historically, Ecuador has received a USD 20–100 premium over the baseline market price. However, the flavor profile of Ecuadorian cacao is so desirable in today's consumer market that it currently commands a premium of USD 800–1,200 per ton.⁹⁷

⁹⁷ Author interview with Karl Walk, purchaser for Blommer Chocolate.

Country	Premium over Market Price, 4 May,
	2007 (per ton, USD)
Venezuela (dried and fermented)	2,000
Java (dried and fermented)	2,000
Ecuador (dried and partially fermented)	800–1200
Ivory Coast (dried and fermented)	320
Dominican Republic (dried and unfermented)	280
Ghana (dried and fermented)	250

Premiums for Several Countries' Cacao Products

Source: Blommer Chocolate

Among the companies currently buying cacao from Ecuador is Blommer Chocolate Company, a large concern that purchases raw cacao for grinding and processing into chocolate products. It is the largest buyer in Ecuador and the largest raw cacao processor in the United States. Blommer customers include large, well-known companies and brands such as Mars, Nestlé, and Hershey, among others. Blommer is pleased with the quality of cacao that they are able to purchase from Ecuadorian producers who have been trained through the Farmer Field Schools. Blommer previously used Ecuadorian cacao in a wide variety of dark-chocolate products, but because of its premium qualities, Blommer has recently been using Ecuadorian cacao for producing high-end, single-origin dark chocolate.

Single-origin dark chocolate can be a branded, value-added product because buyers are looking for the specific Ecuadorian flavor profile. One indication of the repositioning of the value chain toward greater value and competitiveness is the fact that single-origin chocolate from Ecuador is being sold in international markets at a substantial premium. Additionally, the Government of Ecuador (GoE), with support from the Inter-American Development Bank, is promoting Ecuadorian cacao in global, high-end niche markets. Also in line with the repositioning, Ecuadorian farmers and the GoE are encouraging the cultivation and maintenance of Ecuador's heritage cacao trees, which produce that highly desired flavor profile.

Authors: Lisa Carse and Martin Webber – J.E. Austin Associates

Box 11: Thailand GAP Cluster—Positioning Products (and the Value Chain) for Greater Value and Competitiveness

Leveraging Other Value-Chain Initiatives

Box 10 illustrated the concept of the hidden potential in value chains. Key to realizing hidden potential in value chains is the pursuit of higher product scope and differentiation. Thailand's Good Agricultural Practices (GAP) cluster is an excellent example of such an attempt.

Thailand's agricultural sector is an important contributor to its economy. Thailand has one of the most developed agricultural products sectors in Southeast Asia and has been a net exporter of agricultural products for decades. It is known in international markets for the quality of its fresh and processed products, and producers from both developed and developing countries view it as a strong competitor.

The Thai GAP Cluster carried out a major program to implement EurepGAP standards in Western Thailand to gain export certification to service its traditional European supermarket business. (Discussed

in Tool 9, Box 12). The GAP Cluster wanted to further leverage its achievement of EurepGAP standards in domestic Thai markets that have similar market requirements. Thailand has many large super- and hyper-markets in its urban areas, including foreign-owned retail outlets like Tesco and Carrefour that desire similar-quality products for their customers.

Product Positioning

Although value chain processes produced a quality product that conformed to tough European standards, the Thai GAP cluster wanted to position those same

quality products within local super- and hyper-markets in Thai urban areas. This would support a market diversification strategy that reduces the risks of relying on exports linked to its investments in upgrading. The cluster took some of the following initiatives to position itself as a quality vegetable producer:

1. **Branding and certification**: The cluster created the Western Cluster GAP logo and a grading process for its products, which was approved and certified by the Department of Agriculture. The logo and standards appear on all producers' packaging, and farmers are encouraged to promote the brand. The cluster has also done some local advertising of this brand name.

2. Unique shelf space: The cluster was able to

work with supermarkets to gain dedicated shelf space for its vegetables. This separated its products from the open-air vegetables. This space was also branded with point-of-purchase (PoP) displays.

3. **Packaging**: The cluster created unique packaging that highlighted product freshness and high quality, and it was also used to promote ready-to-eat vegetables that were in demand by time-conscious urban professionals.

Summary

To leverage the quality improvements made within the value chain, the cluster positioned itself within local markets that had similar requirements to export markets by using branding, certification, unique shelf space, and packaging that promoted quality and freshness. These improvements are estimated to have generated a 50 percent increase in farmers' bottom lines.

Businesses must make these choices with purpose, or they will simply be out-positioned by their competitors or buyers, which makes understanding the competition a critical aspect of repositioning, as well. Movements to new positions do not have to happen instantly, and it is often most appropriate to adopt changes incrementally, for example, by adding differentiation through higher quality or better variety, and then adding product scope through processing or improved packaging.

Author: Mike Ducker – J.E. Austin Associates

In the case of the cashmere industry (see Figure 45), Mongolia produced the world's best-quality raw cashmere (wool from a breed of small goats) in the world. Mongolia produced very few value-added products (and virtually none of quality), so all exports were in the form of raw cashmere or yarn. Mongolia exported most of its cashmere to processors in Italy and Scotland, where the cashmere was processed and transformed into extremely high-value garments and fashions. The accompanying figure illustrates Mongolia's position in the world cashmere market relative to major competitors and partners, with differentiation represented on the x-axis and product scope represented on the y-axis. (Note that this is an application of benchmarking introduced in Tool 3). Each country's market share as measured in revenues is represented by the relative size of the circles. Note that, in most cases, higher profit per unit (and certainly higher



Figure 45: Product Positioning—Mongolian Cashmere Industry

Source: Nathan Associates Inc. and J.E. Austin Associates

prices per unit) is implied by a position in the top right corner. However, it is important for this graph to be interpreted in the context of local market conditions and the overall profit and profitability of the activity.⁹⁸ While the upper right quadrant is generally associated with high profitability, operations in the lower left may find that high volumes of low value production may generate total profits that outweigh the premium pricing available at other positions.

⁹⁸ Once completed, the graph is strategically revealing; however, compiling the information for the analysis is not straightforward, apart from the size and segmentation of the circles. Once basic sales data are compiled for the value chain's key product segments, the circles must then be positioned on the graph. If available, the most accurate data to measure product scope is industry value-add. In some cases, reliable data on value-add for a specific industry may not be available. In this case, product scope can be gauged by assessing the numbers and types of value-add activities that occur in-country. Product differentiation can be measured in a number of different ways. Market price can be a reflection of the differentiation of a product in a given market, but this is not always the case. Also, at the aggregated level, average price across a number of markets can be diluted and mask differentiation. As a result, differentiation should be measured by a combination of both the average price point at a given level of value-addition as well as specific product attributes and customer perception of the sources of product value. Product attributes and customer perception are qualitative measurements, but can still be measured through market research tools such as customer surveys.

Tool 8: Positioning Products and Value Chains for Greater Value and Competitiveness

China was also a major producer of finished cashmere products but of lower quality than those produced in Scotland and Italy. Seeking new and inexpensive sources of cashmere wool for their high-volume business, Chinese cashmere buyers competed with domestic wholesalers for Mongolian wool stocks. However, Chinese buyers had no incentive to encourage high quality as they planned to feed low value added industrial garment makers in China. Their advantage in the purchasing marketplace was to offer immediate cash. With this incentive, Mongolian herdsmen were ready to sell to Chinese buyers. The perceived need to protect breed and wool quality suffered among herdsmen selling to undiscerning Chinese markets.

Mongolia's downstream buyers recognized an opportunity to increase the Mongolian product's quality differentiation by offering price incentives for higher quality and by implementing both a mark (certification) of quality and a Mongolian brand. At the same time, design and manufacture of cashmere products in Mongolia was encouraged, which added value within the Mongolian industry.

The Mongolian cashmere industry understood the actions necessary to change its positioning in the value chain and acted to move to "the right" (greater differentiation) and "upwards" (more value added) – a more lucrative positioning with less exposure to competitive pressures.

Once an objective is determined, value chain participants must decide which actions are necessary to achieve the desired positioning.

A Step-by-Step Summary for Tool 8: Positioning Products and Value Chains for Greater Value and Competitiveness

- Assess and benchmark size and share of market. Collect net sales for each product segment and each competitor in each product segment.
- Assess and benchmark product scope (or product "offer"). Collect industry value added for each product and competitors' products. If industry value added is not available at the product level, the number and type of value-added activities taking place in-country can be substituted.
- Assess and benchmark product differentiation. Product differentiation should be measured by a combination of both the average price point at a given level of value addition, as well as specific product attributes and customer perceptions of the sources of product value. Product attributes and customer perceptions are qualitative measurements but can be assessed quantitatively through market research tools, such as consumer surveys.
- Understand the possibilities and requirements for repositioning the product, and develop a repositioning strategy. Repositioning a product in a more competitive space requires a combination of activities that both enhance product differentiation and increase in-country, value-added activities.

Case Study: Value Chain Strategies for Market Repositioning— Rwandan Coffee⁹⁹

Introduction

Now that you have been exposed to the positioning tool, what follows is an example of how the tool could have been used in the Rwandan coffee sector. Before 2001, Rwanda was an unknown in the specialty/high-value coffee sector. Today, it is a sought-after supplier of specialty coffees to Europe and the United States. There is no indication that the Government of Rwanda implemented the positioning tool in its entirety when seeking to improve the country's position in the global coffee market, but let's imagine how things could have transpired if that had been the case.

Factors to Consider

When reviewing this case it is important to keep these questions in mind:

- Where is Rwandan coffee positioned currently, and what factors influence that position?
- Where does the industry want to go, and what are the quality, service, and other requirements to make it competitive in that positioning?
- Who are the best global performers, and what puts them in this position?
- What price points and profit margins exist in the various quadrants?



Figure 46: Rwanda Coffee Product Position 1990-2000

Source: J.E. Austin Associates

⁹⁹ Authors: Carlton Jones and Martin Webber - J.E. Austin Associates

Background (1904–2001)

Coffee was introduced in Rwanda in 1904 and was first exported in 1917; it was quickly seen as a major source of income for the country. In 1933, the cultivation of coffee was made compulsory, and, in 1963, the Government of Rwanda (GoR) passed laws making it illegal to uproot coffee trees.

Because of coffee's historical role as a principal source of foreign exchange for the country (averaging 56.7 percent of all exports in the 1990s) and its broad political support, the state was heavily involved in all stages of production, including marketing and dry milling. Nearly all production of coffee ceased, however, during the genocide of 1994. Regionally, Rwanda competed with its southern neighbor Burundi, yet Burundi's coffee was considered to be slightly better in quality than that of Rwanda. Neither country produced coffee that was considered specialty grade, as did in Ethiopia and Kenya. Despite the fact that some of the coffee grown in Rwanda, such as the Bourbon varietal, had specialty-grade potential, all of Rwanda's coffee was considered below commodity-grade, and, when exported, was only used by roasters in blends of low-end, mass-produced coffee destined for Europe and the United States.

Within the value chain, the GoR supported the coffee industry by establishing OCIR-CAFÉ (Office Nationale de Cultures Industrielles-Café) and through it, distributed seedlings, fertilizer, phytosanitary products, and other inputs to growers for free, or at dramatically reduced prices. Growers would apply the inputs to their crops, harvest, and sell semi-washed beans to RWANDEX, the monopoly responsible for dry milling and exporting coffee. The GoR was a majority owner of RWANDEX, and set the prices that growers received from their coffee sales. In fact, the GoR continued to set prices for coffee up until 1998. Though coffee producer associations existed in Rwanda, they were agents of the state that distributed inputs and did little else.

Global Fluctuations Drive Down Prices and Production

Coffee's contribution to Rwanda's foreign exchange declined as the 1990s came to a close. In 1992, production peaked at 39,000 mt but fell by more than half in the years following the 1994 genocide. While production declined, the middle of the decade saw worldwide prices at near record highs. In response to this higher demand, Vietnam and Brazil took advantage of the market opportunity by planting, growing, and flooding the market with commodity-grade coffee. Beginning in 1997, this influx of supply dropped worldwide prices to historic lows. By 2001, average coffee global prices were USD 0.52/lb. Rwanda received USD 0.40/kg (USD 0.18/lb.), which, on average, was below the price of production. With prices this low, each actor in the Rwandan coffee value chain lost money: small growers, processors, exporters, and even the banks that provided lending.¹⁰⁰

Year	1986	1987	1988	1989	1990	1992	1993	1995	1996
Production (Tons)	35,424	43,026	43,026	39,091	39,575	38,970	28,495	21,829	15,239
Standard Coffee Percentage	48.18	30.8	19.4	2	7	0.32	4.25	2.4	0.25
Ordinary Coffee* Percentage	38.59	60.2	75.7	94.7	86.9	93	88.7	92.7	82.9

Table 19: Rwandan Coffee Production

Source: OCIR-CAFE

¹⁰⁰ Rob Henning, OTF Group.

* Ordinary coffee is non-export grade, while standard coffee is exportable, commodity-grade coffee. Neither is considered specialty-grade.¹⁰¹

Rwanda's In-Country Constraints

World coffee prices were not the only determining factor for Rwanda's coffee-price reduction; the country's coffee production never recovered to 1992 production volumes because its existing production process was inefficient. This resulted from Rwanda's disparate methods of coffee farming, the poor health of its coffee trees, the lack of wet-milling stations, and the absence of incentives for re-investment. Growers were not offered higher prices for better-quality beans, so they had little reason to invest in more sophisticated processes of production, harvesting, cleaning, or separation of their bean harvests. Low coffee yields coupled with poor price points influenced farmers to focus on other, higher-margin crops, further diminishing Rwandan coffee's competitiveness in world markets.

Despite the constraints that led Rwanda to produce low-quality, low-quantity, commodity-grade coffee, the GoR and donor partners believed that Rwanda possessed the capacity, environmental conditions (elevation, climate, soil quality, Bourbon trees, etc.) and political will to improve its coffee position in world markets. What Rwanda lacked was technical capacity, market information, and a coherent strategy.

Repositioning the Offer for Rwandan Coffee

In response to the steady declines in production, quality, and export revenue, the GoR and its donor partners began strategy sessions aimed at improving Rwandan coffee's positioning in world markets. These sessions resulted in coffeesector liberalization strategies which, when implemented, began the task of improving Rwandan coffee. Armed with market information, the private sector learned that higher-value coffee was very attractive to global markets, that cupping/taste results indicated Rwanda had significant potential to produce specialty coffee, and that Rwanda could compete with higher-end producers such as Guatemala, Ethiopia, and Kenya. The Rwandan repositioning strategy is illustrated in Figure 47.



Source: J.E. Austin Associates

Contributing to Rwandan Coffee Repositioning

In order for Rwanda to move from being considered a commodity-grade to a specialty-grade coffee producer, its coffee-producing sector needed to address three key areas:

¹⁰¹ OCIR-CAFE

- Increase production, since production levels were insufficient to attract global demand. Activities included distributing improved inputs, supporting growing associations, replanting coffee trees, and constructing wet-mill stations in Rwanda's top 50 coffee-producing districts.
- **Improve quality**. Activities included educating producers on quality and cupping, establishing quality-control mechanisms, investing and technical assistance in wet-mill techniques and operational and financial management, improving infrastructure, strengthening cooperative and association management, strengthening existing institutions (like OCIR-CAFÉ), and providing financial mechanisms throughout the coffee value chain.
- **Promote the Rwandan brand**. Activities included establishing and improving market linkages through trade-show visits, sharing information on the local and global coffee markets with the private sector, and instituting other innovative promotional activities.

These activities were carried out through a variety of implementing partners and sponsored by various donor partners.

Implementation and Outcomes

Through these interventions, Rwanda was effectively able to reposition its coffee and compete in highergrade/higher-priced sectors. July 2002 saw the country's first sales of commercial volumes of specialty coffee, including a sale of 33 mt to Community Coffee in the U.S. By March of the following year, privately financed and operated wet-mill facilities produced fully washed coffee. Production and quality continued to increase, and after visits to and from trade-show buyers, Rwandan specialty coffee made its first sale to Starbucks Coffee Corporation in June 2004. In November 2005, Starbucks selected two privately owned wet-milling facilities for an exclusive distribution program which provided coffee to 5,000 Starbucks retail outlets.

In 1990, Rwanda's commodity-grade coffee fetched USD 1.18/kg (0.54/lb.), but by 2001 its price had decreased to USD 0.40/kg (USD 0.18/lb.). However, through the above interventions, which were implemented by the Partnership to Enhance Agriculture in Rwanda Through Linkages (PEARL) and Agribusiness Development Activity in Rwanda (ADAR) projects funded by USAID, specialty-coffee production and its subsequent price both increased. Without the wet-milling interventions, Rwanda's ability to improve the quality of its beans from ordinary to standard and specialty would have been impossible. Also, if Rwanda had only tried to maximize profits and decrease costs within its value chain without attempting a repositioning strategy, it would have been more difficult to obtain the same results.

Figure 48: Generic Coffee Value Chain



Source: J.E. Ausitn Associates, Inc.

Rwandan Coffee Today and into the Future

Figure 49: Results of Rwandan Coffee's Positions Efforts, 2005



* Represents specialty coffee/not drawn to scale 🛛 ** Represents commodity coffee/not drawn to scale

Source: J.E. Austin Associates, Inc.

In addition to the increased production and price/kg that Rwandan specialty coffee realized, the new wetmilling stations created 4,000 new jobs, and 5,000 rural households saw their income more than double.¹⁰²

Table 20: Recent Rwandan Coffee Production and Prices

Year	Metric Tons Produced	Price/Kg (USD)
2003	162	1.56
2004	747	2.40
2005	1,190	3.10
Source: OCIR-CA	AFE	

Initial success from these interventions has sparked further donor interest in supporting Rwanda's specialty-coffee sector. Partnerships with the GoR have spawned plans to continue increasing production through the construction of an additional 164 wet-milling facilities. Continued investment in the existing wet-mill facilities is required to make better use of water sources and to equip them with water-recycling pumps. These, and other interventions, will allow Rwanda to continue to support its growing specialty-coffee industry.

¹⁰² Chemonics: "Assessing USAID's Investments in Rwanda's Coffee Sector," April 2006.

Year	1998	2000	2001	2002	2003	2004	2005
Production (Tons)	14,268	16,098	18,267	19,796	14,175	29,000	17,000
Standard Coffee Percentage	7.4	19.5	18.5	29.28	32.4	40	45
Ordinary Coffee Percentage	80.5	72.4	75	58.57	55	50	45

Table 21: Standard and	Ordinary	Coffee Percentages ,	1998-2005
------------------------	----------	-----------------------------	-----------

Source: OCIR-CAFE

Source: J.E. Austin Associates, Inc.

As Rwanda gets closer to 2010, continued product positioning will be required to help chart new strategies for the Rwandan coffee market. OCIR-CAFÉ has developed a plan to move away from standard coffee altogether and focus only on specialty varieties by 2008. However, Rwandans could also choose to diversify their product offerings by seeking broader markets for standard coffee while simultaneously maintaining focus on higher prices for specialty coffee. This would leverage the Rwandan coffee sector's increased capacity and maturing coffee acumen and could enable Rwanda to broaden its coffee production to more diverse markets.

In fact, this latter case is ongoing, and Rwanda is more than midway into its positioning strategy. However, it will most likely have future opportunities to re-visit its strategy and take advantage of future market opportunities.



Figure 50: Rwanda Coffee Positioning 2010 and Beyond

Has Rwanda been successful in repositioning its coffee industry? In many aspects, yes; by understanding its position in world coffee markets, and then implementing a strategy to repositioning itself, Rwanda's coffee industry has revitalized itself and has improved the earning potential of many small growers.

Tool 9:Applying Standards and Certifications to Achieve GreaterQuality

Meeting a variety of quality and performance standards is integral to the success of agricultural value chains. However, the importance of specific qualities, standard measurements, informational or other characteristics may not be readily apparent to all the actors in a value chain if they are not familiar with the target market. Aside from normal ethical, consumer health, and safety requirements, the market side of the value chains pays increasing attention to standards and certifications. National and regional markets are imposing increasingly strict requirements for basic market entry (e.g., HACCP, EurepGAP). Powerful buyers and retailers, especially supermarkets, impose additional requirements on their supply chains.

As a result, the value chain must meet increasingly stringent requirements relating to product health and safety, intrinsic product qualities (shape, color, taste, texture, etc.), packaging and labeling, and accompanying information.

Value chains are also beginning to understand that meeting recognized standards is not just a condition for market access but a powerful way to compete for market share and higher unit values. Value chains can obtain price premiums if they meet these standards, especially if they achieve valued product and production certifications. These certifications can go well beyond market-entry requirements and appeal to special customer segments that are willing to pay premium prices (see accompanying figure). Thus, value chains are increasingly recognizing the opportunities inherent in providing certified organic, Fair Trade, bird-friendly, and other high-standard products, as well as the value of marketing and quality-control initiatives that are promoted through seals of quality.

Because a product is affected by many actors—from farm to market—achieving quality standards and certifications is a value chain issue, and the strategies must be value chain-wide strategies.

Vocabulary

- **International trade standards**: Standards applicable to trade, imposed by trade agreement or market-governance fiat
- Value-add standards: Geared to the specific requirements of niche or segmented markets; make a product more specialized or unique
- **Quality standards**: Additional, specific buyer standards such as appearance, size, and packaging

Increasing Product Value Through Standards



Figure 51: Standards Plotted Against Product Value

Source: J.E. Austin Associates, Inc.

Standards Are Part of Value Chain Strategy

Choosing and targeting standards should be part of a strategic market exercise. Many export markets have standards and/or certifications that are needed for entry. By achieving international certification or standards, local value chains have the opportunity to export to other countries and select the market positioning of their products. However, implementing standards does not automatically mean that the value chain can sell in

Figure 52: International and Value-Added Standards



Source: J.E. Austin Associates, Inc.

those export markets; the chains must still market and sell to customers in those countries.

Value added standards allow for entry into certain niche markets; for example, Rainforest Alliance standards for coffee products, appeal to many coffeehouses and specialty marketers in the United States (see accompanying figure). Many individual importers and retailers have their own quality standards that appeal to their particular customer base.

Implementing the processes and systems to meet standards does take resources, so it is important that the value chain's leaders and firms choose the most strategically appropriate target market segments. Comparing implementation costs and the local value chain's capacity to incorporate standards against the benefits of selling up-market is a strategic choice that must be considered before incorporating standards.

Elements of Standards

Implementing standards within a value chain means discussing three elements:

- Certification bodies
- Information channels
- Management oversight with the right governance structures

Certification Bodies

Many standards and certificates have international bodies that certify that products meet certain standards. Several have the ability to certify products and services in developing countries, but cost and timing can be a major issue in bringing these international bodies into developing countries. Therefore – and especially if large numbers of producers or exporters will receive certification – it is often better to set up institutions in the local country. Although it is beyond the scope of this Guide to discuss how to set up a certification body, it is worth noting that several options are available:

- Private company
- Industry association
- NGO
- Government agencies.

Governments are typically geared to certify products and services to protect their citizens' health and welfare, but are typically not best-suited to certify based on other criteria. NGOs, associations, and private firms around the world have successfully set up certification agencies based on international standards.

Box 12: Thailand GAP Cluster—Use of Standards and Certifications to Upgrade Value

In the early 2000s, the GAP cluster in Western Thailand was able to reorient its production to meet the specific certifications needed to enter European and U.S. markets. This cluster was also able to develop and market certifications that add value to its products. These results were achieved through close collaboration among value chain participants.

Importance of Implementing EurepGAP Through the Value Chain

Thailand has long been able to successfully export high-value horticulture products to Europe. Once EurepGAP certification and traceability requirements were put in place, however, Thai vegetable exports dropped by 20 percent. The drop in exports was felt throughout the value chain—by exporters, packers, and growers. The exporters and government made several attempts to implement EurepGAP within the value chain, though without success. This was largely attributable to mistrust among the exporters, government, and growers.

A cluster approach was used to provide the training and certification services by incorporating a local university, Kaesetsart University. The exporters collaborated closely in identifying market requirements, and the university trained 2,000 growers on EurepGAP compliance, sending them to train other farmers. Growers were more willing to accept the new processes when presented by a respected third party (the university) with information support from the exporters. This allowed the exporters to gain market share and increase their exports. The growers benefited by receiving a 50 percent increase in their prices for certified products.

Use of Certification and Branding for Local Markets

To further leverage the output of growers who participated in the EurepGAP training and processes, the cluster developed a Western Thailand Good Agriculture Practice (GAP) logo. The cluster worked on sophisticated packaging, branding, and smaller packages that were geared toward the large supermarkets in Thai cities. It is expected that each party in the value chain will gain greater value by leveraging this certification and brand.

Author: Mike Ducker – J.E. Austin Associates

Thai Vegetable Export Value Chain



Figure 53: Thai Vegetable Export Value Chains

Thai Vegetable Export Value Chain



Source: J.E. Austin Associates

Information Channels

Entrepreneurs – businesses and producers, in general – must be motivated to change the way they produce goods and services to meet standards, and they need information and economic incentives in order to do this. These incentives must include shared knowledge of the requirements (and ideally, the logic behind them), as well as price points that reflect the additional costs and work involved in meeting requirements. There must also be trust within the value chain that the process is stable.

One aspect of creating trust is ensuring that proper information channels are available and being used; these will give small and medium enterprises (SMEs) and producers confidence that they will obtain fair rewards for the costs of implementing new processes to meet standards. This means that the value chain must communicate formally and informally. Formal communication can occur through meetings, cluster activities, integrated supply chain systems, industry associations, conferences, and price points. Informal channels can be created by having a transparent culture within the value chain, especially within the SMEs' supply base. For example, if a farmer sees another farmer in the village benefit from an investment to meet standards or gain a certificate, that second farmer is more likely to upgrade processes based on the observed model.

Management Oversight and Governance

A value chain that is upgrading its standards requires some institution to take responsibility for managing the supply base to meet standards and specifications. Because the value chain is comprised of independent firms and actors that often have different motivations, this can be a difficult task. Credible governance, clearly able to link value chain performance to market rewards, is always needed if the supply response of the value chain is to be coordinated and targeted to market needs.

Typically, there are three types of institutions for management of the value chain:

- Supply chain management instituted by an exporter, processor, or other lead firm
- Associations or cooperatives
- Government agencies.

The success of these various forms of oversight and governance rests with their ability to understand market requirements and translate them into in-chain procurement standards, communicating information effectively and motivating suppliers to respond with needed investments and operations.

Lead Firms

Many lead firms have made a "mindset" change that allows them to view the effective management of their supply chain as an opportunity for growth creation and larger profits, rather than simply as an additional imposed cost. This mindset shift is based on the fact that, if the participants in a value chain work together, the value chain can more easily meet changing market needs and reduce transaction costs. Therefore, the lead firm should take responsibility for

managing its supply base, which includes ensuring that that base is certified or is meeting market standards. In fragmented industries, lead firms are sometimes unable to manage the supply chain to meet market standards, and so the industry would have to leverage other institutions. The need for coordinated response within a fragmented production environment is a key motivator for integration and consolidation within value chains.

Associations and Cooperatives

In fragmented value chains, associations and cooperatives can help manage value chain elements to meet market standards. These institutions are often important when first implementing standards in a value chain; institutions that are producer-owned, visible, and well known can build trust in the standardization process. Properly resourced, they can be very effective in training and outreach activities. In some cases, involving a credible third-party institution (as in the example of Kaesetsart University and the Thai GAP cluster, Box 12) can help overcome initial mistrust between producers and processors/exporters and can help focus all parties on a coordinated approach to meeting standards.

Government Agencies

In many countries, and often in keeping with traditional, historical/colonial, or even philosophical backgrounds, government has played the role of industry governance, including standard-setting, inspection, extension, and communication. Marketing boards or authorities are one prominent set of examples. However, while some state-run agencies are effective in their governance and oversight role – and, in some cases, even step in to redress serious sectoral problems – there are also problems with such models. Government agencies are often slow to understand and respond to market trends, limited in their strategic focus, inflexible in promoting varying standards and price points, subject to public-sector inefficiencies and political and budget pressures, and liable to communicate poorly with the value chain.

Box 13: Ecuador Cacao—Improving Quality at the Producer Level to Achieve Higher Market Prices

In 2003, the Government of Ecuador (GoE) launched an alternative development project focused on the border region between Ecuador and Colombia, which is a locus for cultivation of illicit crops. An assessment of the income and employment potential of other current crops grown in the area pointed to five potential crops, including cacao.

There are two main types of cacao: forastero and criollo. Forastero cacao accounts for approximately 95 percent; most of the remaining 5 percent is criollo.¹⁰³ Criollo cacao has traditionally been produced in Latin America, while forastero cacao has been grown in Africa and Asia. Cacao is graded based on several criteria, including mold content, level of fermentation, and percentage of cocoa butter, all of which affect the flavor profile of the chocolate produced. Criollo cacao is generally more flavorful, but it does not necessarily always command a premium. Cacao buyers and processors pay more attention to the local characteristics that create the particular flavor profile of the product, rather than to the type of cacao grown.¹⁰⁴

¹⁰³ http://www.unctad.org/infocomm/anglais/cocoa/quality.htm.

¹⁰⁴ Author interview with Karl Walk, purchaser for Blommer Chocolate.

Ecuador mainly grows criollo cacao, and its flavor profile is particularly well suited to producing fine dark chocolate, the market segment for which has grown exponentially during recent years.¹⁰⁵ However, quality-control problems and poor handling practices destroyed the conductivity of fine flavor and aroma in about 50 percent of the harvest in Ecuador. This meant that even though Ecuador was growing the criollo variety, 50 percent of the crop could not be sold at a higher price or was unfit for export.

Cacao buyers typically do not buy directly from producers (see the accompanying figure), preferring to





Source: J.E. Austin Associates, Inc.

4580

4000

3500

3000

2500

2000

1500

1000

500

Proces (\$A)

work through collection centers. This is due to the fact that cacao is not a plantation crop, and there is largely an absence of associations that can sell cacao in sufficient quantity.

A value chain analysis revealed several constraints in the cacao sector, including poor production technology, mixing different types of beans without quality differentiation, and selling wet and unfermented cacao.¹⁰⁶ Fermentation changes cacao's flavor profile - sometimes positively and sometimes not. Ecuador's criollo cacao naturally possesses a desirable flavor profile for higherend dark chocolate products, so producers sought to address quality problems as well as reduce post-harvest losses.

Prices and Production of Cacao Farmer Field Schools. supported by USAID, helped educate farmers to better cultivate and protect their trees, add value to their product by differentiating between types of beans, and ferment and dry the beans. Farmers who provide value-added services receive a higher price for their beans; those who sell beans

Figure 56: World Production of Cacao Beans





Source: UNCTAD based on the data from International Cocoa Organization, quarterly bulletin of cocoa statistics

¹⁰⁵ Author interview with Karl Walk, purchaser for Blommer Chocolate.

¹⁰⁶ The process of drying cacao reduces the likelihood of developing mold; currently, the U.S. Department of Agriculture rejects shipments of cacao that exceed 4 percent mold content.

after fermenting and drying them can receive USD 60 per quintal, versus USD 28–30 per quintal if unprocessed.¹⁰⁷ Cacao beans are sold on the New York and London exchanges, where a baseline market price is offered; then, premiums or discounts are applied depending on the country of origin, the quality of the shipments, and other factors. Between 50 percent and 75 percent of Ecuadorian production is currently of sufficient quality to be sold at a higher price.

Historically, Ecuador has received a USD 20–100 premium over the baseline market price. However, the flavor profile of Ecuadorian cacao is so desirable in today's consumer market that it currently commands a premium of USD 800–1,200 per ton.¹⁰⁸ Through the Farmer Field Schools, farmers learned how to add more value to their product by maintaining bean quality, fermenting, and drying the beans before taking them to the collection centers. These actions have led to the production of higher quality cacao and, in turn, to the farmers receiving a higher price for their cacao.

Prices of Various Countries' Cacao Products

Country	Premium over Market Price,
	4 May, 2007 (per ton, USD)
Venezuela (fermented)	2,000
Java (fermented)	2,000
Ecuador (partially fermented)	800–1,200
Ivory Coast (fermented)	320
Dominican Republic (unfermented)	280
Ghana (fermented)	250
Author: Lisa Carse – J.E. Austin Associates	

In Summary

It is a strategic decision to instill standards or certifications into the value chain based on target market requirements. The choice of standards and certifications needs to be an informed one. To implement these, effective elements of certification bodies, information channels, and management oversight must be established.

¹⁰⁷ Technical Evaluation of the Ecuador Northern Border Income and Employment Project Implemented by ARD, Inc. ¹⁰⁸ Author interview with Karl Walk, purchaser for Blommer Chocolate.
Case Study: Uganda Nile Perch Quality Management and Certification¹⁰⁹

Points to Consider

This case study looks at certification processes that help ensure safety and quality for the final consumers of Nile perch. The case considers the Nile perchexport value chain in Uganda, which extends from Lake Victoria fishermen to dinner tables in Europe and around the world, and how quality and safety issues impact that value chain. The Ugandan value chain successfully improved its quality management when faced with potentially losing its sizable European market. This case also highlights the value that different certifications have within the value chain.

Nile Perch Value Chain as of 2006

Background

Opportunistic growth in the 1990s led to an undisciplined value chain which caused sector quality issues.

In the early 1990s, Kenya, and then Uganda, started exporting Nile perch to Europe. Fish became a major Ugandan export during the mid-1990s. Despite the profitability of these initial years, the Ugandan fish industry failed to invest in quality control measures central to the value chain's long term strength.

From 1997 to 2000, several health incidents in the European Union and in Uganda caused the EU to place numerous bans and restrictions on Nile perch imports (and other fish) from Uganda, as well as on other countries that export fish from Lake Victoria. (See Nile Perch Certification Timeline for details). Problems first surfaced when Spanish and Italian authorities found contamination in Nile perch and issued bilateral bans. These steps forced the European Commission to send several missions to Uganda to test the product and the processes within the value chain. Some of the problems that were identified from these missions included:

- Fish-processing plants failing to meet conditions laid out in EU regulations
- The Uganda National Bureau of Standards (UNBS) was issuing health certificates incorrectly

Nile Perch Certification Timeline

February 1997: Spain and Italy claim that their authorities have detected high levels of bacterial contamination, impose bilateral ban on fishery product imports March 1997: EU inspection confirms 'serious microbiological contamination' April 1997: EU requires mandatory tests on imports of Nile perch from the 3 East African countries December 1997–June 1998: Following an outbreak of cholera in East Africa, the EU bans fresh fish imports and imposes mandatory tests on frozen fish from East Africa November-December 1998: EC sends a Food and Veterinary Office mission to Uganda to assess compliance in production conditions; two processing plants are found not compliant December 1998: The two non-compliant plants are removed from the list of approved establishments; in the same month, the Uganda press reports instances of fish poisoning in Lake Victoria March 1999: Based on press reports, a number of District authorities ban fish sales. UNBS notifies the EU that it cannot guarantee the safety of fish exports April 1999: EU meets in Brussels with representatives of competent authorities from Uganda, Kenya and Tanzania to discuss test results; the EU announces a ban of exports of fresh and frozen fish from the three countries August 1999: EU mission to assess resources and capabilities of competent authority in Uganda in relation to control of pesticide residues; mission provides 10 recommendations; UNBS responds to EU report but does not provide all requested documentation or details. A private laboratory is established in Kampala , accredited to perform pesticide-monitoring tests; industry adopts the 'voluntary code of practice' for quality control; DFR revises inspection manual. Transfer of competent authority from UNBS to DFR July 2000: Ban lifted as the EU accepts guarantees that Uganda had put in required procedures for safety of exports; country qualifies for temporary certification in List II

(From DISS 2005)

May 2001: Uganda goes back to List I

¹⁰⁹ Authors: Michael Ducker and Martin Webber

[–] J.E. Austin Associates

- Lack of microbiological check tests to support government health certificates
- Lack of routine government monitoring for presence of chemicals in fish and water
- Lack of sanitary infrastructure and fencing at landing sites
- Fish handling was not hygienic throughout the chain

In response to the missions' findings two Ugandan processing plants were prohibited from exporting Nile perch to the EU.

In 1999, the Ugandan press reported instances of fish poisoning in Lake Victoria. The UNBS notified the EU that it could not guarantee the safety of fish exports, and the EU then banned all fish imports from Lake Victoria. The EU import bans had wide-ranging effects in Uganda; in addition to lower fish exports and export revenue, fishing communities suffered tremendous damage, as did fish processors and related service industries (packaging, transport, etc.). As a result of the bans, three plants closed down completely. The remaining plants worked at 20 percent capacity, while 60-70 percent of employees were laid off.

Intervention: The Value Chain and Public Sector Working Together to Comply and Rebuild Confidence in the European Market

The EU bans shut off the supply of a product that was much in demand in the EU marketplace, and the market had few substitutes. Pressures on both the demand and supply sides of the market therefore provided major incentives for actors throughout the value chain to face the problem. The members of the Uganda Fish Processors and Exporters Association (UFPEA), European fish

importers, the Government of Uganda (GoU), and the EU worked together and took action to deal with the issues. The GoU and UFPEA had several working-group meetings to design an action plan. Their major initiative was to restructure the government agencies responsible for managing Uganda's certification process. Since Europe was the fish's major destination, the government and industry wanted its certification process to match the EurepGAP process.

Creating a EurepGAP Certification Process in Uganda

The GoU streamlined its fish regulatory and inspection systems, with the Department of Fisheries (DFR) formally becoming the sole competent authority for fishsafety issues. DFR would be responsible for certifying fish exports as EurepGAP-compliant, and it needed to achieve HACCP accreditation in order to regulate the value chain to EurepGAP standards. Figure 57: Ugandan Nile Perch Value Chain



h Source: J.E. Austin Associates, Inc.

About EU Fish Import Lists

In 1998, the EU started segregating fish exporting countries into three lists. List I countries could export fishery products to the EU from any establishment approved bv the competent authority. List II countries were authorized to export on the basis of a specific list of approved establishments. List III countries were deemed unable to provide guarantees appropriate inspection of and monitoring. In order to export from additional these countries, documentation and checks were needed and only individual establishments approved by the EU could export.

The DFR completely revised its guidelines as well as its monitoring and inspection systems. A manual of standard operating procedures was established to guide inspections, and training programs were carried out for inspectors.

DFR's central offices are staffed with 17 inspectors who monitor the overall system and operations at processing plants. Another 20 or so inspectors operate at the 14 landing sites that are approved as sources of fish for export. These inspectors issue local fish-health inspection certificates that are required to move fish from a landing site to a processing factory. These certificates contain information on both the supplier and buyer of fish at the landing site and on the fish's origin. However, DFR did not have the capacity to handle quality control with the fishermen or at many of the landing sites away from major urban centers. There are 600 total landing sites in Uganda, but only the 14 are approved for export. As a result, different monitoring procedures were put in place at the local level, including the formation of committees at landing sites and Beach Management Units (BMUs) that started registering boats and gear. This approach follows what, in fisheries, is known as co-management – power-sharing between state and local communities and a shift of responsibilities from the formation of Lake Management Organizations.¹¹⁰ As of 2004, BMU-managed landing sites were not authorized to export to Europe, so many processors, agents, and traders transfer fish at one of the 14 approved landing sites.

The Important Role of Pesticide Laboratories in the Certification Process

The EurepGAP certification process requires biochemical tests to check for pesticide residue. No laboratory in Uganda could perform the pesticide residue test, so samples were being shipped to Belgium, where a Belgian firm, Chemipher (U) Ltd., did the tests. Chemipher recognized that there was sustained business for them in Uganda, so they opened a laboratory there. Having a local lab helped

Source: J.E. Austin Associates, Inc.



streamline the certification process and reduce costs.

ISO 9001 Certification

Many of the fish processorsexporters wanted to add another layer of quality standards to their production, mostly to demonstrate, (and to help the market perceive.) that they were serious about quality. With the assistance of a USAID-financed project¹¹¹ trained the that fish processors in ISO 9001 (a of ISO 9000) subset processes, the processors were then certified by an outside consulting firm.

¹¹⁰ Danish Institute for International Studies, 2005.

¹¹¹ Under the "SPEED" project, this and other assistance to the fish industry was led by J.E. Austin Associates, within a consortium managed by Chemonics Inc.

Industry-Level Certifications

The association of fish processors, UFPEA, has adopted a voluntary code of Good Manufacturing Practices (GMPs). The fish processors implemented HACCP and good quality management systems with support from EU funding, and 11 processors were upgraded to handle fish for export (see the accompanying figure). The voluntary GMPs were monitored by UFPEA and proved helpful in improving processors' practices.

Figure 59: Good Manufacturing Practice & HACCP

Testing & Certifying ·····• Movement of product	Good Manufay	eturing Pr	actice & HACCP
Fisherman Landing Sites	Fish Logistic F (Transport and Middlemen) /	ish Processors Exporters	European Markets
	Uganda Fish Processors & Exporters Association Good Agricultura practices	Individual Fish processors HACCP	S. AREAS

Source: J.E. Austin Associates, Inc.

Company code	HAACP compliant	When started process of HACCP compliance	Number of plants upgraded**	Lenght of process of achieving compliance (months)	Total estimated cumulative expenditure to reach HACCP compliance, capital cost (US\$ '000)	Extra recurrent costs (US\$/year)*
Α	Y	1998	2	12		39600
в	Y	2001	1	12		
С	Y	1997	1	48	1927	65800
D	Y	1997	1	12	1000	
E	Y	2000	1	24		45000
F	Y	1995	1	36		72000
G	Y	1998	2	36	1000	70000
н	Y	1997	1	12	1500	80000
I	Y	2000	1	12	200	43000
Average			11	23	1125	59343

Notes: *estimate includes: product testing, extra QM labour, materials, annual training ** Plants built after 2001 (already to HACCP specifications) not included here

Figure 60: HACCP implementation Source: Stefano Ponte "Bans, Tests, and Alchemy: Food Safety Standards and the Ugandan Fish Export Industry." Danish Institute for International Studies 2005

The Results: A More Competitive Industry than Its Neighbors – and One That Has Gained a Reputation for Good Quality

In short, compliance with EU standards (including HACCP procedures) by the Ugandan fish industry in reaction to the import bans resulted in:

- Streamlined regulation under a single strong and competent authority (DFR)
- Formulation of a new fisheries policy
- Improved monitoring and inspection systems supported by inspection manuals, standard operating procedures, and the training of inspectors
- Regional efforts for the harmonization of handling procedures in the three countries sharing Lake Victoria
- Upgrading of a (small) number of landing sites and plans for upgrading a substantial number of others
- Upgrading of processing plants' procedures and layouts
- Opening up of the U.S. market, which also requires HACCP compliance
- Installation of two local laboratories (Chemipher and UNBS) and general improvement of service provision to the industry
- Increased number of processing plants and improved export performance (see accompanying figure)



Figure 61: Uganda's Nile Perch Exports

Source: EUROSTAT, J.E. Austin Associates, Inc.

Lessons Learned

• There was and still is a strong incentive for the value chain and government to work together to ensure a quality product. The potential total losses of poor quality management were very apparent to all parties.

- Paradoxically, banning the supply of all Nile perch from the Lake Victoria region created incentives for European fish importers and the EU itself to support Uganda because there were few replacement sources.
- Recognition of the importance of quality management, and of building a secure reputation for quality, led to actions beyond the minimum needed, such as implementing ISO 9001 and Good Manufacturing Processes.
- The fish processors and exporters were natural actors to take the lead in determining and disseminating information about the standards required within the value chain. They were responsible for a large part of the value-added process and were the portion of the value chain most knowledgeable about export markets' requirements. They could serve as the "control point" for the rest of the value chain. The government had reason to listen to them because they employed a large number of people and recognizably controlled the market linkage which tens of thousands of people working within the value chain helped to supply.
- There is multi-layer monitoring at the fish processing and export stages, but monitoring is much weaker at the landing sites and with the fishermen. Currently, local governments do not have the capacity to do much monitoring, so the exporters and processors have most of the responsibility to ensure that they receive fish that meets the EU market's quality requirements.
- Fishermen and fish traders also understand and remember that markets can be "turned off" and, consequently, have incentives to supply fish that meets quality requirements.

Tool 10: Identifying Needed Support Services for the Value Chain

Profitable value chains are supported by services that allow the chain to grow, be more efficient, and enhance its competitiveness. The overall objective is to improve the depth and breadth of services currently being provided to the value chain, enabling them to be commercially sustainable, and to help those services emerge where they are not being provided. The availability of new and better services should enhance the profitability of the whole value chain. Commercially sustainable services will typically involve specialization on the part of the service deliverer; thus, there are important linkages with the deepening of the value chain, as discussed in Tool 4: Upgrading and Deepening the Value Chain.

The range of services that can add value and strength to a value chain is vast. Examples include input supply, market information and product development support, business management and consulting, transportation and logistics, quality assurance (including certifications), skills, extensions and training, veterinary services, and credit and other financial services.

Box 14: Uganda Cotton—Enterprise-Linked Extension Services Model

Uganda's cotton production dramatically decreased during the 1970s, and the sector has not yet fully recovered. Supported by USAID,¹¹² the Government of Uganda sought to increase Uganda's cotton production and revenue by encouraging small farmers to grow more and higher quality cotton. Greater production would enable ginneries to operate at closer to full capacity. By involving private business in serving the needs of the farmers at the input and financing stages, the ginneries were able to encourage farmers to increase the supply of cotton available to the ginneries. This was accomplished through a model that has been successfully replicated through eight lead ginners representing the interests of supporting ginners in eight designated cotton production zones; the model has also been extended to other sectors.

The Enterprise-Linked Extension Model for the cotton industry begins with a ginnery that is performing at under-capacity, where there is nevertheless market demand for cotton. For example, the Nykatonzi ginnery has a productive capacity of 20,000 bales per year, yet in the early 2000s, it was only processing about 10,000 bales annually.¹¹³

The farmers already had supplier relationships with the ginneries; however, the industry and the GoU recognized that these farmers needed extension services if they were going to produce cotton in sufficient quantities for the ginneries to operate efficiently.

Inputs and service needs were identified by analyzing the underperformance of the sector vis-à-vis historic levels, including value chain and GAP assessments. Farmers did not have access to needed inputs, and they lacked the technical knowledge to increase their yields. The identified needs included:

- Training and extension
- Technical advice
- Access to inputs
- Financing of inputs
- A guaranteed buyer.

¹¹² Initially the SPEED (Support for Private Enterprise Expansion and Development) Project, and subsequently APEP (Agricultural Productivity Enhancement Program). Both projects ere implemented by Chemonics, Inc.

¹¹³ Chaffee, Ralph. 2 April 2002. Enterprise Linked Extension Services in Uganda.

Since output could be boosted through the use of these inputs, the ginneries worked with stockists to provide these inputs to the farmers. The purchase agreements between the ginnery and the farmer provided a risk-mitigation mechanism for the stockists' investment. The farmers' need to purchase these inputs led to the development of financial services programs for the farmers. The ginneries worked with financial service providers to give the farmers access to the financial resources they needed. All of these steps were able to take place because the farmers had guaranteed buyers in the ginneries through purchase agreements.

For extension services, ginneries were encouraged to communicate their needs and engage directly with farmers. The ginners, with support from USAID/APEP, set up demonstration plots and trained farmers in cotton production techniques, including the use of fertilizers and pesticides; farmers were trained by lead farmers, collaborating farmers, and site coordinators. An employee of the ginnery, called a site coordinator, oversees the plots, coordinates field trips, provides information to farmers, and supervises lead farmers who act as extension agents. Lead farmers pass on cotton-production technologies to collaborating farmers for adoption through trainings, field days, and hands-on exercises. Technical advice is provided by the ginneries and USAID Agricultural Productivity Enhancement Program (APEP).

The benefits to the ginneries included increased and more reliable cotton supply, greater operational efficiency, and greater loyalty among farmers to the ginnery. The benefits to the farmer included increased production, productivity, and profits; greater knowledge of the market; sophisticated production techniques; and a guaranteed buyer for their supply. The model has successfully increased cotton production and revenues in Uganda. More ginneries are in operation, and farmers have been able to increase their yields threefold to around 600 kg/hectare. (As a best-practices benchmark, Australian yields in 2005 were 2,080 kg/hectare.)

Results

The model successfully increased cotton production and revenues in Uganda. It has now also been replicated in several sectors, including maize and sunflowers. In the maize sector, corporate linkages have been strengthened, farmers have been economically empowered, and support services have been created. Production and sales volume of maize remain high, and quality has improved. In the sunflower sector, an additional USD 6 million dollars in farm income has been generated in three years, with 35,000 farmers joining a dedicated production system.¹¹⁴

Source: Uganda Bureau of Statistics Authors: Lisa Carse and Martin Webber – J.E. Austin Associates

It is absolutely in the interest of businesses, entrepreneurs, and associations to ensure provision and access to services that would enhance their value chain's profitability. In many cases, demand for services can also offer to opportunities for small and medium enterprises to enter the value chain market as suppliers.

In the specific case of financial services, relationships between value chain actors can also enable financial flows by making the potential client more attractive to traditional financial institutions. The benefits of these buyer-supplier value chain relationships – specifically, a more secure market and improved skills – make potential borrowers (suppliers) more credit-worthy to financial institutions.

¹¹⁴ Mark Wood, USAID APEP Project, 27 March, 2007.

Services can be delivered through many appropriate mechanisms. Services along the value chain can be provided by both public and private entities; commercially sustainable approaches are certainly preferable. Additionally, services may be delivered as part of another commercial transaction, such as in the case of embedded services. For example, a processor extends credit in the form of "virtual" working capital to a small farmer when the processor provides seeds, fertilizer, or pesticides as part of an outgrower scheme.¹¹⁵ In this context, for example, formal financial institutions can enter the equation and make credit flows available, perhaps even extending the offer to other financial services (savings, transfers, and longer-term loans). This would extend the depth, breadth, and sustainability of the services that were previously provided (mainly short-term working capital).

Box 15: Sri Lankan Cinnamon

World trade in spices, their related products, and herbal healthcare products is estimated to be USD 200 billion per year with an annual growth rate of over 5 percent. In Sri Lanka, growing and processing spices provides cash income to over 400,000 smallholders and many processors. Sri Lanka commands over 80 percent of the world's true cinnamon production and exports close to 13,000 mt per year in the form of quills in different

Figure 69: Cinnamon Exports Value Chain



Source: J.E. Austin Associates, Inc.





grades, mainly in bulk form, at a value of USD 50 million.

Cinnamon zeylanicum, the source of the spice, is a small, unassuming evergreen that is native to Sri Lanka's west and southwest. Cinnamon has been popular for ages, imported to Egypt as early as 2000 BC. Demand for cinnamon helped drive the European "age of discovery" beginning with Vasco da Gama's first trip to Asia in 1497. There is "true" Ceylon cinnamon, and then there are other spices which are incorrectly referred to as cinnamon, such as Cassia (C. aromaticum), Indonesian cinnamon (C. burmannii) and others (e.g. C. loureirii). The price differential between true cinnamon and Cassia is 4:1. While European and Latin American markets distinguish the varieties, the U.S. market does not. Thus, in the U.S., low-quality cassia sells for a fraction of the price of the Sri Lankan

¹¹⁵ The term 'outgrower scheme' is often reserved for schemes where agri-business has considerable control over the smallholder production process, providing a large number of services, such as input credits, tillage, spraying and harvesting. The smallholder provides land and labour in return for this comprehensive extension/input package. The high-value horticultural export sector is currently the focus of considerable outgrower scheme development (for example, Hortico in Zimbabwe and Homegrown in Kenya).

product.

Members of the spice industry cluster formed the Sri Lankan Spice Council (TSC) was formed in 2001 to establish Sri Lanka as one of the top five branded, value-added spices and allied products marketers in the world. TCS has worked hard to differentiate its Ceylon cinnamon from lower-quality substitutes. One problem that TSC addressed was that Ceylon cinnamon and Cassia are classified under the same Harmonized System (HS) code of the World Customs Organization (WCO). In collaboration with the U.S. Department of Commerce and Sri Lanka Customs, and after deliberations with the WCO, TSC succeeded in obtaining a separate subheading for "Ceylon cinnamon" in the HS code.

TSC has carried out several actions to improve product quality and market access. TSC carried out a market analysis to locate the best sales prospects for whole spices and to identify the kind of bottling and packaging that is required for optimal value creation in the U.S. TSC has also worked with local spice smallholders and suppliers to improve post-harvest handling and management of spices in order to increase the quality of spices that are delivered to Sri Lankan exporters.

Author: Michael Gorman – J.E. Austin Associates

To implement this tool, it is useful to map the particular services that are currently being provided (their sustainability, quality, and location within the chain), as well as those services that are potentially viable that can improve the chain's performance. The accompanying Figure 70 provides an example of what the resulting map could look like.



Figure 70: Mapping Actual and Potential Business and Financial Services

Source: Albu and Griffith (from Practical Action); "Mapping the Market: A Framework for Rural Enterprise Development Policy and Practice"

The mapping is, of course, related to Tool 2, Designing Informed Strategies Across the Value Chain, as well as to Tool 3, Conducting Benchmarking and Gap Assessments of Value Chains. Indeed, sound value chain analysis involves mapping the actors and, by benchmarking against competitors, identifying needed services that could enhance the value chain. The focus here, though, is in implementing the results of analytical methods, and, given its importance and potential, this implementation is treated as a separate tool in this section.

Once the map of current and potential services is complete, interventions can be developed to introduce a service for which there is potential demand within the chain, as well as to improve the quantity, quality, and sustainability of those currently being offered.

Entrepreneurs, businesses, and practitioners will need to carry out feasibility analyses and develop business plans. Within development projects, the approach will often involve one or more pilot initiatives carried out in close collaboration with value chain participants, providing adequate support to encourage emergence of the market for service. In most cases, services that are provided on a market basis will be more sustainable, as well as those provided by specialized institutions. As many of the services are sound, replicable businesses, they can be good examples of the replicable business models discussed in Tool 5.

A Step-by-Step Summary for Tool 10: Identifying Needed Support Services for the Value Chain

• Along the value chain, map services that are currently being provided, as well as those services that are potentially viable and can lead to improvements in the chain's

performance. Benchmarking and analyzing gaps against other value chains are particularly useful tools for identifying potential services not being provided.

- Include in the map indications of the services' sustainability, quality, and location within the chain.
- Conduct feasibility studies and develop business plans to introduce services for which there is potential demand within the chain and to improve the quantity, quality, and sustainability of those currently being offered. In most cases, services that are provided on a market basis will be more sustainable, as well as those provided by specialized institutions.

Case Study: Identifying Needed Support Services for the Value Chain—Zambia Cotton¹¹⁶

Introduction

Commercially viable extension services can be vital to a value chain's health. In the case of Zambia's cotton industry, support services were provided in a variable market that featured too many independent cotton traders and service providers, ultimately resulting in significant loan defaults, decreases in production, and the failure of many service providers. This case examines how a large private firm identified and altered the way it provided extension services to small growers, helping to stabilize and reinvigorate Zambia's cotton value chain.

Points to Consider

When reviewing this case, consider the following questions:

- What led to the need to provide support services?
- When is it appropriate to consider implementing different models?
- What role does the private sector play in implementation?

Background

From the late 1970s until 1994, Zambia's cotton purchasing, processing, and marketing was controlled by the state-owned Lint Company of Zambia (LINTCO). During that period, LINTCO purchased seed cotton from an estimated 140,000 small farmers at a fixed price and extended services such as the provision of certified seeds, pesticides, sprayers, bags, and advice on growing techniques. LINTCO was the principle buyer of seed cotton, the sole provider of extension services, and the sole distributor of inputs on credit. Production fluctuated during this period but generally trended downward. By the early 1990s, LINTCO was operating at a loss, having accumulated substantial unpaid debts.

Figure 71: Zambian Cotton Exports, 1990-1994



¹¹⁶ Authors: Carlton Jones and Martin Webber – J.E. Austin Associates

These trends began to change in 1994. Zambia's president, Frederick Chiluba, implemented a wide-ranging restructuring of Zambia's economy, including privatizing the cotton industry through the Zambia Agency Privatization (ZPA). ZPA facilitated the sale of LINTCO to Lonrho Cotton and Clark Cotton, two private ginners/ exporters. The two firms operated in different regions of the country, and both worked through direct outgrower schemes that offered inputs and extension services on credit to farmers. When farmers sold their cotton to ginners, the cost of the inputs and services they received before harvest was deducted from their selling price. Because both ginners operated in otherwise underserved markets, their practice of providing seed, other inputs, and services credit involved minimal risk. on Repayment rates averaged 86 percent.



Figure 72: Zambia's Cotton Value Chain



The expansion of cotton production attracted four new ginning and assembly entrants to the market.¹¹⁷ National cotton production increased through 1997.

To serve the increased demand for cotton, some ginners contracted independent outgrower agents to recruit more farmers to produce and source cotton to them. These outgrower agents received inputs from the ginners, distributed them to their farm networks on credit, and when harvest time came, sold the



Figure 73: Zambian Competition for Cottonseed

Source: J.E. Austin Associates, Inc.

cotton gathered from the farmers back to their respective ginners. In the same way that the increased cotton production attracted new ginners, the apparent success of the outgrower agents attracted

¹¹⁷ Amaka Holdings opened in 1997, but is now closed, having gone into receivership. Continental Ginnery opened in 1997, Mulungushi Textiles (now Zambia China Mulungushi) Textiles opened in 1999. Northern Growers began servicing in 1986, but expanded in 1997.

additional independent agents to the market. These agents purchased inputs from other sources and offered them to any farmer on credit. In return, that farmer would sell cotton back to the independent agent, who would sell to any ginner willing to pay.

This unchecked proliferation of ginners and independent agents led to many problems. By 1998, ginning capacity exceeded production by 50,000 tons per year.¹¹⁸ With too many agents competing for a limited number of farmers sourcing was chaotic. This competition led to growing distrust and a lack of transparency in price-setting, with agents and ginners vying to outbid each other for cotton. Some farmers, despite having agreed to outgrower contracts with specific ginners, sold their cotton to any agent willing to pay more than the contract price. Some agents purposefully sought to outbid outgrower contracts to acquire cotton. The entire value chain became volatile as ginners and outgrowers experienced increased incidences of defaulted loans. Some agents compensated for large portfolios of defaulted loans by marking up the inputs they sold to the remaining farmers who had remained loyal. This made it even more difficult for those remaining farmers to make a profit and resulted in even higher rates of loan default.

By 1999, the entire Zambian cotton value chain was in a crisis. At this time, Lonrho, while negotiating its own sale to Dunavant, began laying the groundwork for a new service and extension model. At the time of its sale, Lonrho was projected to post a USD 2 million per year shortfall. To lower the likelihood of their own default, the remaining ginners stopped making embedded service contracts with growers. In 2000, with no other means to finance production, farmer output decreased to 50,000 mt, with just 2,500 mt exported (both less than half of 1998 levels). Many of the outgrower agents that contributed to the repayment crisis closed down, and the entire value chain had an average credit default rate of 53 percent. Amaka Holdings, a ginning company, went out of business, leaving the remaining ginners to retool in an effort to survive the market crash.

Dunavant's Distributor Model

After taking over Lonrho's operations, Dunavant, a privately held U.S.-based cotton company, further implemented and perfected Lonrho's service and extension The new approach model. impacted Zambia's cotton industry in two ways. First, the service-extension model showed that outgrower schemes could work with little risk of loan defaults if the schemes were properly designed and managed. Second, Dunavant used its distributor model to significantly expand its production network.

Dunavant's distributor model was very different from previous schemes. In the past, Lonrho relied on a large number of direct company employees, including almost 800 extension agents, to carry out the required







activities. Overhead in the previous model was significant, and this burden was greatly exacerbated when borrowers defaulted. In the new distributor model, Dunavant used almost no direct-hire employees to deliver services. Instead, "distributors" were mobilized, via formal written contracts, to identify farmers. The distributors would acquire the inputs from Dunavant on credit, deliver them along with technical advice to the farmers, and ensure that farmers sold their cotton back to Dunavant in order to recover the input credit. In this scheme, the distributor's compensation was directly tied to her/his amount of credit

¹¹⁸ Tschirley, David, "Cotton in Zambia: An Assessment of its Organization, Performance, Current Policy Initiatives, and Challenges for the Future," 2004.

recovered. Paid on a graduating scale, the more credit a distributor recovered, the more the distributor earned.

Table 22: Dunavant Compensation Plan

Source:Dunavant

farmers'

realized

Dunavant

commitment to the success of the

program by offering two types of training: credit management and

practices. These courses laid a

strong foundation of sound business

operations and management for the

distributors, allowing them to assess creditworthiness

As Dunavant's distributor network

arew, its outgrower network followed

suit with the average distributor handling 65 growers each. In step

with this growth was Dunavant's

repayments. In the year prior to the

in

increase

rapidly growing their portfolios.

Recovery Percentage	Compensation Rate
65–84-	5 percent of the total credit
85–99-	Add'l 7.5 percent of the total credit
100	Add'l 12.5 percent of the total credit

showed

its

while

The accompanying table illustrates the distributor compensation plan. Distributors could earn up to 21.5 percent (0.65 * 0.05 + $0.85^{\circ} 0.075 + 1.0^{\circ} 0.125 = 0.0215)^{119}$ in commissions based on their efforts. To maintain performance, Dunavant dropped any distributor who could not bring in a minimum of 50 percent of their credit portfolio. By 2003, the company raised this minimum credit recovery rate to 60 percent. In some of the better-performing regions, the cut-off rate went to levels as high as 80 percent.

Another unique aspect of Dunavant's distributor model was the "work-in, live-in" principle. Each distributor was required to farm cotton themselves and live in the same area as their farmer network.

Figure 75: Zambian Cotton Exports Turn Around



scheme's implementation, Dunavant's reported recovery rate was 67 percent. After one year in the system, the rate grew to 80 percent - in the second year, to 88 percent and, after three years, to 93 percent.

Conclusions: Growth of a Sector

Dunavant's successful rollout of its distributor model of providing services to farmers encouraged other ginners to follow with similar arrangements. Not all ginners replicated the model exactly, and most didn't place as rigorous an emphasis on training and distributor selection as Dunavant. Still others, such as Zambia's other large ginner, Clark Cotton (Clark's southern African holdings were purchased by Cargill in 1996,) maintained the old model and continued to extend services to farmers via its own employees.

¹¹⁹ Tschirley, David, "Cotton in Zambia: An Assessment of its Organization, Performance, Current Policy Initiatives, and Challenges for the Future," 2004

In all, the entire Zambia cotton value chain has experienced rapid growth since the sharp drop of the early 2000s. This rapid recovery was led entirely by the private sector, in large part because of Dunavant's ability to identify, design, and deliver innovative ways to extend services to small farmers. As of the 2005/2006 season, Dunavant was outperforming other Zambian ginners in terms of volume and produced an estimated 115,000 mt of cotton lint. Clark/Cargill is second in the Zambia market with 60,000 mt.

Tool 11: Improving the Operating Environment by Promoting Public/Private Dialogue

Enabling Environment

The enabling environment for business is critical to the development and growth of a vibrant private sector and its firms. Constraints in the enabling environment can limit business growth, such as through:

- Increasing costs
- Decreasing competitiveness
- Decreasing product and service quality
- Increasing business and investment risk
- Decreasing foreign and domestic investment
- Constraining business' willingness to pursue long-term strategies.

The value chain's ability and willingness to upgrade and make strategic choices often depends on the development of a long-term strategy. However, a burdensome regulatory, legal, and policy environment can severely hinder the growth of industries, and diminish competitiveness and success in global markets. A poor business environment not only limits local firms' capital investments, it also obstructs foreign investment. Reforming policies so that they are favorable to businesses can encourage dynamism and efficiency.

Value chains' integrated nature makes them susceptible to constraints in the enabling environment at any level. It affects their costs and ability to act, from the most basic supplier of raw materials to the broker who arranges for a large international shipment of finished goods.

Government is often the principal actor in changing regulatory issues and is responsible for improvements and reforms. Often the private sector's interactions with government (and those of development partners as well) tend to be on the national level. However, to a value chain, the capacities and wisdom of local government can be even more important due to their impact on strategic portions of the supply or value chain. Lacking incentives and experience, local governments tend to be less sophisticated in their understanding of the constraints faced by value chain members in the enabling environment. Businesses within the value chain often have more personal relationships with local officials, which can be both a benefit and a barrier to enacting pro-growth change. Often, and especially in lower-income countries, the private sector feels powerless to effect change or believes that promoting change is not their responsibility. However, the private sector must often serve as the impetus for reform and sometimes even lead the search for solutions by providing government officials with irrefutable data and a plan for change through an open dialogue process.

Public/Private Dialogue

Through both official and ad hoc forums and vehicles for dialogue between the public and private sectors, the value chain can communicate its opportunities and challenges within the enabling environment. Through the value chain's sharing of industry strategies developed

through analytical tools such as positioning and benchmarking, the public sector can understand the impacts of its own actions, decisions, and capacities and more effectively prioritize issues and actions to align with private-sector needs.

Public/private dialogue (PPD) implementation can take several forms but generally follows the format described below. This applies to national, sub-national, or sector-specific dialogues.¹²⁰

- *Establish a mandate*. This can include memorandums of understanding, formal government mandates, or temporary initiatives.
- *Align with an institution*. Possible choices would be an investment promotion agency, government ministries, donor agency, or a newly created and independent institution. Institutional alignment also includes the location of meetings and where the initiative's secretariat will be housed.
- *Clearly defined structure and rules for participation*. The PPD process should have a secretariat and working groups for specific issues. The secretariat will serve as the coordinating body, and working-group members will come from private sector, governments, associations, academia, and possibly donors. Participants should be decision-makers and influential members of the private sector who want to facilitate change. As broad and representative a group as possible should be chosen and should include small businesses, minorities, and women to the extent possible.
- *Identify and support a champion and a facilitator*. A strong but not overly powerful champion should be chosen from the constituency to represent and drive the process. Also, dialogue facilitators should serve as brokers and help participants negotiate and set timetables and priorities.
- Set clear goals and reach them. Determine the types of outputs the PPD process will generate. These can take the form of roadmaps, a policy whitepaper, or reform proposals, in addition to softer outputs such as cooperation, which emerge through dialogue. Quantifiable results early on can help build momentum and credibility for the process.
- *Public outreach and communication*. This step is vital for disseminating broader-scale changes in the enabling environment and generating larger-scale buy-in for reforms. It can include branding of the PPD initiative, media education and awareness campaigns, and social marketing.
- *Monitor and evaluate outcomes.* Documentation of clear successes and obstacles is important to building trust and generating traction for change.

It is important to have channels of communication open with national and local municipal governments, as they each have a different influence on limiting or facilitating investment and growth. For example, businesses cannot take advantage of an excellent customs system with low tariffs if the roads leading to the port are impassable. The value chain must know which levels of government determine the policies that most affect their operating environments and work to engage them. The value chain may also be able to form common cause with local public sector leader to advocate at the national level. In order to effectively communicate as a unit, value chain participants (or industries they represent) must first develop a clear consensus strategy that outlines and prioritizes the policy changes needed.

¹²⁰ Source: PPD Handbook World Bank.

Tool 11: Improving the Operating Environment by Promoting Public/Private Dialogue

Public/private dialogue can be a tool for encouraging policy changes that enhance the competitive position of the firms within the value chain and link the value chain's strategy to the dialogue process. PPD's success rests on several factors that affect the quality of the dialogue and its strength as a forum for advocacy. First, the private sector must clearly understand the public sector's role in the value chain. The public sector may often be blamed for, or looked to for solution to, problems that are or should be more in private producers' control. It is also helpful to understand that the policy and regulatory environment have both legal and enforcement components and that these may differ substantially. The infrastructure environment and political power dynamics between various stakeholders will make up the climate of dialogue.

Last	Range of issues to identify and tackle while designing or maintaining partnerships					
1- Ignition	Government willingness	Cross- spectrum support	Business priorities	Linkages with existing organizations	Sense of urgency	Establishing credibility
2- Participation	Selection mechanisms	Terms of membership	Choosing key individuals	Striking a balance in representation	Including SMEs	Civil society participation
3- Structure	Permanent secretariat	Individual leadership	Working groups	Government structure	Transparency and rules of engagement	Institutional flexibility
4- Goals & outputs	Mission statements	Managing expectations	Quantifiable outcomes	Reform type and importance	Monitoring and accountability	Clarity and credibility
5- Role of donors	Type and level of support	Public image	Quality control	Avoiding favoritism	Sponsorship v. direction	Ownership transfer
6- Outreach	Branding and marketing	Using the media	Engaging the grassroots	Enlisting the public	Targeting decision- makers	Sharing experience

Table 23: Sample Checklist of Issues to Address at Various Stages of the PPD Process

Source: www.publicprivatedialogue.org

A benchmarking activity of specific business environment elements can be a starting point for generating a forum for constructive dialogue among stakeholders along the value chain, as it will generate credible information for decision-making. Sound prioritization and decision-making are difficult if parties are seen as advocating biased or unfounded positions or asking for special favors.

Once parties have agreed to discuss their common interests, a public/private dialogue must establish certain parameters for effective discussion. Some general, overarching principles must be in place for the forum to be effective in reducing barriers to value chain efficiency. A public/private dialogue must include the "right" people from both the public and private sectors. This means that key decision-makers and knowledgeable, credible opinion leaders must be present, rather than just the highest-ranking member of an organization. Participants must also be encouraged to commit to the process and leave egos and political infighting at the door.

They should all be operating from the same base of information and have access to the same data. This may be the benchmarking data gathered through a previous exercise. The use of benchmarking data can be especially helpful it is obtained through public-private collaboration.

The dialogue should have clear goals, collaboratively agreed upon. It is important that all of the stakeholders involved agree on the discussion's topics and goals to avoid sessions that get off-track or that degenerate into blame and fault-finding. Every PPD mechanism should have a method for providing feedback and input into policy formation.

The private sector must make a credible case for change through the strength of its strategy and by clearly communicating its goals. By using sound data, credible and objective outside experts to provide analysis, and well-facilitated discussion, the private sector can create a non-threatening and constructive dialogue environment.

It is helpful for the private sector to explain the tax and revenue impacts for the public sector if changes are put in place. The private sector should be able to link changes to meeting government goals, job creation, and revenue increases.

Public/private dialogue can encourage action that benefits the value chain by educating and informing the public sector. The private sector should be able to communicate those points at which the greatest value is added (or subtracted) from the value chain and why in order to make it easy for the public sector to decide on actions that will contribute to the value chain's strength.

Box 16: Tanzanian Coffee and KILICAFE: Productive Public/Private Dialogue

Tanzania has the climate and altitude to produce high-quality coffee, but several constraints prohibited it from competing seriously in the global specialty coffee market—the only growing segment of the market—and the one offering attractive prices. These included an aging population of coffee trees, a scarcity of well-operated wet-milling stations, and a restrictive enabling environment that forced all coffee to be sold through a blind national auction system. These constraints limited the industry, resulted in low prices to growers, providing them little incentive to produce high-quality coffee or reinvest in their plantations. The coffee sector actors wanted change but were unaware of their options. Likewise, the Government of Tanzania (GoT) knew that something needed to be done to improve the enabling environment but didn't know where to begin.

In 2001, TechnoServe helped coffee growers form the Association of Kilimanjaro Specialty Coffee Growers, (now called KILICAFE). The organization is a farmer-owned association that provides services to its members. These services include coffee marketing, provision of inputs, quality coffee production, lobbying for regulatory change, and cupping services for quality identification.

Recognizing that changes in the enabling environment would be important to the sector, TechnoServe brought the coffee sector's various stakeholders together for round-table discussions. These discussions included the Tanzanian Coffee Board, KILICAFE members and private buyers. Participants noted that they were the first discussions that actually involved all actors in the coffee sector. As a result, in partnership with stakeholders, TechnoServe undertook the Coffee Taxation & Benchmarking Initiative (CTBI), to investigate Tanzania's taxation policies and regulations governing the sector.

The study involved benchmarking comparisons with five 'peer' coffee-producing nations: Costa Rica, Guatemala, Ethiopia, Kenya, and Uganda. In addition, TechnoServe facilitated a tour for Tanzania's Minister of Agriculture to Costa Rica to meet his counterpart, as well as other coffee stakeholders in the country. Findings from the study showed that, compared to the other countries, Tanzania had the highest taxes on coffee producers (up to 21 percent) and the lowest rate of reinvestment. In some instances, countries levied significantly lower taxes, and these lower taxes helped alleviate the effect of major price drops and improve competition.

TechnoServe and the other stakeholders discussed the CBTI study and reached out to the Government of Tanzania, making the following recommendations:

- Harmonize taxes
- Reduce steps and fees for licensing
- Allow smallholder growers of specialty coffee to bypass the national auction system (Moshi Coffee Auction) and negotiate/sell directly to specialty buyers (direct export)

In June 2003, Tanzanian Minister of Finance Basil Mramba announced sweeping reforms that included the CTBI study's recommendations. The tax and marketing law reforms took effect in October 2003. In March 2004, KILICAFE, able to bypass the Moshi Coffee Auction and negotiate directly with specialty buyers, sold over 23,000 pounds of fully washed Arabica coffee directly to Peet's Coffee and Tea, a U.S.-based specialty coffee roaster. This sale involved five small grower groups, which represented 645 small growers. The small growers received a 150 percent price premium on the coffee sold to Peet's compared to other growers. This was the first direct grower-to-roaster transaction in Tanzania's history. Later that year, Peet's and KILICAFE created the "Tanzania Kilimanjaro Limited Edition" coffee brand, which was sold exclusively by Peet's coffee stores and distributors.

Today, taxes in the Tanzanian coffee sector have decreased from 21 percent to a range of 14–16 percent. The steps and fees associated with licensing are also fewer, and lower, respectively. However, the reform that has helped spark revitalization in the industry was the ability to directly export specialty coffee. Since the Peet's sale in 2005, KILICAFE has continued to negotiate and sell high-grade coffee to Peet's, Starbucks, Illycafe and a range of other global specialty buyers.

Author: Carlton Jones– J.E. Austin Associates

Associations and institutions for collaboration, as Michael Porter has called them, are either formal or informal organizations that facilitate the exchange of information among members and can help foster cooperation. They can take the form of regional, social, or professional networks or they can be industry or professional associations that cut across various industry sectors and value chains. These institutions play a significant role in strengthening and promoting the value chain or industry agenda. There are types of institutions called business membership organizations (BMOs) that serve a variety of specialized functions (see accompanying figure).

BMOs can improve the business environment by building and supporting trust among its members, establishing a forum for dialogue, collective strategy, and planning, increasing the ability to communicate with one voice through advocacy activities, and assisting with other actions (see Tables 24 and 25).

The International Finance Corporation/World Bank has an excellent resource detailing bestpractice implementations of PPD initiatives. These cases, along with guides and comprehensive information on PPD, can be found at the online forum www.publicprivatedialogue.org.

ВМО Туре	Defining Factor	Typical Functions and Services
Business Associations		
Trade/industry Associations	Occupation/Industry	Arbitration, quota allocation, industry standards setting, lobbying, quality upgrading
SME associations	Size of firm	Entrepreneurship training and consulting, finance schemes, group services
Women's associations	Gender	Entrepreneurship training, microfinance, gender-specific advocacy
Employers' associations	Labor relations	Interest representation vis-à- vis unions, professional information, and training
Confederations	Apex bodies	High-level advocacy, general business information, research, coordination of member associations
Bi-national associations	Transnationality	Trade promotion, trade fairs, matchmaking
Chambers	Geographic region	Delegated government functions, arbitration courts, basic information services, matchmaking, local economic development

Table 24: Types of Business Membership Organizations and Functions

Source: World Bank

Table 25: Pursuing Effective Dialogue

Ineffective	Effective
Individual Company	Industry Clusters
Ad-hoc Complaints	Comprehensive Vision
Operational Level	• Strategy
Laundry Lists	Priorities
Anecdotal Evidence	Data and Analysis
Concessions	Co-Responsibility
Opposite Sides	Same Side of Table

Source: J.E. Austin Associates, Inc.

Case Study: Improving the Operating Environment Through Public/Private Dialogue—Botswana Cattle Producers Association¹²¹

Introduction

Understanding the role the private sector plays in facilitating change at the local, regional, and national government levels is important when considering changes to the enabling environment for value chains. If the private sector does not speak with an informed, unified voice, doesn't have a common agenda, and doesn't understand how to engage the government, change is unlikely.

This case of the Botswana Cattle Producers Association is ongoing and is still realizing results. It demonstrates how change is occurring after private-sector participants in Botswana's red-meat value chain organized to effectively engage the government for mutually beneficial changes in the red-meat sector.

Points to Consider

Look for some of the steps outlined in the tool:

- Was a mandate established?
- What institutions did the private sector align with as it organized?
- Who was identified as a champion/facilitator for the group?
- What were some of the goals, were they clear, and how were they achieved?
- What vehicles were used to engage the government?
- Are there any measurable outcomes?

Background

In Botswana, the red-meat industry is one of three main economic pillars (behind only diamond mining and tourism) that support the economy. But with declining exports and losses experienced by cattle farmers and the state-run Botswana Meat Commission (BMC), something had to be done to turn the situation around. From contributing around 30 percent of GDP at independence in 1966, the sector shrank to 3 percent or less of GDP in 2004. Likewise, over the same period, the contribution of beef to national export revenues fell from 70 percent to 1.7 percent.¹²² In 1998, beef export value was USD 92,570,000. By 2004 it was USD 46.380.000.¹²³ Over the same period, export quantity fell from 25,000 tons to 8,600 tons.



Figure 76: Botswana Red Meat Value Chain

Established in 1966, the BMC is the country's sole exporter and domestic *Source: J.E. Austin Associates, Inc.*

wholesaler of beef. This status, in addition to other privileges, are conferred by the BMC Act. Botswana's main beef export market is the European

¹²² Jefferis, Keith "How Trade Liberalization Can Contribute to Solving the Crisis in the Cattle and Beef Sector," 2005.
¹²³ FAOSTAT

¹²¹ Author: Carlton Jones – J.E. Austin Associates

Union. Protected as a monopoly, the BMC acts as the country's single industrial processor and export channel and sets the prices it pays to cattle producers. The BMC operates abattoirs in Lobatse, Francistown, and Maun that were designed to handle an output of 8,000 cattle and 500 small stock per day (Lobatse), 400 cattle and 140 small stock per day (Francistown), and 100 cattle per day (Maun), but each operated at levels well below their capacity. The Maun abattoir was permanently closed in 1996 after an outbreak of cattle lung disease necessitated the culling of all cattle in Ngamiland.¹²⁴ Despite the closing of one of the three abattoirs, the remaining two abattoirs were still underutilized. This underutilization drove up unit costs, and the BMC began losing money. Around the same time, EU beef prices leveled, followed by the revenues from cattle sales in Botswana. Cattle producers were severely impacted, as well, and were being paid well below export parity (the price neighboring cattle producers received in Namibia and South Africa).

Botswana cattle farmers often complained among themselves and recognized that it was becoming increasingly difficult to operate, but never organized to generate change in their institutional and enabling environment. However, when losses continued and export revenue continued to decline, cattle producers could use neither their land nor their cattle as collateral against loans. Led by Philip Fischer (a producer himself), cattle producers approached a USAID-funded project, the Southern Africa Global Competitiveness Hub (SAGCH), for assistance in forming a national association. Its goal was to have

existing regional cattle-producing associations speak with one voice to create a common agenda and to lobby that agenda for policy change.

Birth of the Botswana Cattle Producers Association

SAGCH was able to provide the technical assistance that led to the formation of the Botswana Cattle Producers Association (BCPA), an organization that represents the country's 60,000 cattle producers. The BCPA organized, drafted by-laws, registered as an official industry association, elected its first chairperson (Mr. Fischer), and established a common agenda from which to lobby the government for action.

Initial components of the common agenda included the desire to receive export price parity for their livestock, an improved production system, and the liberalization of Botswana's entire red-meat sector.





Uncovering the Problems



Energized by their ability to organize and hopeful for the future, the BCPA again approached the SAGCH for assistance in conducting a study to uncover all the constraints in the red-meat market in order to provide much-needed statistical data to support their planned advocacy efforts. The study was conducted by the former Deputy Governor for the Bank of Botswana, Keith Jefferis. Mr. Jefferis is currently an independent consultant and is highly respected within the country. His study produced several key findings.

Declining National Herd

Though the data was poor, Mr. Jefferis' study suggested that the national herd population in Botswana was decreasing from its peak of approximately 3,000,000 animals in the early 1980s to about 2,500,000. Rising costs, drought, and cattle-producer cash flow problems were all cited as contributing to the herd population decline.

¹²⁴ Botswana Meat Commission

Figure 78: Exports of Botswanan Beef, 1990-2004



Additionally, he noted that the existing production system, which produced mature cows and oxen, would not provide enough 'offtake' to keep up with rising demand. The combination of a declining herd size, stagnating offtake, and rising domestic demand for beef were all contributing to reduced cattle sales to the BMC, and hence reduced throughput and capacity utilization.¹²⁵

High Costs Throughout the Value Chain

With high fixed-unit input costs, and abattoirs operating at around 50 percent capacity, the BMC operated at levels severely below built capacity. This inefficiency caused the cost per unit of output to rise. Compared with neighboring countries, Botswana was operating at less than half the volume

levels that were expected for a competitive, commercially run operation to survive.¹²⁶ Jefferis concluded that this excess capacity was, in part, the result of the 1990 opening of the Francistown abattoir.

Declining Revenues

As the BCPA suspected, the study also revealed that the declining prices paid by EU markets resulted in revenue declines and subsequent lower prices paid to cattle producers (in real terms, after adjusting for inflation). Coupled with the Government of Botswana's (GoB's) desire to make the consumption of beef affordable for all Botswanans, prices remained artificially low. Export cattle producers, without competitive alternative markets for their beef, were limited to selling to only one buyer, the BMC monopoly. This limited their ability to increase profits.

The sector conducted a benchmarking exercise that compared Botswana's cattle market to those in Namibia and South Africa. Comparing these markets, the study also determined that Namibian abattoirs paid 40 percent more than the BMC paid its cattle producers, while South Africa paid twice as much to its producers.

Operating Under Monopolistic Conditions

The study also highlighted a glaring problem: The BMC was operating as a money-losing monopoly whose inefficient operation was leading to the unsustainability of the entire red-meat market in Botswana. BCPA producers of export-quality beef were forced to sell at below-market prices to the BMC for exporting. Live cattle exports were banned, as were beef imports. This artificial market structure meant that, without interventions, the national herd would continue to decline, and cattle producers would continue to lose money and may even be forced to close their operations or move to other countries in order to survive.

BCPA Publishes Findings and Engages GoB

Shortly after the research was concluded, the above findings were published in a policy paper, "How Trade Liberalization Can Contribute to Solving the Crisis in the Cattle and Beef Sector." This paper, also

¹²⁵ Jefferis Interview 2007.

¹²⁶ Jefferis, 2005

prepared by Mr. Jefferis, highlighted some of the industry's challenges and made policy recommendations meant for consideration by the GoB and BMC. In May 2005, the BCPA assembled industry stakeholders with the assistance of the SAGCH, and formally presented its findings, and Mr. Jefferis' recommendations. These included:

Changing the Cattle Production System

Shifting modern production to techniques in Botswana would increase offtake rates and keep up with anticipated demand. As mentioned earlier as a finding from the research, Botswana's production system (the Oxen System) was based on selling mature male and female cattle to abattoirs. The modern 'weaner' system is the recognized preferred method of cattle production in the world. In this method, male calves are sold to feedlots as soon as they are weaned. While in feedlots, these weaned male calves are intensively fed and fattened for three to four months, then sold directly to abattoirs. At the farm, remaining herds are then comprised predominately of cows and calves.



Figure 79: Red Meat Value Chain Recommendations

Source: J.E. Austin Associates, Inc.

In production, the benefits of the weaner

system are that farms have higher proportions of (productive) cows, rather than (unproductive) oxen. Farms adopting weaner-system production methods are able to provide younger, better-quality meat because weaners (aged less than a year) are sent to abattoirs instead of mature cattle (whose average age is three years). Additionally, environmental impacts are reduced since animals graze for shorter periods and are younger, reducing the amount of land needed for them to graze.

Weaner-produced cattle are also less impacted by drought, as they make fewer demands on the land for grazing.

This change from an oxen system to a weaner system is a more efficient means of production and can potentially increase offtake from the existing 270,000 to 700,000 per year. More importantly, this production increase does not rely on increasing the national herd size.

Unfortunately, shifting to a weaner system isn't possible until prices for cattle are more aligned with costs. That means that BCPA members need to be paid export-parity prices, which does not occur under the existing legislation.

Revising the BMC Act

In Botswana, the BMC Act of 1966 reserves the exportation of live animals or their edible products for the abattoir unless the Minister of Agriculture permits otherwise in writing. In addition, such a permit should be issued with the BMC's consent according to Section 21, which addresses controls on the export of cattle and the licensing of export slaughterhouses.¹²⁷

¹²⁷ Botswana Meat Commission.

Case Study: Improving the Operating Environment Through Public/Private Dialogue

Jefferis' policy recommendation was to revise the BMC Act to remove the prohibition on non-BMC export slaughterhouses and beef exports, and to lift the ban on live-cattle beef imports. Jefferis also included the introduction of a national auction system as the primary method of cattle sales and price determination. accompanied by widely available beef-pricing publications. This would allow the entire beef and cattle industry to be regulated by competition and free-market principles, instead of the existing structure where prices were set solely by the BMC. With BCPA members able to get export-parity prices, they would then be able to afford the switch to weaner production systems.

Based on estimates, if the BMC Act were revised to allow for competition, prices would approximate those found in South Africa, allowing the entire Botswana cattle market to recover.

Privatizing the BMC

Operating as a monopoly, the BMC hindered growth in the sector. Selling off non-core assets, closing unproductive abattoirs, and restructuring BMC functions toward governance and health enforcement and compliance would allow the industry to move away from its existing protectionist position to being competitive.

To disseminate its research and findings, the BCPA invited key stakeholders and the national press to hear about the challenges faced by Botswana cattle producers. The event sparked a national debate about how to address the challenges of BCPA members. Just as important, the BCPA was able to produce credible data from which to make its case and, for the first time, BCPA members Source: J.E. Austin Associates, Inc. felt empowered to continue to press for change in their sector.



Figure 80: Initial Implementation of Recommendations – Red Meat

Outcomes

As with several of the cases illustrated in this Guide, this case is still evolving. Changes have occurred, but more liberalization is necessary to enable BCPA members to export to preferred markets. One immediate success for the BCPA was its ability to garner support, which influenced the GoB to strongly consider its recommendations. Their organization is now a recognized partner of the GoB and is actively engaged in policy discussions that affect the red-meat industry. The policy paper which sparked the public/private dialogue that revitalized the industry was later translated into Setswana for the House of Chiefs and other Botswanan stakeholders. The report's data was seen as one of the single most important outputs that allowed the BCPA to influence the GoB.

Figure 81: Red Meat Value Chain Next Steps



Source: J.E. Austin Associates, Inc.

In December 2005, the BMC increased the prices that it pays to producers by an average of 40 percent, so they are now in-line with neighboring Namibia. It is expected that this price increase will offset some of the losses experienced by BCPA members and will allow them to increase the offtake sent to abattoirs. In March 2006, a further 40 percent price increase for top-grade (prime) beef also went into effect. By July 2006, the GoB and the BMC publicly adopted the BCPA's two primary objectives—to pay export-parity prices and to support conversion to a weaner and feedlot production system. Both price increases still affect BMC profitability, but as profits for cattle producers increase, they will be able to invest in increasing their offtake which will also positively affect abattoir operations.

Conclusions: PPD Effects on the Red-Meat Value Chain

Producers are still not being paid full export-price parity. The latest price increases and the GoB's commitment to adopt weaner production and export-price parity signify to the BCPA that their long-awaited reforms are closer to becoming reality. The BCPA is now working to assist the GoB in implementing them. As the weaner system is adopted, new service providers (feedlot operators) will be necessary to support the industry. Some cattle producers are now looking to expand their businesses into feedlot operations in anticipation of the changes.

Going forward, there are several levels of the value chain that could be affected based on the BCPA's policy recommendations. The most notable are the adoption of a national auction system and the possibility of exporting directly to other markets (bypassing the BMC altogether). The BCPA continues to press for change in its industry and recognizes that continued research and reliable economic data are its best ammunition. As one observer said, "Having a unified voice was important, but more important than that was the sound economic data we presented to the government. Without that data, a unified voice would have been discounted."

Tool 12: Achieving Synergies Through Clustering

Background on Clusters

Cluster strengthening and cluster-development initiatives began to gain popularity as economicdevelopment tools in the 1980s. Since the late 1990s, cluster development and promotion programs have been applied with success in developing and emerging economies, and are starting to take hold in Africa. The cluster approach emphasizes collaboration among cluster members to achieve objectives that are beyond the capabilities of individual firms or even value chains.

"Clusters are geographic concentrations of interconnected companies, specialized suppliers, service providers, and associated institutions in a particular field that are present in a nation or region."¹²⁸ Associated institutions can include trade organizations, funding organizations, universities, research organizations, and labor unions, among others. Thus, for our purposes, the cluster is not linear as with a supply or value chain; it is an associated, highly communicating collection of participants, coordinated to focus on achieving objectives that are important to the value chain, or more often, to multiple value chains.

Clusters, which often have a geographic focus, arise naturally or can be encouraged to grow around existing assets, firms, or value chains. However, the process of clustering, approaches that encourage cluster-style communication and interaction, and associated cluster initiatives, can play a large role in strengthening firms, value chains, and the economic regions in which they operate.

As a collection of related firms, the cluster benefits from proximity and can increase those firms' competitiveness in the marketplace by exploiting synergies and linkages such as:

- Improved market access
- Firm specialization
- Access to information and market intelligence
- Bargaining power
- Policy change
- Collaboration
- Implementation of standards
- Producer-marketer collaboration
- Supply chain management
- Branding and image
- Shared investment in workforce development, certification and skills, technology, and services.

Cluster-based approaches help firms and value chains reduce transaction and information barriers, and, therefore, costs and inefficiencies, as they identify common interests and develop common strategies. This, in turn, can lead to a more formal public/private dialogue process, build

¹²⁸ Institute for Strategy and Competitiveness, Harvard Business School

collaboration, and exploit synergies among the cluster's organizations. Cooperation that arises from clustering can not only increase productivity but can also align firms to solve common problems. The strengthening of existing clusters and the fostering of cluster development can begin with a mapping exercise that helps identify potential synergies that could lead to the benefits described above.

The Relationship Between Value Chains and Clusters

At their core, clusters are centered on elements of one or many value chains. Because clusters emphasize locational factors or other commonalities, they can incorporate many value chains or portions of them that operate across clusters. Clusters can form across many links in the value chain or deepen a specific link. From a development perspective, assistance to the value chain is part of the cluster approach. Clustering is a tool to strengthen firm relationships through development of a private sector-led strategy, building of foundations for dialogue, and creation of opportunities for upgrading of skills and technology.

Cluster Initiatives

An initiative to develop clustering approaches or to strengthen clusters will reinforce the value chain's interests. In developing countries, the most successful cluster initiatives are typically a facilitated but private sector-led process.

When a cluster is promoted by an industry or value chain, its proponents, of course, select a value chain focus. When implemented with government or development partner assistance, clusters need to be selected for participation in the cluster initiative. Thus, it is important to first identify several potential core industries or value chains and then help them to self-select based on their resources, industry willingness to participate, and growth potential. Since the cluster initiative is a facilitated but private sector-led process, the second step is for the facilitator(s) who will be engaging with the industry to gather as much information as possible on broader economic and industry data, principal firms within the industry, and interests for the firms and associated stakeholders.

A clustering initiative (CI) typically takes place in five phases:

- 1. Initial research and engagement leading to the formation of a cluster initiative (usually through a memorandum of understanding)
 - Develop and present the rationale for cluster cooperation based on data gathered with identified leadership.
 - Generate strong interest in cooperation leading to mutual commitments within the cluster. Can be facilitated through a specific association or organization.
- 2. Industry diagnostics leading to strategy development
 - Cluster coordinators, industry consultants, and facilitators work with cluster members to evaluate and analyze the current state of the industry and identify potential solutions.
 - A private-sector-led industry strategy (including objectives, strategy, and an action plan) is designed and agreed upon.

- 3. Initial implementation of strategic initiatives
 - Issues outlined in the strategy are prioritized and action plans designed with coordination of those involved.
 - "Quick win" projects, lobbying for policy changes, or initial coordination with academic or associative bodies may be implemented with little funding.
- 4. Identification of projects and sources of financing
 - Larger-scale projects may need funding from within the industry, government, banks or other sources of investment.
- 5. Building sustainability
 - Continuity of cluster efforts after the CI depends on creating a strong sense of ownership and on finding leaders who champion continuing improvements.

A cluster map is valuable planning tool for analyzing the current state of the cluster and can identify where various value chains link into it. Using the map for benchmarking and gap analysis can also identify weaknesses in cross-cutting areas such as skills availability, transport, support, and infrastructure. Figure 82 offers and example of a cluster map for cut flowers in Kenya.



Figure 82: Kenya's Cut Flower Cluster

Source: Harvard Business School student team research (2007) - Kusi Hornberger, Nick Ndiritu, Lalo Ponce-Brito, Melesse Tashu, Tijan Watt

Strategic objectives of the cluster should be clearly outlined, and cluster members should be strongly committed to those goals. Implementation often includes dialogue with government and advocacy for changes in regulations and procedures. This process is easier if the cluster has a champion in government who can create traction for the implementation plan and who acts as a facilitator with the government. Intractable government policies, such as promotion of strong state-owned enterprises, could limit the growth achieved through clusters.

A successful cluster-strengthening process can help attract foreign direct investment and the participation of multinational corporations. If a cluster can develop and implement a successful strategy, it can generate internal firm investment, government investment, and external and co-investment opportunities. In addition, clustering projects can be used in conjunction with national initiatives such as special economic zones or export processing zones.

Business associations, trade associations, industry groups, and similar institutions can play an important role in cluster development and sustainability. They have convening power, can provide services in line with the industry strategy, and can advocate on behalf of the industry cluster. The cluster itself can be institutionalized for these same reasons; however, it is important that this does not happen until processes are firmly in place.

Cluster Evaluation

Clusters and cluster initiatives can be evaluated in several ways. The most important indicator of success is whether firms and cluster members are reaping the benefits of the strategies they have implemented. Such successes could include new sales and exports, reduced costs, and increased access to necessary machinery and markets, as well as to appropriate labor and technology. One framework for evaluating successes is the PAID approach developed by J.E. Austin Associates. This approach, described in more detail in the discussion of Tool 13, looks at improvements in four areas:

- 1. Process indicators
- 2. Action indicators
- 3. Investment indicators
- 4. Delivered results

Case Study: Achieving Synergies Through Clustering — Kenya Avocados¹²⁹

Introduction

The term "cluster" is often used loosely and commonly incorporates a variety of value chain elements. In most instances, it is meant to involve a geographic concentration of similar businesses within the same or related value chains. Generally speaking, this Guide has defined a cluster as a geographic concentration of interconnected companies, specialized suppliers, service providers, and associated institutions in a particular field that are present in a nation or region.¹³⁰ These clusters can incorporate or work across value chains, and, thus, are not linear. This case examines how a Kenyan market-linkage strategy, along with a cluster initiative, strengthened firms and the overall value chain in the fresh-avocado sector. The case first introduces the Kenyan avocado value chain, then describes initiatives to create and then to strengthen the value chain's market linkages. The case describes the broader avocado cluster map and the role of cluster actors.

Points to Consider

When reviewing this case, consider the following;

- Clustering is a tool to strengthen firm relationships through the development of a private sectorled strategy, build a foundation for dialogue, and create the opportunity for upgrading skills and technology.
- What were the steps taken to facilitate clustering?
- What were some of the benefits realized from establishing the avocado cluster?

Background

Mexico and the United States have dominated the global avocado export market for the last three decades. During that time, Kenya has consistently ranked as a top-20 producer. Its best ranking was 11th, in 1961.¹³¹ Kenya's main export market is Europe, to which it supplies mainly the Fuerte variety avocado. In 2003, 92 percent of avocados produced by Kenya were of the Fuerte variety, while the other 8 percent were the Hass¹³² variety. From the mid-1970s to the late 1990s, Kenya consistently ranked 19th among world producers, jumping to 13th in 1998. Kenya was able to maintain and even to improve its position based on natural market linkages to Europe and its ability to provide avocados during off-peak seasons. However, by 2000, a switch in European consumer tastes to the Hass variety, and other factors discussed below, pushed Kenya's market rank back to 16th. Unfortunately for Kenyan small growers, prices paid for the Fuerte avocado did not keep pace with other cash crops. The farmers, therefore, did not reinvest in avocado trees, and avocado quality declined.

By 2002, small Kenyan avocado farmers had no direct links with exporters for export-grade avocados. In late 2002, the government of Kenya (GoK) and USAID funded a project to improve market linkages between small avocado growers and exporters, while supporting a cluster initiative that strengthened the entire sector. A value chain analysis¹³³ identified many important points, such as the fact that the average avocado small-grower had seven or fewer trees, and only 5 percent had more than 20 trees. The analysis also confirmed the predominance of indigenous Fuerte variety avocados, which were more susceptible to disease and were in declining demand in European markets. This analysis also showed that avocado

¹²⁹ Authors: Carlton Jones and Martin Webber of J.E. Austin Associates, Inc.

¹³⁰ Institute for Strategy and Competitiveness, Harvard Business School.

¹³¹ FAOSTAT.

¹³² Deloitte, 2003.

¹³³ Conducted by Deloitte's Emerging Markets Group.

small-growers were very discouraged because of factors including rampant crop disease, little access to technical assistance, almost no access to finance, and dependence on poor-quality Fuerte avocados. With low Grade-1 fruit yields, poor prices for the little Grade-1 fruit harvested, and no market for Grade-2 fruit, some farmers cut down their avocado trees to sell for firewood.

The goal of the Kenya Business Development Services¹³⁴ (Kenya BDS) Project was to improve access to markets and access to competitive skills by supporting lead firms in backward integration strategies, which provided extension services and supply contracts to small growers. In addition to supporting lead firms, the project helped to establish small outgrower producer groups. By supporting the avocado sector through these means, it was expected that a corresponding cluster initiative would take shape that would bolster the entire value chain, bring in other actors, and improve the relationship between exporters and small growers.



Figure 83: Kenyan Avocado Production (in tons) 1994–2003

Source: Fresh Produce Exporters Association of Kenya, 2003

The Kenya Avocado Value Chain

The avocado value chain was a broker-dominated spot market whose value chain resulted in low prices and unreliable markets for growers (see accompanying figure). There were no stand-alone business services supporting the sector, nor was there a market for Grade-2 "reject" avocados. Instead, almost all small-grower avocados were sold for domestic consumption. Constraints identified in this value chain included:

- Lack of information and knowledge of the markets
- Absence of farmer marketing groups
- Limited access to inputs
- Limited access to resources, and weak incentives, for upgrading
- · Weak vertical and horizontal linkages within the value chain
- Lack of trust among producers, brokers, and exporters

After fruit was sold to brokers, those brokers then sourced it to various markets, local supermarkets, processors, and exporters. The little export-quality fruit that were produced were packed at one of seven packhouses situated within two hours of the production areas and, after pre-cooling, were loaded and transported to containers in the port of Mombasa. Average transport time to Europe was 13–21 days.

¹³⁴ Kenya BDS is a five year, USD 5,000,000 project implemented by EMG with a period of performance from September 2002–September 2007.

Case Study: Achieving Synergies Through Clustering—Kenya Avocados

As implementation of the Kenya BDS project began, lead firms were identified that were interested in participating in providing extension services to small scale growers. The first such lead firm, East African Growers (EAGA), was very interested in participating in the project after realizing the benefits of integrating backward into the supply chain to improve EurepGAP eligibility. With the support of Kenya BDS, EAGA began providing farm extension services to producer groups (comprised of 400 farmers) such as pruning, manure application, spraying, grafting, grading, collection, and transport. Later, other exporters, like Indu-Farm Ltd., joined the market linkage activity. Indu-Farm signed memoranda of understanding to provide supply contracts to producer groups and committed an agronomist to work with farmers on good agricultural practices to prepare them for meeting EurepGAP certification. Supply contracts guaranteed the market and allowed growers to focus on improving quality. The result was a twofold and even threefold increase in prices over those previously received by growers who sold to brokers.





Source Renya DD5

The Kenya Avocado Cluster

As market linkages continued to form between exporters and small growers, other supporting industries were established, supported, and ultimately relied upon to provide services to the avocado sector. As the cluster map demonstrates, these services were provided by a variety of industries and input providers.

Role of Cluster Actors

A variety of actors support the avocado cluster in Kenya. Some are existing businesses that seek to take advantage of market opportunities that exist in the newly revived avocado sector. Others are firms and entrepreneurs established through technical assistance. The cluster also includes institutions and associations that support the sector through research, certification, and advocacy. Each actor plays a useful role in the cluster and is presented in the accompanying figure in the categories of Supporting Industries and Input Suppliers/Providers.


Figure 85: Kenyan Avocado Value Chain and Cluster

Source: Kenya BDS, J.E. Austin Associates, Inc

Supporting Industries

Avocado oil processors: To facilitate a market for the otherwise rejected Grade-2 Fuerte avocados that most small growers produced, oil processors (Ruiru Natural Oils, All Green Ltd. and others) were funded by private investors, supported by the project, and linked to small growers. Other processors that already processed other fruits and nuts (Sunmango, Ltd. and Av-Oil Industries, Ltd.) purchased new equipment to retool their plants and enter the avocado sector. Through supply contracts, these processors provided a market for not only Grade-2 but also Grade-3 Fuerte avocados. Because processors required a steady supply of mature fruit for optimum operational efficiency in their plants (an average of 20 mt per processor, per day), Kenya BDS stepped in to facilitate additional producer-group supply contracts to ensure that supply was met. The oil processed by these facilities was shipped in crude form to South Africa to be refined for cosmetics. The introduction of these processors created an assured market for all grades of avocados produced on small-grower farms at a 25-50 percent higher price than that previously received from brokers.

Educational Organizations

KARI/NHRC: The Kenya Agricultural Research Institute (KARI) is Kenya's national organization for coordinating agricultural research. Its National Horticultural Research Centre focuses on horticultural research. KARI/NHRC continues to play a vital role in the avocado cluster by training producer groups and service providers on proper horticultural techniques. Many avocado technical assistance trainings also take place on KARI campuses.

Industrial Organizations

FPEAK: The Fresh Produce Exporters Association of Kenya is a member-supported private association in the horticultural industry. It is Kenya's principal association representing growers and exporters. FPEAK's mission is to develop Kenyan horticulture in the global marketplace. In addition to marketing and technical support, it lobbies for its constituents' agendas, administers Kenya-GAP (a code of practice for Kenya's horticultural industry), and promotes member compliance in other international standards.

Case Study: Achieving Synergies Through Clustering—Kenya Avocados

AAK: The Agricultural Association of Kenya is the umbrella organization for manufacturers, formulators, repackers, importers, distributors, farmers, and users of pest-control products.¹³⁵ AAK assisted in the training of agrochemical service providers on the proper and safe application of pesticide on small-grower farms. Once properly trained, agrochemical service providers are then eligible for AAK certification.

Government Organizations

KEPHIS: The Kenya Plant Health Inspectorate Services organization emerged as a partner to help train newly formed private nursery-service providers. KEPHIS is a regulatory agency for quality control of agricultural input and produce in Kenya. It coordinates all matters relating to crop pests and disease control and advises the Director of Agriculture on appropriate seeds and planting materials for export and import.¹³⁶

PCPB: A department in Kenya's Ministry of Agriculture, the Pest Control Products Board regulates the importation, manufacture, exportation, distribution, and use of pest-control products in Kenya. It ensures that only PCPB-assessed products are used in the country.

HCDA: The Horticultural Crop Development Authority is a Kenyan parastatal organization tasked with promoting and regulating the horticultural sector. It helped train agrochemical service providers in regulatory compliance and in understanding the agronomic and physiological aspects of avocados.

Input Suppliers/Providers

Raw Materials and Natural Resources

KARI Nurseries and private nurseries: KARI and KPHIS provided private nurseries with much-needed technical services to help convert and/or replenish avocado farms with Hass varieties. Services provided included top-working, pruning, planting, and providing root-stock and scion materials for producer groups. Each participating nursery was inspected and registered with HCDA

Technology

Agrochemical spray service providers: One identified service that avocado-producer groups needed was the introduction of agrochemicals to mitigate the risk of pest infestation and disease. Commercializing this service aspect was vital to ensuring the vitality of the avocado value chain. As such, service providers were identified, trained, and equipped through KARI/NHRC, KEPHIS, AAK, and PCPB so that the providers understood the complexities of safe and appropriate pesticide applications. The integrated pest-management program had two main components: developing and mentoring independent service providers and bringing in commercial financial arrangements that enabled farmers to access these services. The service providers came from varying backgrounds; some were drawn from within the farming community, while others were brokers who saw an opportunity to remain vitally linked to the changing avocado sector.

Top-working services providers: With a high percentage of Fuerte avocados on each small-grower farm, extension workers promoted on-farm diversification using Hass varieties. Severing as ecological insurance against variety specific pests and diseases, Hass avocados also showed a slight price preference on international markets. More importantly, the campaign served to mobilize economically marginal native trees, already common on farms, for export production. This mobilization relied on grafting, the technique of attaching fruit producing branches of one tree to another. To facilitate this conversion, top-working commercial grafting services had to be introduced to the farmers. Awareness training provided for producer groups created demand for the service providers, and roughly 7,000¹³⁷ farmers pre-registered for variety conversion. In some districts, the demand for variety conversion

¹³⁵ AAK- http://www.agrochem.co.ke/.

¹³⁶ KEPHIS, http://www.kephis.org/.

¹³⁷ Deloitte, July 2005.

exceeded 100,000 trees nearly twice the program's capacity. With the average price of top-working estimated at KSh 250/tree, conversion also presented a significant business opportunity for entrepreneurs willing to be trained in this intervention. As top-workers were trained, they were linked with producer groups to begin servicing trees. In as little as 12 months, the trees began to produce the new, higher-valued Hass avocados.

Capital

Equity Bank: Addressing a major constraint in the avocado value chain, Equity Bank provided credit facilities that enabled avocado-producer groups to purchase agrochemical services on credit. Only producer groups with existing supply contracts and a history of buying and selling to a lead exporter were eligible to participate in this program. After applying for credit, eligible producers received their answer within 48 hours. The credit covered the labor for spraying, as well as the purchase and delivery of pesticides. All the costs were covered under a fixed-price-per-tree-sprayed plan, which is paid at the start of avocado season and then deducted by the bank on a draw-down basis every time spraying takes place or upon payment for the avocados delivered. Credit facilities were also established for service providers to buy and maintain mechanical spray-pump equipment, rent agrochemical stores, and purchase fuel.

Machinery and Equipment

Achelis Kenya Ltd.: This subsidiary of a Taiwanese import/export company played an integral role in the avocado cluster by selling and training service providers in its motorized agrochemical-sprayer product line.

Other machinery providers included Israeli firms who supplied pressing equipment used for Grade-2 and Grade-3 avocado processing.



Figure 86: Kenyan Avocado Exports 1975–2005

Results

Like many of this Guide's cases, the Kenyan avocado case is ongoing. By late 2006, the program had worked with 14,240 small-holder farmers organized into 475 producer groups.

Each small grower is accessing embedded services in extension, produce collection, transportation, and a guaranteed market whose prices average 3–4 times those of the brokers. Grade-1 fruit yields have increased from less than 15

percent to over 65 percent per tree. Participating farmers are collectively averaging between 30 and 40 MT of avocado production per day. In addition, the introduction of avocado processors has provided a stable market for all small-grower production grades and varieties, with crude oil exports reaching South Africa and Italy.

Other benefits from the backward linkages interventions and cluster initiatives include:

- Improved quality control
- A clear path of transition from Fuerte to Hass avocado production
- Faster and cheaper access to inputs

• Improved market linkages and information sharing

One of the program's most significant results is the commercial viability of the entire sector as demonstrated by the "replicated effect" that took place among exporters. Besides EAGA, four other lead firms have adopted the program and are working down-stream with contract growers.

Tool 13: Monitoring Achievements in Value Chain Performance

Purpose of Monitoring

Monitoring and evaluation (M&E) methods can be used as a management tool that enables value chain participants and promoters to track implementation progress, evaluate value chain performance, and identify the impact of initiatives. In its most basic form, M&E uses indicators to identify progress toward achieving

Questions to Ask

 What does the value chain really want to monitor?
 At what level does the value chain measure the impacts?

intermediate targets or ultimate goals. Monitoring systems comprise procedural arrangements for data collection, analysis, reporting, and feedback. In business, monitoring is used to measure the return on inputs as a means to verify an investment's success before returns are realized, as well as to receive guidance on prioritization and decision-making for business programs.

M&E is, of course, already a well-developed field of practice with many established and effective implementation methods and processes. In private-sector development, monitoring and evaluation methods have three principal modes: project monitoring, performance evaluation, and impact assessment. Effective impact assessment often depends on good program evaluation and, in turn, systematic project monitoring.

Project monitoring (PM) refers to the systematic collection and analysis of data and information about inputs, activities, and outputs. It is used for management purposes – to monitor progress against a plan, identify emerging issues and problems, and take corrective measures.

Performance evaluation (PE) is a periodic study at key points in a project cycle that examines a program or organization's performance more broadly (e.g. looking at staff behavior, organizational structures and methodologies, and the efficient achievement of outputs). Performance evaluations should assess the effectiveness of different factors being used to create change and as opportunities for implementing organizations to learn about themselves.

Impact assessment (IA) measures outcomes and examines changes at higher levels of a program (i.e. the goal and purpose level). IA's effectiveness depends on having a baseline or control group with which to compare results so that changes can be properly attributed. IA should be able to identify which factors contributed to changes and in what fashion. Impact assessment exercises are also key opportunities for both program staff and beneficiaries to use in monitoring progress and affecting the evolution of program design.¹³⁸

The focus of many M&E implementations has been on the "project" cycle of international development interventions. As such, the many insights gained from these studies have had many positive effects on increasing the efficacy and efficient implementation of projects and their goals. Unfortunately, the design and implementation of performance-monitoring methods useful

¹³⁸ Adapted from an online contribution from Mike Albu of Practical Action found at: "Impact Assessment: An Online Speaker's Corner Discussion led by Gary Woller and hosted by microLinks.org" September 26-28, 2006. More tools available online at: http://www.enterprise-impact.org.uk/.

for measuring the actual performance of a value chain and of its participants (rather than a project) have yet to be properly adapted for use in private-sector development contexts.

Thus, for the purposes of implementing value chain improvements, the challenge is to provide methodologies that can be used by value chain participants to monitor the implementation of their own programs. Value chain participants will be interested in such monitoring to maintain focus, reach objectives, and track changes in value chain performance and development, including increases in productivity and other business measurements of success.

Monitoring activities should provide information about the current status of a project and general trends pertinent to the value chain. Some key points on monitoring (and related evaluation) include:

- Measurements are periodically carried out.
- Monitoring is done for the specific purpose of checking the status of the agreed-upon process or initiative or to evaluate progress toward a strategic or management objective.
- The results of monitoring should be evaluated and should provide guidance for action.

While there is often a great deal of anecdotal evidence of the impact of value chain initiatives and actions, formal efforts to provide clear measurements against a baseline or against a program

of strategy implementation are often lacking. Some problems with traditional monitoring approaches include:

- Systems that are not attuned to the industry's measuring vocabulary
- Difficulty in attributing changes to program interventions
- Monitoring that does not necessarily provide insights into better business practices to drive the industry forward
- Monitoring responsibility that is not clearly delegated or resources that are not allocated

Implementing Monitoring in Value Chains

Monitoring interventions require stakeholders to determine which indicators are most appropriate

Three Different Perspectives on M&E

- Business perspective: monitor VC performance and provide feedback to inform decisionmaking
- Process perspective: provide guidance on value-chain participants' focus as they reach their objectives
- Programmatic: ensure that the money spent on implementation is achieving the intended results

and useful for tracking program implementation and value chain performance. Traditionally, measurements of productivity or "value addition" are used to convey improvements in performance. While measuring productivity is sufficient for measuring business performance, it does not necessarily provide measurement for implementation strategies, especially when the impact of implementation on value chain performance is seen only in the medium- or longer-term. Therefore, process indicators are also needed.

The level at which an impact measurement should be taken will depend on the nature of the initiative. In various circumstances, measuring at the firm level, value chain aggregate level, country or national level, or individual level may all be appropriate.

Figure 86¹³⁹ illustrates an example of how value chain initiatives can begin to formulate effective monitoring and related evaluation programs. These require concerted actions by the various value chain participants. When assisting the value chain to develop its own monitoring and evaluation process, one must consider the following proposed steps.

- The value chain must be understood by expert analysis
- Key benchmarks must be determined
- Industry leaders must be consulted
- Program design must include M&E
- Different programs and initiatives must be linked to the value chain to take advantage of synergies
- Performance measurements inform effectiveness

Value Chain Performance Monitoring



Figure 87: M&E VC Model

Source: adapted from multiple FIAS-IFC presentations on Value chains by Subramanian & Sur

Box 17: Standards for Ghanaian Pineapples

Ghana's pineapple market has been built on strong entrepreneurialism from the private sector and a demand for Ghanaian sea-freighted pineapples in Europe, especially outside the core French market. During the late 1980s and early 1990s, Ghana carved a niche in the EU market as a primary supplier of top-quality, air-freighted pineapples, controlling about 60 percent of the estimated annual 20,000 mt of pineapples air-freighted to Europe. A small amount (2,710 mt) were sea-freighted, but the bulk were air-freighted (13,054 mt worth over USD 400,000). To increase exports to EU markets, sea-freighting was explored. The Sea Freight Pineapple Exporters of Ghana (SPEG) is a consortium of 14 exporter firms supported by USAID and the Government of Ghana (GoG). SPEG was established to facilitate sea-freighting Ghanaian pineapple varieties by gaining the necessary scale to operate at competitive prices.

¹³⁹ Taken from a presentation at the IFC on value chains.

To ensure quality in its pineapple exports, SPEG worked with the Ghana Standards Board to develop a poster that would be used by producers and packers alike as a simple guide to the minimal quality standards required for exporting. The easy-to-read poster was posted in strategic

Figure 88: Sample SPEG Customer Showing Desired Pineapple Qualities - USAID



locations for exporters, traders, and farmers so that all actors within the Ghana pineapple chain consistently maintained the standards for color, shape, crown condition, and other pineapple characteristics.

Author: Carlton Jones – J.E. Austin Associates

In practice, the evaluation focus for value chains (and clusters) has been monitored in valueadded increases as a measure of productivity improvement. However, single, quantitative measures can ignore the process side of a value chain initiative. One comprehensive evaluation approach is the PAID framework,¹⁴⁰ a measurement system for evaluating value chain initiatives that has been adapted from the approach to cluster evaluation.

The PAID M&E framework (as featured in Box 18) not only measures delivered results but also co-investment by both the private and public sectors. This tool can be used in value chain projects on the basis of proper benchmarks being determined by participant organizations and

¹⁴⁰ Developed by J.E. Austin Associates and applied in several countries.

institutions. For example, the *Process Indicators* measure the ability of value chain (VC) practitioners to engage the proper VC-leading authorities and market players. The *Action Indicators* could track the VC initiative's progress by looking at the changes that have occurred since the intervention. The *Investment Indicators* gauge the "buy-in" by VC participants and those actors indirectly determining the health of the VC's competitiveness. *Delivered Results* measure the essential goal of any value chain initiative, which is to increase productivity, capture value, create jobs, and grow business.

Box 18: The PAID M&E Framework

<u>**P**</u>rocess Indicators track the ability of the implementers to engage the leadership of the value chain, elicit a strong response, and structure a collaborative agreement. These indicators should be set at the project outset and are very important during the first year of activities.

<u>A</u>ction Indicators track whether progress is being made in implementing the strategic initiatives identified in the first phase of the project. These cannot be programmed at the outset.

Investment Indicators track co-investments by counterparts. Co-investment typically begins modestly (through the contribution of cluster executive time, provision of workshop venues, etc.) but then accelerates as the initiative moves to maturity. It may take time to prepare feasibility studies and projects and to secure financing. The most significant investments occur at the implementation stage of action initiatives.

<u>D</u>elivered Results Indicators focus on increases in productivity and value added but may also include export revenues, employment growth, new enterprise growth, average wages, and average profitability in the industry. It must also be demonstrated that the project contributed directly (wholly or partially) to delivering these results. The results become measurable after some time, often years. However, there are delivered results that are more qualitative. The emergence of trust and cooperation within an industry, business-government and academic collaboration, changes of mindset, and spread effects can be observed but are not easily quantified.

Author: Kevin X. Murphy – J.E. Austin Associates

Conclusion

Methodologies such as the PAID framework can be effectively applied to value chain initiatives as long as they are designed to measure impacts experienced by the various value chain participants and the various segments of the VC. Monitoring for value chains should focus on two types of performance: 1) implementation of strategy and 2) increases of productivity. This Guide has presented 13 different types of tools and approaches that one can associate with implementing value chain programs. Most of these tools can be targets for monitoring, given measurement indicators before, during, and after an intervention.

SUMMARY AND CONCLUDING THOUGHTS

Value chains are a key framework for understanding how a product moves from the producer to the customer. As such, the value chain perspective provides an important means to understand the business-business relationships, mechanisms for increasing efficiency, and ways to enable business to increase productivity and add value. It provides a reference point for improvements in services and the business environment. It is a vehicle for pro-poor initiatives and for linking small businesses with the market.

Value chains reside at the core of high-impact and sustainable initiatives focused on improving productivity, competitiveness, entrepreneurship, and SME growth.

Each of the tools and approaches described in this Guide contributes to understanding value chains and implementing improvements in value chains. The tools (and accompanying cases) incorporate several repeated principles (described in Section 3) that are important to almost any action:

- Market access
- Market focus
- Partnering
- Collaboration and information sharing
- Innovation

And of course the six program implementation themes described in Section 4.

- Designing strategies and business plans—Obtaining and using information
- Developing robust new businesses
- Supplying the market—Aligning supply to match market opportunity
- Reaching the market—Market positioning and market opportunities
- Improving the business and policy environment
- Monitoring results in value-chain development

As described, the intent has been to present a selection of tools and approaches that are crucial to implementing value chain concepts. But space constraints have limited the number of tools, and we hope that future work will add to the toolkit. Some of the elements of future work might usefully include tools and approaches focused on elements such as:

- Market-based approaches (segmentation, branding)
- Logistics as a tool
- Managing the distribution chain
- Branding (i.e. connecting differentiation with standards and with service qualities)
- Workforce and skills competitiveness
- Jumpstarting value chains

While the cases discuss many products, they include relatively little description of highly transformed products. There is also a bias towards exported agricultural products. Both of these areas would merit future case focus.

Space limitations have restricted the number of cases that could be included in the Guide. There are opportunities to describe many more experiences in future work. Some of the cases that were excluded at the last minute from this document because of space constraints were:

- Sri Lanka Coir—Benchmarking the Value Chain and Integrating the Information Into Strategy
- Achieving Economies of Scale in Uganda's Coffee Sector
- Sri Lankan Rubber—Stretching the Value Chain to Value-Added Products
- Sharing Information in the Value Chain Senegal IFLEX

We hope that this Guide provides a first step in understanding implementation challenges and approaches and that the many stories of successful value chains encourage businesses and practitioners.

The Guide can be disseminated in many ways and used as the basis for:

- Training programs
- Project preparations
- Encouraging additional research and case writing
- Future Value Chains Guides or Toolkits

If this Guide assists program design and implementation, provides useful approaches to businesses in their decision-making, planning, and operations, and encourages further research, then we will be delighted that it has served its purpose.

Bibliography

Sections 1-3 and General Sources:

- Altenburg, Tilman. April 2006 draft. "Donor Approaches to Supporting Pro-poor Value Chains." German Development Institute, Report prepared for the Donor Committee for Enterprise Development Working Group on Linkages and Value Chains
- Asian Development Bank. January 2005. "Agricultural Commercialization, Value Chains, and Poverty Reduction." Making Markets Work Better for the Poor, Discussion Paper no. 7.
- Austin, James E. "Managing in Developing Countries: Strategic Analysis and Operating Techniques," Free Press, 2002.
- Austin, James E. "The Collaboration Challenge," Jossey-Bass, 2000.
- Austin, James E. "Strategic Management in Developing Countries," The Free Press, 1998.
- Austin, James E. "Agroindustrial Project Analysis," World Bank/Johns Hopkins Press, second edition, 1992.
- Bernet, Thomas, André Devaux, Oscar Ortiz and Graham Thiele. 2005. "Participatory Market Chain Approach." BeraterInnen News 1/2005. Agridea International.
- Bolnick, Bruce, Andrea Camoens, and Julia Zislin of Nathan Associates Inc. August 2005. "Tanzania Economic Performance Assessment," USAID Country Analytical Support Report, p. 14.
- Canadian International Development Agency. October 2004. "Programme de Développement des Marchés Agricoles (PDMAS) - AgMarkets Sénégal: Horticulture du Sénégal Cadre stratégique – Horizon 2010." GEOMAR.
- Development Alternatives, Inc. for USAID. January 2005. "Strengthening the Role of AIDS-Affected SMEs in Productive Markets: BDS on the Margins HIV/AIDS Paper." microREPORT #27.
- Dunn, Elizabeth, 2005. "AMAP BDS Knowledge and Practice Task Order Lexicon." microNOTE No. 6, USAID, Washington, DC.
- The Economist. February 15, 2007. "Big Chains Enjoy a Buyer's Market."
- English, P., S. Jaffee and J.J. Okello. 2006. "Exporting out of Africa: The Kenya
- Faivre Dupaigre B., Baris P., and Liagre L., 2004. "Étude sur la Competitivité des Filières Agricoles dans l'Espace UEMOA", IRAM, UEMOA Commission-DDR.
- Food and Agriculture Organization of the United Nations, FAOSTAT located at: <u>http://faostat.fao.org/default.aspx</u>
- Fu, Hsin-Pin, Tien-Hsiang Chang, Wan-I Lee, and Cheng-Chin Lu. 2006. "A Study on the valuechain strategies of enterprises adopting electronic marketplaces in free and controlled markets." Journal of Manufacturing Technology, vol. 17, Issue: 5, pg. 621 – 632.
- Gabre-Madhin, Eleni and Nicholas Minot. 2004. "Successes and Challenges in Promoting Africa's Horticultural Exports." International Food Policy Research Institute IFPRI and World Bank.
- Gereffi, Gary and Miguel Korzeniewicz. 1993. "Commodity Chains and Global Capitalism." Greenwood Publishers.
- Gibbon, Peter. November 2003. "Commodities, Donors, Value-Chain Analysis and Upgrading." Danish Institute for International Studies, paper prepared for UNCTAD.
- Gupta, S. & Y. Yang. "Unblocking Trade: To underpin growth, African needs to adopt a comprehensive approach to boosting trade." *Finance and Development Magazine*, December 2006, Volume 43, Number 4.
- Haggblade, Steven J. and Matthew Gamser. November 1991. "A Field Manual for Subsector Practitioners: Tools for Microenterprise Programs: Nonfinancial Assistance Programs." Growth

and Equity through Microenterprise Investments and Institutions (GEMENI) Program, funded through microLINKS, USAID.

- Horticulture Success Story." In R. B. Liebenthal and M.L Fox (Eds) *Attacking Africa's Poverty: Experiences from the Ground*. World Bank, Washington DC.
- Humphrey, John. January 2004 revised. "Commodities, Diversification and Poverty Reduction." Paper presented at FAO symposium on the State of Agricultural Commodity Market Research Rome, 15-16 December 2003.
- Humphrey, John. 2005 "Shaping Value Chains for Development: Global Value Chains in Agribusiness." GTZ, Eschborn, Germany.
- Humphrey, John and Antje Oetero. 14 November 2000. "Strategies for Diversification and Adding Value to Food Exports: A Value Chain Perspective." UNCTD.
- Humphrey, John and Hubert Schmitz. 2000. "Governance and Upgrading: Linking Industrial Cluster and Global Value Chain Research." University of Sussex, Institute of Development Studies. IDS Working Paper 120.
- Humphrey, John and Hubert Schmitz, 2001. "Governance in Global Value Chains." University of Sussex, Institute of Development Studies. Published in IDS Bulletin, Vol. 32, No 3, 2001.
- Ingram, Michael. "Summary of Kenya Value Chain Analysis." Note No.8, September 2005, World Bank.
- Janecka, Izabela. March 2006. "Sourcing in Low-Cost Countries." Eye for Procurement.
- J.E. Austin Associates. May 2006. "Rwanda Export Promotion Action Plan." Submitted to the Government of Rwanda: Rwanda Investment and Export Promotion Agency and Ministry of Finance and Economic Planning. With assistance from the World Bank.
- Kaplinsky, Rafael. March 2000. "Spreading the Gains from Globalization: What can be learned from value chain analysis?" unpublished paper, Institute for Development Studies, University of Sussex.
- Kaplinsky, Rafael and Jeff Readman. 2001. "Integrating SMEs in Global Value Chains: Towards Partnership for Development." UNIDO.
- Kaplinsky, Rafael and Mike Morris. 2002. "A Handbook for Value Chain Research." IDRC. http://www.ids.ac.uk/ids/global/pdfs/VchNov01.pdf
- Kula, Olaf, Jeanne Downing, and Michael Field. February 2006. "Globalization and the Small Firm: A Value Chain Approach to Economic Growth and Poverty Reduction." USAID microREPORT #42, prepared under the Accelerated Microenterprise Advancement Project Business Development Services Knowledge and Practice Task Order
- Lusby, Frank and Henry Panlibuton. April 2004. "Promotion of Commercially Viable Solutions to Subsector and Business Constraints." Office of Microenterprise Development, USAID.
- McCormick, Dorothy and Hubert Schmitz. November 2001. "Manual for Value Chain Research on Homeworkers in the Garment Industry." IDS.
- Miehlbradt, Alexandra O., and Mary McVay. 2006. "Implementing Sustainable Private Sector Development: Striving for Tangible Results for the Poor: The 2006 Reader." SDC, ILO.
- Moran, Theodore H. August 2001. "Parental Supervision: The New Paradigm for Foreign Direct Investment and Development." *Policy Analyses in International Economics*, Issue 64.
- Murphy, Kevin. "Private Sector Contribution to Education Reform: 10 Bridges to a Competitive Workforce." J.E. Austin Associates.
- Murphy, Kevin. September 2003. "PAID Methodology: Monitoring and Evaluation of Cluster Competitiveness Initiatives." J.E. Austin Associates.
- Nugawela, Patrick and Roland Oroch. April 2005. "Cashew Sub-sector Strategic Framework." DfID.
- Porter, Michael E. "Competitive Strategy." Free Press, 2004.

- Porter, Michael E. "Attitudes, Values, Beliefs, and the Microeconomics of Prosperity," in Culture Matters: How Values Shape Human Progress, (L.E. Harrison, S.P. Huntington, eds.), New York: Basic Books, 2000.
- Porter, Michael E. "Location, Competition and Economic Development: Local Clusters in a Global Economy," Economic Development Quarterly 14, no. 1, February 2000: 15-34. Also reprinted in Globalization and the Location of Firms (J. Cantwell, ed.), Cheltenham: Edward Elgar Publishing, 2004.
- Porter, Michael E. "On Competition." Boston: Harvard Business School Press, 1998.
- Porter, Michael E. "Competitive Advantage: Creating and Sustaining Superior Performance." Free Press, 1998.
- Porter, Michael E. "The Competitive Advantage of Nations." Free Press, 1998.
- Posthumus, Hans. 14 May 2004 Draft. "Capacitating Sector Analysis: A Practical Training Methodology to Analyze Value Chains." CAPSA.
- Raikes P., Jensen M.F., and Ponte S. 2000. "Global Commodity Chain Analysis and the French Filière Approach: Comparison and Critique." Economy and Society, Volume 29, Number 3, 1 August 2000, pp. 390-417(28)
- Raney, Terri et al. 2005. "The State of Food and Agriculture: Agricultural trade and poverty: Can trade work for the poor?" FAO Agriculture Series (SOFA)-36.
- Reardon, Thomas, and Luis Flores. May 2006. "Customized Competitiveness Strategies for Horticultural Exporters: Central America Focus with Lessons from and for other Regions." Food Policy, Vol 31, Issue 6, December 2006, pg. 483-503.
- Roduner, Daniel. July 2004. "Draft Report on Value Chains: Analysis of existing theories, methodologies and discussions of value chain approaches within the development cooperation sector." Swiss Agency for Development and Cooperation (SDC).
- Roduner, Daniel, Agridea Gerrits and Andreas Gerrits. February 2006. "Compilation of insights on the online debate Value Chains in Rural Development (VCRD): The role of Donors in Value Chain Interventions." Swiss Agency for Development and Cooperation
- Salinger, BL and JD Stryker. November 2004. "Guide to commodity-based export diversification and competitiveness strategies for African countries." report prepared for UNCTAD/JITAP, Cambridge, MA: AIRD.
- Springer-Heinze, Andreas. 15 June 2004 draft. "Info-Cadena: Instruments to Foster Value Chains." GTZ.
- Steen, Cynthia, Rich Magnani and Lara Goldmark. June 2005. "Competitive Strategies for Agriculture-related MSES: From Seeds to Supermarket." USAID, prepared by Development Alternatives Inc. microREPORT #37
- Sturgeon, Timothy J. "How Do We Define Value Chains and Production Networks." Background Paper Prepared for the Bellagio Value Chains Workshop, September 25 October 1, 2000. Rockefeller Conference Center, Bellagio, Italy. Institute of Development Studies (IDS).
- TechnoServe. November 2004. "Partnerships for Agribusiness Development, Agricultural Trade, and Market Access; A Concept Note for NEPAD."
- Temu, Andrew E., and Anna A. Temu. 2005. "High Value Agricultural Products for Smallholder Markets in Sub-Saharan Africa: Trends, Opportunities and Research Priorities." Prepared for An International Workshop on How Can the Poor Benefit from the Growing Markets for High Value Agricultural Products? 3rd to 5th October 2005 International Center for Tropical Agriculture, Cali, Colombia. ICTA.
- van Roekel, Jan, Ronald Kopicki, Carry J.E. Broekmans, and Dave M. Boselie. 2001. "Building Agri-Supply Chains Issues and Guidelines." World Bank, Washington, DC.
- van Roekel, Jan, Sabine Willems and Dave M. Boselie. 19 August, 2002. "Agri-Supply Chain Management: To Stimulate Cross-Border Trade in Developing Countries and Emerging Economies." World Bank.

- Webber, C. Martin, January 2006, "Approaches to Economic Growth Value Chains and Competitiveness." prepared for the workshop on *Nurturing the Sources of Economic Growth in Tanzania*, held in Dar es Salaam, Tanzania, January 19-21, 2006. J.E. Austin Associates.
- Webber, C. Martin. January 2006. "Tanzania: Value Chain and Competitive Strategy Current Approaches to Private Sector Development and Economic Growth." PowerPoint Presentation. J.E. Austin Associates.
- Webber, C. Martin. March 2006. "Enabling an Inclusive Private Sector: Opportunities and Key Constraints." J.E. Austin Associates.
- Webber, C. Martin. May 2006. "Enabling an Inclusive Private Sector: Opportunities and Key Constraints in Burundi." PowerPoint Presentation. J.E. Austin Associates.
- Webber, C. Martin. August 2006. "*Uganda National Competitiveness Workshop:* International Experience in Setting and Implementing Competitiveness Strategies." PowerPoint Presentation. J.E. Austin Associates.
- Westlake, Michael. 2005 "Addressing marketing and processing constraints that inhibit agrifood exports: A guide for policy analysis and planners." FAO Agricultural Services Bulletin 160, Rome.
- World Development Indicators 2007, World Bank.

Tool 1: Choosing Priority Sector Value Chains for Interventions

Box Case Study: The Example of Senegal's Projet Croissance Economique

- J.E. Austin Associates for IRG. October 2006. "Support for Accelerated Growth and Increased Competitiveness, Cashew Value Chain, Senegal: Analysis and Strategic Framework for Sub-Sector Growth Initiatives." Washington, DC.
- J.E. Austin Associates for IRG. October 2006. "Support for Accelerated Growth and Increased Competitiveness, Mango Value Chain, Senegal: Analysis and Strategic Framework for Sub-Sector Growth Initiatives." Washington, DC.
- J.E. Austin Associates for IRG. September 2006. "Bissap Value Chain Senegal, Analysis and Strategic Framework for Sub-Sector Growth Initiatives." Washington, DC.
- J.E. Austin Associates for IRG. September 2006. "Support for Accelerated Growth and Increased Competitiveness, Artisan Textile Value Chain, Senegal: Analysis and Strategic Framework for Sub-Sector Growth Initiatives." Washington, DC.
- Nugawela, Dr. Patrick. February 2007. "La Chaine de Valeurs de la Filiere Cajou au Sénégal." Value Chain Presentation.

List of Sources Interviewed:

- Dr. Patrick Nugawela, J.E. Austin Associates, USAID Projet Croissance Economique au Sénégal
- C. Martin Webber, J.E. Austin Associates

Case Study: Prioritizing Value Chains by Using Comparative Analysis – Value Chain Selection in Mozambique

- Gergely, Nicolas. April/May 2005. "Economic Analysis of Comparative Advantage for Major Agricultural Cash Crops in Mozambique." FAO.
- Master, William A. and Alex Winter-Nelson. May 1995. "Measuring the Comparative Advantage of Agricultural Activities: Domestic Recourse Costs and the Social Cost-Benefit Ration." American Journal of Agricultural Economics. Vol. 77, N. 2, pp.243-250. <u>http://links.jstor.org/sici?sici=0002-9092%28199505%2977%3A2%3C243%3AMTCAOA%3E2.0.CO%3B2-I</u>

- O'Keefe, Thomas. November 2001. "A Study of Revealed Comparative Advantage for AGOA-Eligible SADC Countries: Identifying Products Afforded Genuine Preferential Access to the U.S. Market under the African Growth and Opportunity Act." Mercosur Consulting Group, Ltd. Submitted by Chemonics International, Inc. Gaborone, Botswana.
- World Bank. February 23, 2006. Mozambique Agricultural Development Strategy: Stimulating Smallholder Agricultural Growth. Report No. 32416-MZ. World Bank, Agriculture, Environment, and Social Development Unit, Africa Region.

List of Sources Interviewed:

- Nicolas Gergely, Independent Consultant
- Patrick Labaste, The World Bank

Case Study: Mali Geomar Value Chain Strategy

- Geomar International. July 2006. "Profil Stratégique pour le Développement de La Compétitivité des Filières Agricoles du Mali : Identification des Opportunités Stratégiques de Valorization et de Diversification de l'Agriculture Malienne, Version Préliminaire." Montreal, QC.
- World Bank. June 6, 2005. "Project Appraisal Document on a Proposed Credit in the Amount of SDR 30.7m to the Republic of Mali for an Agricultural Competitiveness and Diversification Project." Washington, DC.

Tool 2: Designing Informed Strategies Across the Value Chain

Case Study: Understanding the Value Chain and Integrating Information into Strategy: Nigerian Domestic Catfish

- Dixie and Ohen. May 2006. "The Market for Catfish and Other Aquaculture Products in Nigeria."
- Engelmann and Swisscontact. June 2005. "PLP Technical Note #4: An Inventory of BDS Market Assessment Methods for Programs Targeting Microenterprises." Swisscontact. http://www.seepnetwork.org/files/2991_file_TechNote4_c.pdf
- EurekNews & BBC News. http://news.bbc.co.uk/2/hi/science/nature/4187522.stm
- "Hidden Harvests: Unlocking the Potential of Aquaculture in Africa." August 22-25, 2005. Technical Review Paper Aquaculture. NEPAD "Fish for All" Summit. Abuja, Nigeria.
- Moehl, Halwart, and Brummet. 2005. "Report of the World Fish Center Workshop on Small-Scale Aquaculture in SSSA." FAO.
- TechnoServe. November 2004. "Partnerships for Agribusiness Development, Agricultural Trade, and Market Access: A Concept Note for NEPAD." Washington, DC.

Tool 3: Conducting Benchmarking and Gap Assessments of Value Chains

Box Case Study: Uganda Benchmarking Constraints in the Coffee Industry

- Chaffee, Ralph. April 2, 2002. "Enterprise Linked Extension Services in Uganda." USAID Support for Private Enterprise Expansion and Development (SPEED) Project, Uganda.
- Uganda Bureau of Statistics, http://www.ubos.org/.

List of Sources Interviewed

• Mark Wood, USAID APEP Project

Case Study: Uganda Floriculture

- Gabre-Madhin, IFPRI and Vette, V.E.K. Adviesgrouep B.V. for the World Bank. "Uganda Horti-floriculture Sector Technical Note 2." Prepared for ESSD and PSD Departments, Africa Region, World Bank in the framework of the Uganda Export Growth and Competitiveness Project.
- Vereniging van Bloemenveilingen Nederland (VBN)
- Gabre-Madhin, Eleni Z. and Hans de Vette, 2004, "Uganda Horticulture Sector Development Study," prepared in the framework of the Private Sector Competitiveness II Project, the World Bank.

Tool 4: Upgrading and Deepening the Value Chain

Box Case Study: Mongolian Meat Industry—Upgrading the Value Chain

• Written by C. Martin Webber with information from J.E. Austin Associates and Nathan Associates.

Box Case Study: Deepening the Value Chain: Glass Jar Production in Armenia

- Written by C. Martin Webber with information from secondary sources from the field
- E-mail correspondence with Sevak Hovhannisyan, USAID/CAPS Project, Armenia

Case Study: Kenya Green Beans

- Knopp, David. October 2005. "From Seed to Shelf: Value Chain Dynamics in Kenyan Horticulture." Value Chain Presentation. Emerging Markets Group. Arlington, VA.
- Jaffee, Steven. 2003. "From Challenge to Opportunity: Transforming Kenya's Fresh Vegetable Trade in the Context of Emerging Food Safety and Other Standards in Europe." Agriculture and Rural Development Discussion Paper. World Bank. Washington, DC.
- Hallam, David, et al. 2004. "The market for non-traditional agricultural exports." Commodities and Trade Division, FAO.
- Irwin, Grant, Parker, and Morgan. January 2005. "Strengthening the Role of AIDS-affected MSEs in Productive Markets BDS on the Margins HIV/AIDS Paper: microREPORT #27." USAID.
- Minot, Nicholas and Margaret Ngigi. August 2004. "Are horticultural exports a replicable success story? Evidence from Kenya and Côte d'Ivoire." International Food Policy Research Institute and Egerton University, paper presented at the InWEnt, IFPRI, NEPAD, CTA conference "Successes in African Agriculture" Pretoria December 1-3, 2003.
- Okado, Mark. 1999. UN Conference on Trade and Development, "Background Paper on Kenya Off-season and Specialty Fresh Vegetables and Fruits: Lessons of Experience from the Kenya Horticulture Industry."

List of Sources Interviewed:

• David Knopp, COP-USAID Kenya BDS Program, Emerging Markets Group

Tool 5: Identifying Business Models for Replication

Box Case Study: Replicable Business Models - Rwanda Coffee Washing Stations

• Associated Press. March 3, 2006. "Rwandan Coffee in Your Cup: Starbucks Introduces Product in N. America." Montreal Gazette.

- Bureau pour le Developpement de la Production Agricole (BDPA). September 2005. "Analyse Economique et Financière de la Filière Café." France.
- Chemonics International, Inc. 2001. "Assistance a la Dynamisation de l'Agribusiness au Rwanda: ADAR, Year 1 Work Plan." Washington, DC.
- Chemonics International, Inc. June 2002. "Considerations for Initiating a Specialty Coffee Industry in Rwanda." Washington, DC.
- Chemonics International, Inc. July 2004. "ADAR Agribusiness Development Assistance 2004: 2nd Qtr Progress Report, April 1 to June 30, 2004." Washington, DC.
- Chemonics International, Inc. December 2005. "Assessing USAID's Investments in Rwanda's Coffee Sector: Best Practices and Lessons Learned to Consolidate Results and Expand Impact." Washington, DC.
- J.E. Austin Associates. May 2006. "Rwanda Export Promotion Action Plan." Submitted to the Government of Rwanda: Rwanda Investment and Export Promotion Agency and Ministry of Finance and Economic Planning. With assistance from the World Bank.
- Lecraw, Donald. January 31, 2005. "The Coffee, Tea, Tourism, and Handicrafts Sectors: Export Potential and Contribution to Growth and Poverty Reduction, draft version." Integrated Framework, Diagnostic Trade Integration Study.
- Office des Cafés du Rwanda (OCIR CAFÉ). November 2005. "New Action Plan 2006-2008, For the Development of the Rwanda Coffee Sector."
- On the Frontier. July 2002. "Rwanda Coffee Strategy and Action Priorities, Rwanda Competitiveness and Innovation Program." Work Group on Coffee. Presentation.
- On the Frontier. April 2004. "The Rwandan National Competitiveness and Innovation Project: A National Transformation Process." Washington, DC.

Box Case Study: Dairy Pakistan – Identifying and Replicating Business Models within the Value Chain

- The PISDAC II Evaluation Report, Marco Arocha, J.E. Austin Associates, 2006.
- The White Revolution, report by Justin Stokes, J.E. Austin Associates, February 2006.

Case Study: Identifying and Implementing Replicable Business Models – Mozambique Cashews

- AfricanCashew.com. 20 December 2006. General Country Statistics: Mozambique. http://www.africancashew.com/countries/facts.php?id=30
- Artur, Luis and Nazneen Kanji. November 2005. "Satellites and Subsidies: Learning from Experience in Cashew Processing in Northern Mozambique." IIED. London. http://www.iied.org/NR/agbioliv/documents/SatellitesandSubsidiesKanji.pdf
- Carin Vijfhuizen, Luis Artur, Nazneen Kanji and Carla Braga. November 2003. "Liberalization, Gender, and Livelihoods: The Cashew Nut Case." Working Paper 2: Mozambique Phase 2- The South., January - December 2003.
- http://www.poptel.org.uk/iied/docs/sarl/WP2e_moz_phase2_cashew_eng.pdf
- CONFIDENTIAL TECHNOSERVE POWERPOINT
- "Enhancing the Global Competitiveness of Mozambique's Cashew Industry." TechnoServe, Inc. Washington, DC. http://www.technoserve.org/news/TNSCashewBrief.pdf
- Frey, Sharon. April 5, 2006. "Leading Firms in the Value Chain: The Development of a Globally Competitive Cashew Industry in Mozambique." Presentation given during the workshop, "Role of Lead Firms in the Value Chain: Benefits to Micro- and Small Enterprises." SEEP Network Annual Meeting. October 2005.

/files/3408_file_TNS_role_of_lead_firms_SEEP_preso_10_28_05_final.pdf

- Harilal, K.N., Nazneen Kanji, J. Jeyaranjan, Mridul Eapen, and Padmini Swaminathan. 2006. "Power in Global Value Chains: Implications for Employment and Livelihoods in the Cashew Nut Industry in India." http://www.iied.org/pubs/pdf/full/14514IIED.pdf
- January 31, 2005. "New Trade and Enterprise Development (NEW TREND) Program Final Performance Report." Technoserve, Inc. Washington, DC.
- Kanji, Nazneen. March 2004. "Corporate Responsibility and Women's Employment: The Cashew Nut Case." IIED. London. http://www.iied.org/pubs/pdf/full/16005IIED.pdf
- Kanji, Nazneen, Carin Vijfhuizen, Luis Artur and Carla Braga. March 2004. "Liberalization, Gender, and Livelihoods: The Mozambique Cashew Nut Case Summary Report." http://www.iied.org/pubs/pdf/full/9554IIED.pdf
- KIT, Faida MaLi, and IIRR. 2006. "Chain Empowerment: Supporting African Farmers to Develop Markets." Ch. 3: Chain Actors. Royal Tropical Institute, Amsterdam; Faida Market Link, Arusha; and International Institute of Rural Reconstruction, Nairobi.
- Labaste, Patrick, ed. 2005. "The European Horticulture Market: Opportunities for Sub-Saharan African Exporters." World Bank Working Paper No. 63. The World Bank. Washington, D.C.
- Londner, Steve. June 2005. "Expanding Economic Opportunities for Smallholder Farmers through Development of a Globally Competitive Cashew Industry in Mozambique." IAMA's USAID/World Bank Post-Conference Workshop. http://www.globalfoodchainpartnerships.org/chicago/presentations/SteveLondnerCashewMozam bique.pdf
- McMillan, Margaret, Karen Horn Welch, and Dani Rodrik. 2003. "When Economic Reform Goes Wrong: Cashew in Mozambique." Margaret McMillan, Tufts University; Karen Horn Welch, Stanford University; Dani Rodrik, Harvard University. Brookings Trade Forum, pp.97-165.
- Pearce, Douglas and Myka Reinsch. August 2005. "Small Farmers in Mozambique Access Credit and Markets by Forming Associations with Assistance from CLUSA CGAP Agricultural Microfinance." Case Study No. 5. http://www.cgap.org/portal/binary/com.epicentric.contentmanagement.servlet.ContentDeliverySe rvlet/Publications/html_pubs/AMCaseStudy_05.html
- Piper, Tim. January 2007. "Choosing Between Strategies: Adapting Industry Approaches to Specific Value Chain Analysis Using Three Comparative Commodities." Experiences in the banana, coffee and cashew value chains. Technoserve, Inc.. Washington, DC. http://www.technoserve.org/news/SEDWorkshop2007.pdf
- Southern African Regional Poverty Network (SARPN). September 2006. "Agricultural Intensification in Mozambique: Infrastructure, Policy and Institutional Framework When Do Problems Signal Opportunities?" www.sarpn.org.za/documents/d0002233/index.php
- "Strengthening Mozambique's Capacity for Agricultural Policy Analysis, Productivity, Growth and Poverty Reduction." Michigan State University Agricultural Economics Department. Lansing, Michigan. http://www.aec.msu.edu/fs2/mozambique/index.htm
- Swennenhuis, Joss. February 2003. "CASCA Support Programme for Cashew Processing Sub-Sector Case Study for BDS Reference Guide of SNV (Stichting Nederlandse Vrijwilligers)." SNV Mozambique
- TechnoServe, Inc. November, 2003. "Cashew in Eastern and Southern Africa" TechnoServe, Inc. for the East Africa Fine Coffee Association. Washington, DC.
- "USAID: Telling our Story: Simple Technology Revitalizes Mozambique." USAID. Washington, DC. http://www.usaid.gov/stories/mozambique/ss_mozambique_nuts.html
- Walter, Jake. October 2006. "Note from the Field: Quality Pricing for Cashews Pays Off." Technoserve, Inc. Washington, DC. http://www.microlinks.org/ev_en.php?ID=13537_201&ID2=DO_TOPIC
- Walter, Jake. October 19, 2006. "What Drives Competitiveness in the Mozambique Cashew Value Chain?" Power Point Presentation. Technoserve, Inc. Washington, DC.

- Wilson, John S. and Victor O. Abiola, eds. June 2003. "Standards and Global Trade: A Voice for Africa." World Bank. Washington, DC. http://www1.worldbank.org/publications/pdfs/15473frontmat.pdf
- Wilson, John S. August 4, 2003. "Standards and Global Trade: A Voice for Africa." World Bank. Washington, DC. http://siteresources.worldbank.org/INTRANETTRADE/Resources/Topics/StandardsTradeAfrica_ DFID SeminarAugust03.pdf \
- World Bank. May 2, 2006. "Project Brief on a Proposed Credit in the Amount of SDR 14.0 Million (US\$20.0 Million Equivalent) and Proposed Grant from the Global Environment Facility Trust Fund in the Amount of US\$6.2 Million to the Government of Mozambique for a Market-Led Smallholder Development in the Zambezi Valley Project." Environment, Rural and Social Development Unit, AFTS1 Country Department 2, Mozambique Africa Region. http://www.gefweb.org/documents/Council_Documents/GEF_C28/documents/288905-17-06Revised-MzBriefWPEntry.pdf

List of Sources Interviewed:

- Antonio Miranda, Miranda Caju
- Shakti Pal and Jake Walter, Technoserve Mozambique

Tool 6: Capturing Value Through Forward and Backward Integration

Box Case Study: Benefits of Vertical Integration – ZEGA and Zambia's Horticultural Value Chain

- Cassidy, Dermot. June 2006. "Annex 2: Working Paper on Review of Zambia's Current System of Phytosanitary Management." World Bank. http://siteresources.worldbank.org/INTRANETTRADE/Resources/Topics/Standards/Zambia_ann exes.pdf
- Sergeant, Andrew and Mirvat Dewadeh. April 2006. "Annex 6: Working Paper on Current Status of Zambia's Agricultural exports." World Bank. http://siteresources.worldbank.org/INTRANETTRADE/Resources/Topics/Standards/Zambia_ann exes.pdf
- Wijnands, J., "Sustainable Industrial Networks in the Flower Industry," International Society for Horticulture Science, Scripta Horticulturae, No. 2, http://www.actahort.org/chronica/pdf/sh_2.pdf

List of Sources Interviewed:

- Steve Humphreys, FinTrac
- C. Martin Webber, J.E. Austin Associates

Box Case Study: Bulgaria Wine – Integrating Operations to Secure Sourcing of Raw Material

• Vinzavod-Assenovgrad. http://www.mavrud.com/en/index.htm

List of Sources Interviewed:

- Borislav Georgiev, Independent Consultant
- Krassen Stanchev, IME

Case Study: Capturing Value through Integration – The Ghanaian Pineapple Industry and Blue Skies Holdings

• Blue Skies. http://www.bsholdings.com/

- Danielou, Morgane and Christophe Ravry. November 2005. "The Rise of Ghana's Pineapple Industry." World Bank, Africa Region Working Paper Series No. 93.
- Ghana Export Promotion Center News. http://www.gepcghana.com/news.php?item=4&n
- Hallam, Liu, Lavers, Pilkauskas, Rapsomanikis, and Claro. 2005. "The Market for Non-Traditional Agricultural Exports." Commodities and Trade Division, FAO.
- Imbert, Eric. March 2003. "The World Pineapple Market: When Growth Goes Hand in Hand with Diversity."Prepared for Food and Agricultural Organization (FAO) conference, Committee on Commodity Problems, Intergovernmental Group on Bananas and on Tropical Fruits, Third Session, Puerto de la Cruz, Spain, 11-15 December 2003.
- Rougé, B. and M. N'Goan. 1997. "L'Ananas en Afrique de l'Ouest et du Centre." Acta Horticulturae 425: 75-82.
- Seafreight Pineapple Exporters of Ghana (SPEG). http://www.ghana-exporter.org/speg/

List of Sources Interviewed:

- Nick Railston Brown, Technoserve Ghana
- Susan Bornstein, Technoserve, Washington DC
- Jean Michel Voisard, TIPCEE Ghana
- Emmanuel Owusu, GHANA SPEG

Tool 7: Creating and Taking Advantage of Economies of Scale

Case Study: Creating and Taking Advantage of Economies of Scale – The Ghana and Côte d'Ivoire Experiences

- Danielou, Morgane and Christophe Ravry. November 2005. "The Rise of Ghana's Pineapple Industry." World Bank, Africa Region Working Paper Series No. 93.
- Ghana Export Promotion Center News. http://www.gepcghana.com/news.php?item=4&n
- Hallam, Liu, Lavers, Pilkauskas, Rapsomanikis, and Claro. 2005. "The Market for Non-Traditional Agricultural Exports." Commodities and Trade Division, FAO.
- Imbert, Eric. March 2003. "The World Pineapple Market: When Growth Goes Hand in Hand with Diversity."Prepared for Food and Agricultural Organization (FAO) conference, Committee on Commodity Problems, Intergovernmental Group on Bananas and on Tropical Fruits, Third Session, Puerto de la Cruz, Spain, 11-15 December 2003.
- Rougé, B. and M. N'Goan. 1997. "L'Anana en Afrique de l'Ouest et du Centre." Acta Horticulturae 425: 75-82.
- Seafreight Pineapple Exporters of Ghana (SPEG). http://www.ghana-exporter.org/speg/

List of Sources Interviewed:

- Nick Railston Brown, Technoserve Ghana
- Susan Bornstein, Technoserve, Washington DC
- Jean Michel Voisard, TIPCEE Ghana
- Emmanuel Owusu, GHANA SPEG

Case Study: Creating and Taking Advantage of Economies of Scale Within the Mozambican Cashew Value Chain

• AfricanCashew.com. 20 December 2006. General Country Statistics: Mozambique. http://www.africancashew.com/countries/facts.php?id=30

- Artur, Luis and Nazneen Kanji. November 2005. "Satellites and Subsidies: Learning from Experience in Cashew Processing in Northern Mozambique." IIED. London. http://www.iied.org/NR/agbioliv/documents/SatellitesandSubsidiesKanji.pdf
- Carin Vijfhuizen, Luis Artur, Nazneen Kanji and Carla Braga. November 2003. "Liberalization, Gender, and Livelihoods: The Cashew Nut Case." Working Paper 2: Mozambique Phase 2- The South., January December 2003.

http://www.poptel.org.uk/iied/docs/sarl/WP2e_moz_phase2_cashew_eng.pdf

- CONFIDENTIAL TECHNOSERVE POWERPOINT
- "Enhancing the Global Competitiveness of Mozambique's Cashew Industry." Technoserve, Inc. Washington, DC. http://www.technoserve.org/news/TNSCashewBrief.pdf
- Frey, Sharon. April 5, 2006. "Leading Firms in the Value Chain: The Development of a Globally Competitive Cashew Industry in Mozambique." Presentation given during the workshop, "Role of Lead Firms in the Value Chain: Benefits to Micro- and Small Enterprises." SEEP Network Annual Meeting. October 2005.

 $/files/3408_file_TNS_role_of_lead_firms_SEEP_preso_10_28_05_final.pdf$

- Harilal, K.N., Nazneen Kanji, J. Jeyaranjan, Mridul Eapen, and Padmini Swaminathan. 2006. "Power in Global Value Chains: Implications for Employment and Livelihoods in the Cashew Nut Industry in India." http://www.iied.org/pubs/pdf/full/14514IIED.pdf
- Kanji, Nazneen. March 2004. "Corporate Responsibility and Women's Employment: The Cashew Nut Case." IIED. London. http://www.iied.org/pubs/pdf/full/16005IIED.pdf
- Kanji, Nazneen, Carin Vijfhuizen, Luis Artur and Carla Braga. March 2004. "Liberalization, Gender, and Livelihoods: The Mozambique Cashew Nut Case Summary Report." http://www.iied.org/pubs/pdf/full/9554IIED.pdf
- KIT, Faida MaLi, and IIRR. 2006. "Chain Empowerment: Supporting African Farmers to Develop Markets." Ch. 3: Chain Actors. Royal Tropical Institute, Amsterdam; Faida Market Link, Arusha; and International Institute of Rural Reconstruction, Nairobi.
- Labaste, Patrick, ed. 2005. "The European Horticulture Market: Opportunities for Sub-Saharan African Exporters." World Bank Working Paper No. 63. The World Bank. Washington, D.C.
- Londner, Steve. June 2005. "Expanding Economic Opportunities for Smallholder Farmers through Development of a Globally Competitive Cashew Industry in Mozambique." IAMA's USAID/World Bank Post-Conference Workshop. http://www.globalfoodchainpartnerships.org/chicago/presentations/SteveLondnerCashewMozam bique.pdf
- McMillan, Margaret, Karen Horn Welch, and Dani Rodrik. 2003. "When Economic Reform Goes Wrong: Cashew in Mozambique." Margaret McMillan, Tufts University; Karen Horn Welch, Stanford University; Dani Rodrik, Harvard University. Brookings Trade Forum, pp.97-165.
- Pearce, Douglas and Myka Reinsch. August 2005. "Small Farmers in Mozambique Access Credit and Markets by Forming Associations with Assistance from CLUSA CGAP Agricultural Microfinance." Case Study No. 5. http://www.cgap.org/portal/binary/com.epicentric.contentmanagement.servlet.ContentDeliverySe rvlet/Publications/html pubs/AMCaseStudy 05.html
- Piper, Tim. January 2007. "Choosing Between Strategies: Adapting Industry Approaches to Specific Value Chain Analysis Using Three Comparative Commodities." Experiences in the banana, coffee and cashew value chains. Technoserve, Inc.. Washington, DC. http://www.technoserve.org/news/SEDWorkshop2007.pdf
- Southern African Regional Poverty Network (SARPN). September 2006. "Agricultural Intensification in Mozambique: Infrastructure, Policy and Institutional Framework When Do Problems Signal Opportunities?" www.sarpn.org.za/documents/d0002233/index.php

- "Strengthening Mozambique's Capacity for Agricultural Policy Analysis, Productivity, Growth and Poverty Reduction." Michigan State University Agricultural Economics Department. Lansing, Michigan. http://www.aec.msu.edu/fs2/mozambique/index.htm
- Swennenhuis, Joss. February 2003. "CASCA Support Programme for Cashew Processing Sub-Sector Case Study for BDS Reference Guide of SNV (Stichting Nederlandse Vrijwilligers)." SNV Mozambique.
- TechnoServe, Inc. November, 2003. "Cashew in Eastern and Southern Africa" TechnoServe, Inc. for the East Africa Fine Coffee Association. Washington, DC.
- Technoserve, Inc. January 31, 2005. "New Trade and Enterprise Development (NEW TREND) Program Final Performance Report." Washington, DC.
- "USAID: Telling our Story: Simple Technology Revitalizes Mozambique." USAID. Washington, DC. http://www.usaid.gov/stories/mozambique/ss_mozambique_nuts.html
- Walter, Jake. October 2006. "Note from the Field: Quality Pricing for Cashews Pays Off." Technoserve, Inc. Washington, DC.
- http://www.microlinks.org/ev_en.php?ID=13537_201&ID2=DO_TOPIC
- Walter, Jake. October 19, 2006. "What Drives Competitiveness in the Mozambique Cashew Value Chain?" Power Point Presentation. Technoserve, Inc. Washington, DC.
- Wilson, John S. and Victor O. Abiola, eds. June 2003. "Standards and Global Trade: A Voice for Africa." World Bank. Washington, DC. http://www1.worldbank.org/publications/pdfs/15473frontmat.pdf
- Wilson, John S. August 4, 2003. "Standards and Global Trade: A Voice for Africa." World Bank. Washington, DC.

http://siteresources.worldbank.org/INTRANETTRADE/Resources/Topics/StandardsTradeAfrica_DFID_SeminarAugust03.pdf

 World Bank. May 2, 2006. "Project Brief on a Proposed Credit in the Amount of SDR 14.0 Million (US\$20.0 Million Equivalent) and Proposed Grant from the Global Environment Facility Trust Fund in the Amount of US\$6.2 Million to the Government of Mozambique for a Market-Led Smallholder Development in the Zambezi Valley Project." Environment, Rural and Social Development Unit, AFTS1 Country Department 2, Mozambique Africa Region. http://www.gefweb.org/documents/Council_Documents/GEF_C28/documents/288905-17-06Revised-MzBriefWPEntry.pdf

List of Sources Interviewed:

- Antonio Miranda, Miranda Caju
- Shakti Pal and Jake Walter, Technoserve Mozambique

Tool 8: Positioning Products and Value Chains for Greater Value and Competitiveness

Box Case Study: Ecuadorian Cacao—Positioning the Value Chain for Greater Value and Competitiveness

- ARD. "Ecuador Northern Border Income and Employment Project." http://www.ardinc.com/projects/detail_region.php?id=118
- InfoComm: Market Information in the Commodities Area. "Cocoa: Quality." UNCTAD. http://www.unctad.org/infocomm/anglais/cocoa/quality.htm

List of Sources Interviewed:

• Karl Walk, Purchaser, Blommer's Chocolate.

Box Case Study: Thailand GAP Cluster – Positioning Products (and the Value Chain) for Greater Value and Competitiveness

• Arocha, Marcos. Monitoring & Evaluation Report of the USAID-funded Southeast Asia Competitiveness Initiative (SEACI). J.E. Austin Associates, 2004.

List of Sources Interviewed:

• Anan Patannathanes, KIAsia

Case Study: Value Chain Strategies for Market Repositioning -- Rwandan Coffee

- Associated Press. March 3, 2006. "Rwandan Coffee in Your Cup: Starbucks Introduces Product in N. America." Montreal Gazette.
- Bureau pour le Developpement de la Production Agricole (BDPA). September 2005. "Analyse Economique et Financière de la Filière Café." France.
- Chemonics International, Inc. 2001. "Assistance a la Dynamisation de l'Agribusiness au Rwanda: ADAR, Year 1 Work Plan." Washington, DC.
- Chemonics International, Inc. June 2002. "Considerations for Initiating a Specialty Coffee Industry in Rwanda." Washington, DC.
- Chemonics International, Inc. July 2004. "ADAR Agribusiness Development Assistance 2004: 2nd Qtr Progress Report, April 1 to June 30, 2004." Washington, DC.
- Chemonics International, Inc. December 2005. "Assessing USAID's Investments in Rwanda's Coffee Sector: Best Practices and Lessons Learned to Consolidate Results and Expand Impact." Washington, DC.
- J.E. Austin Associates. May 2006. "Rwanda Export Promotion Action Plan." Submitted to the Government of Rwanda: Rwanda Investment and Export Promotion Agency and Ministry of Finance and Economic Planning. With assistance from the World Bank.
- Lecraw, Donald. January 31, 2005. "The Coffee, Tea, Tourism, and Handicrafts Sectors: Export Potential and Contribution to Growth and Poverty Reduction, draft version." Integrated Framework, Diagnostic Trade Integration Study.
- Office des Cafés du Rwanda (OCIR CAFÉ). November 2005. "New Action Plan 2006-2008, For the Development of the Rwanda Coffee Sector."
- On the Frontier. July 2002. "Rwanda Coffee Strategy and Action Priorities, Rwanda Competitiveness and Innovation Program." Work Group on Coffee. Presentation.
- On the Frontier. April 2004. "The Rwandan National Competitiveness and Innovation Project: A National Transformation Process." Washington, DC.

List of Sources Interviewed:

- Rob Henning, on the Frontier, Washington DC
- C. Martin Webber, J.E. Austin Associates
- Patrick Nugawela, J.E. Austin Associates
- Warren Weinstein, J.E. Austin Associates
- Maurice Wiener, Chemonics, Inc., former Chief of Party, USAID/ADAR Project

Tool 9: Applying Standards and Certifications to Achieve Greater Quality

Box Case Study: Thailand GAP Cluster—Use of Standards and Certifications to Upgrade Value

• Arocha, Marcos. Monitoring & Evaluation Report of the USAID-funded Southeast Asia Competitiveness Initiative (SEACI). J.E. Austin Associates, 2004.

List of Sources Interviewed:

Anan Patannathanes, KIAsia

Box Case Study: Ecuador Cacao—Improving Quality at the Producer Level to Achieve Higher Market Prices

- ARD. "Ecuador Northern Border Income and Employment Project." http://www.ardinc.com/projects/detail_region.php?id=118
- InfoComm: Market Information in the Commodities Area. "Cocoa: Quality." UNCTAD. http://www.unctad.org/infocomm/anglais/cocoa/quality.htm

List of Sources Interviewed:

• Karl Walk, Purchaser, Blommer's Chocolate.

Case Study: Uganda Nile Perch Quality Management and Certification

- Ponte, Stefano, "Bans, Tests and Alchemy: Food Safety Standards and the Ugandan Fish Export Industry," Danish Institute for International Studies, DIIS Working Paper no. 2005/19, http://www.isn.ethz.ch/pubs/ph/details.cfm?lng=en&id=18423
- Ponte, Stefano. 2005. "Bans, Tests, and Alchemy: Food Safety Standards and the Ugandan Fish Export Industry." Danish Institute for International Studies.
- World Bank. January 2006. "Uganda, Standards and Trade: Experience, Capacities, and Priorities," http://siteresources.worldbank.org/INTRANETTRADE/Resources/Topics/Standards/Uganda_Sta ndards_final.pdf

List of Sources Interviewed:

- Patrick Nugawela, J.E. Austin Associates
- Philip Borel de Bitche, GreenFields Uganda LTD
- C. Martin Webber, J.E. Austin Associates

Tool 10: Identifying Needed Support Services for the Value Chain

Box Case Study: Uganda Cotton—Enterprise-Linked Extension Services Model

• Support for Private Enterprise Expansion and Development (SPEED). June 2003. "Support for Private Enterprise Expansion and Development, Private Sector Takes on Development Role." SPEED Bulletin.

Box Case Study: Sri Lankan Cinnamon

- Bolnick, Bruce. November 2003. "The Economic Impact of Cluster Initiatives Under the Competitiveness Initiative Project Interim Assessment and Recommendations," Nathan Associates Inc.
- "A Competitiveness Strategy for Sri Lanka's Spices Industry," Developed by The Spices Industry Cluster, supported and funded by The Competitiveness Initiative, a joint project of the United States Agency for International Development (USAID), Nathan Associates Inc., and J.E. Austin Associates. Colombo, Sri Lanka June 2002.
- "Helping restore the legacy of 'Ceylon cinnamon." July 2005. USAID. http://www.usaid.gov/lk/news/tos/july05-cinnamon.html.
- Iqbal, Mohammad. "International trade in non-wood forest products: An overview", Food and Agriculture Organization (FAO), Rome, Italy 1993 available online at: http://www.fao.org/docrep/x5326e/x5326e00.htm#Contents.
- USAID. "Request for Proposal: Connecting Rural Economies (CORE) for Sri Lanka," 2007.

Case Study: Identifying Needed Support Services for the Value Chain—Zambia Cotton

- Food Security Research Project. March 26, 2000. "Improving Smallholder and Agri-Business Opportunities in Zambia's Cotton Sector: Key Challenges and Options." Lusaka, Zambia.
- Regional Agricultural Trade Expansion Support (RATES) Program. August 2003. "Cotton-Textile-Apparel Value Chain Report: Zambia" USAID.
- Tschirley, David, Ballard Zulu, and James Shaffer. February 2004. "Cotton in Zambia: An Assessment of Its Organization, Performance, Current Policy Initiatives, and Challenges for the Future." Working Paper No. 10. Food Security Research Project. Lusaka, Zambia.
- Tschirley, David, Colin Poulton, and Duncan Boughton. June 2006. "The Many Paths of Cotton Sector Reform in Eastern and Southern Africa: Lessons from a Decade of Experience." Working Paper No. 18. Food Security Research Project. Lusaka, Zambia.

List of Sources Interviewed:

• Ben Sekamatte, Dunavant Cotton, Zambia

Tool 11: Improving the Operating Environment by Promoting Public/Private Dialogue

Box Case Study: Tanzanian Coffee and KILICAFE: Productive Public/Private Dialogue

• TechnoServe, Inc. May 29, 2006. Case Study on Coffee in Tanzania. Washington, DC.

List of Sources Interviewed:

- Thom Dixon, TechnoServe Tanzania
- Susan Bornstein, TechnoServe

Case Study: Improving the Operating Environment by Promoting Public/Private Dialogue-Botswana Cattle Producers Association

- "Advocating for Cattle Policy Reform." http://www.satradehub.org/CXA_html/ss_case_study.html
- "Cattle Farmers Get Higher Prices." USAID. http://www.usaid.gov/stories/botswana/ss_bw_meat.pdf
- Food Chain Center, "Cutting Costs Adding Value in Red Meat." http://www.foodchaincentre.com/cir.asp?type=1&subtype=6&cir=117
 - o "f) Concept What does Value in Red Meat mean to Consumers?" September 2003.
 - o "g) Concept Where to find Savings in the Red Meat Chain?" September 2003.
- Jefferis, Keith. "How Trade Can Help to Revitalize Botswana's Beef and Cattle Sector." http://www.satradehub.org/CXA_html/docs/reports/How%20Trade%20Can%20Help%20to%20 Revitalise%20Botswana%20Final.pdf
- Monnane, Monnane M., Research Fellow. "Experience on Competition Policy: The Case of Botswana." Botswana Institute for Development Policy Analysis.
- "Presidential Initiatives 2005: Global Action, Focused Results." http://pdf.usaid.gov/pdf_docs/PDACG999.pdf
- "Trade Hub Happenings." Issue No. 45. http://www.satradehub.org/CXA_html/docs/HH-Dec2006%20-%20Jan2007.pdf / http://www.satradehub.org/CXA_html/hh_issue45.html

List of Sources Interviewed:

• Keith Jefferis, Southern Africa Global Competitiveness Hub Consultant

- Martin Norman, Hub Manager, Southern Africa Global Competitiveness Hub
- Philip Fischer, President, Botswana Cattle Producer's Association

Tool 12: Achieving Synergies Through Clustering

• Porter, Michael E. "Creating a Competitive South Africa." Institute for Strategy and Competitiveness, Harvard Business School. Johannesburg, South Africa. 3 July 2007. Graphic Sources: Harvard Business School student team research (2007) - Kusi Hornberger, Nick Ndiritu, Lalo Ponce-Brito, Melesse Tashu, Tijan Watt.

Case Study: Achieving Synergies Through Clustering--Kenya Avocados

- "Development of the Kenyan Avocado Value Chain." Kenya BDS Activity Brief. Emerging Markets Group. Arlington, VA.
- Emerging Markets Group. November/December 2006. "Marketing News: Kenya Horticulture Development Program." Kenya BDS. Arlington, VA.
- Hallam, David and Pascal Liu. 2004. "The Market of Non-Traditional Agricultural Exports." FAO.
- Jaffee, Steven. 2003. "From Challenge to Opportunity: Transforming Kenya's Fresh Vegetable Trade in the Context of Emerging Food Safety and Other Standards in Europe." Agriculture and Rural Development Discussion Paper. World Bank. Washington, DC.
- Minot, N & M. Ngigi. December 1-3, 2003. "Are Horticultural Exports a Replicable Success Story? Evidence from Kenya and Côte d'Ivoire." International Food Policy Research Institute and Egerton University. Paper presented at the InWEnt, IFPRI, NEPAD, CTA Conference "Successes in African Agriculture." Pretoria.
- Okado, Mark. 1999. "Background Paper on Kenya Off-season and Specialty Fresh Vegetables and Fruits: Lessons of Experience from the Kenya Horticulture Industry." UN Conference on Trade and Development.
- Shah, Tiku. November 2003. "Kenya Avocados." Presentation on the market. Fresh Produce Exporters Association of Kenya (FPEAK).

List of Sources Interviewed:

• David Knopp, Chief of Party, USAID/BDS Project

Tool 13: Monitoring Achievements in Value Chain Performance

Box Case Study: Standards for Ghanaian Pineapples

- Danielou, Morgane and Christophe Ravry. November 2005. "The Rise of Ghana's Pineapple Industry: From Successful Takeoff to Sustainable Expansion." World Bank. Washington, DC.
- "Ghana: European Tastes Affect Country's Pineapple Exports." All Africa. http://allafrica.com/stories/printable/200612110733.html
- Ghana News Agency. 11 October 2004. "GEPC orders pineapple plantlets." http://www.ghanaweb.com/GhanaHomePage/NewsArchive/artikel.php?ID=67530
- Yawson, David and Luis Kluwe Aguiar. December 6-10, 2006. "Agility in the Ghanaian International Pineapple Supply Chain." Presentation to the International Symposium on Fresh Produce Supply Chain Management. Chiang Mai, Thailand.

List of Sources Interviewed:

• Malick Antoine, ESSD Africa, World Bank

Box Case Study: The PAID M&E Framework

Adapted from model developed by Kevin X. Murphy, J.E. Austin

Acronyms

AAK	Agricultural Association of Kenya
ADAR	Agribusiness Development Activity in Rwanda
AEPP	Agricultural Export Promotion Project (Senegal)
AGOA	African Growth and Opportunity Act
AgSSIP	Agricultural Services Subsector Investment Project
AĬA	Agro Indistria Associadas
AMAP/BDS	Accelerated Microenterprise Advancement Project/Business Development Service
APEP	Agricultural Productivity Enhancement Program (USAID)
APROFA	Agence pour la Promotion des Filières Agricoles (Mali)
ASTM	American Society for Testing and Materials
AUTM	Association of University Technology Managers
BCPA	Botswana Cattle Producers Association
BDS	Business Development Services (Kenya)
BMC	Botswana Meat Commission
BMO	Business Membership Organization
BMUs	Beach Management Units (Uganda)
CFA	Communauté française d'Afrique
CI	Clustering Initiative
CRTA	Colombo Rubber Traders Association
CTBI	Coffee Taxation & Benchmarking Initiative (Tanzania)
DFID	Department for International Development (UK)
DFR	Department of Fisheries (Uganda)
DPL	Dipped Products Ltd. (Sri Lanka)
DRC	Domestic Resource Cost
EAGA	East African Growers (Kenya)
EFEG	Exotic Fruit Exporters Association of Ghana
ESSD Africa	Environmentally and Socially Sustainable Development Department Africa (World
	Bank)
EU	European Union
EurepGAP	Euro-Retailer Produce Working Group Good Agricultural Practices; a private-sector
	body that sets voluntary standards for certifying agricultural products around the
	globe
FAO	Food and Agriculture Organization of the United Nations
FAOSTAT	FAO Statistical Database
FCFA	CFA franc
FDI	Foreign Direct Investment
FOB	Freight on Board
FPEAK	Fresh Produce Exporters Association of Kenya
GAP	Good Agricultural Practices
GDP	Gross Domestic Product
GMP	Good Manufacturing Practice (Uganda)
GoM	Government of Mozambique
GoR	Government of Rwanda
GoS	Government of Senegal
GoSL	Government of Sri Lanka
GoU	Government of Uganda
GPSCA	Gabinete de Promoção do Sector Agráno
GTZ	Gesellschaft für Technische Zusammenarbeit

HAG	Horticultural Association Ghana
HACCP	Hazard Analysis and Critical Control Point
HCDA	Horticultural Crop Development Authority (Kenya)
HCDA	Horticulture Crops Development Agency
HS	Harmonized System
IA	Impact Assessment
IDA	International Development Association (World Bank)
IFC	International Finance Corporation
IFLEX	Information sur les Fruits et Légumes d'Exportation du Sénégal
KARI/NHRC	Kenya Agricultural Research Institute/National Horticultural Research Centre
KEPHIS	Kenya Plant Health Inspectorate Services
KILICAFE	Association of Kilimanjaro Specialty Coffee Growers (Tanzania)
LIFFE	London International Financial Futures and Options Exchange
LINTCO	Lint Company of Zambia
M&E	Monitoring and Evaluation
MCI	Mongolian Competitiveness Initiative
MDG	Millennium Development Goals
MPI	Ministry of Plantation Industries
MPS-ABC	A Good Agricultural Practices certification under EurepGAP
MRDP	Moneragala Rubber Development Program
n.e.s.	Not elsewhere specified
NCC	National Competitiveness Council (Ireland)
NPV	Net Present Value
NTAE	Non-Traditional Agricultural Export
OCAB	Office Centrale des Producteurs-Exportateurs d'Ananas et de Bananes (Côte
	d'Ivoire)
OCIR-CAFÉ	Office Nationale de Cultures Industrielles-Café
PA	Planters Association of Ceylon
PAID	Process indicators, Action indicators, Investment indicators, Delivered results
PCPB	Pest Control Products Board (Kenya)
PE	Performance Evaluation
PEARL	Partnership to Enhance Agriculture in Rwanda Through Linkages
PISDAC	Pakistan Initiative for Strategic Development and Competitiveness
PM	Project Monitoring
PoP	Point of Purchase
PPD	Public/Private Dialogue
PSD	Private Sector Development
PRI	Plastics & Rubber Institute of Sri Lanka
R&D	Research and Development
RCA	Revealed Cost Advantage
RDD	Rubber Development Department
ROI	Return on Investment
RRI	Rubber Research Institute
SADC	South African Development Community
SAGCH	Southern Africa Global Competitiveness Hub
SLAMERP	Sri Lanka Association of Manufacturers & Exporters of Rubber Products

Acronyms (continued)

SMEs	Small and Medium Enterprises
SPEG	Seafreight Pineapple Exporters of Ghana
SRI	Sri Lanka Society of Rubber Industry
SSA	Sub-Saharan Africa
SWOG	Strategic Working Group
SWOT	Strengths/Weaknesses/Opportunities/Threats
TCI	The Competitiveness Initiative (Sri Lanka)
TIP	Trade and Investment Program
TSC	Sri Lankan Spice Council
TZS	Tanzanian Shilling
UBA	Union Bananière Africaine
UFEA	Uganda Flower Exporters Association
UFPEA	Uganda Fish Processors and Exporters Association
UNBA	Uganda National Bureau of Standards
USAID	U.S. Agency for International Development
USDA	United States Department of Agriculture
VA	Vinzavod-Assenovgrad
VC	Value Chain
WCO	World Customs Organization
XOF	Franc CFA (West African currency)
ZEGA	Zambian Export Growers Association
ZPA	Zambia Privatization Agency

List of Text Boxes (in Order of Appearance)

- Box 1: How to Choose Value Chains for Intervention: The Example of Senegal's Projet Croissance
- Box 2: Uganda: Benchmarking Constraints in the Coffee Industry
- Box 3: Tanzania Cotton—Benchmarking Costs
- Box 4: Mongolian Meat Industry—Upgrading the Value Chain
- Box 5: Deepening the Value Chain: Glass Jar Production in Armenia
- Box 6: Replicable Business Models—Rwanda Coffee Washing Stations
- Box 7: Dairy Pakistan—Identifying and Replicating Business Models within the Value Chain
- Box 8: Benefits of Vertical Integration—ZEGA and Zambia's Horticultural Value Chain
- Box 9: Bulgaria Wine—Integrating Operations to Secure Sourcing of Raw Material
- Box 10: Ecuadorian Cacao—Positioning the Value Chain for Greater Value and Competitiveness
- Box 11: Thailand GAP Cluster—Positioning Products (and the Value Chain) for Greater Value and Competitiveness
- Box 12: Thailand GAP Cluster—Use of Standards and Certifications to Upgrade Value
- Box 13: Ecuador Cacao—Improving Quality at the Producer Level to Achieve Higher Market Prices
- Box 14: Uganda Cotton—Enterprise-Linked Extension Services Model
- Box 15: Sri Lankan Cinnamon
- Box 16: Tanzanian Coffee and KILICAFE: Productive Public/Private Dialogue
- Box 17: Standards for Ghanaian Pineapples
- Box 18: The PAID M&E Framework

List of Cases (in Order of Appearance)

Case Study:	Prioritizing Value Chains by Using Comparative Analysis-Value Chain Selection in	
	Mozambique	
Case Study:	Understanding the Value Chain and Integrating Information Into Strategy: Nigerian	
	Domestic Catfish	
Case Study:	Uganda Floriculture: Benchmarking and Gap Analysis	
Case Study:	Kenyan Green Beans and Other Fresh Vegetable Exports	
Case Study:	Identifying and Implementing Replicable Business Models—Mozambique Cashews	
Case Study:	Capturing Value Through Integration—The Ghanaian Pineapple Industry and Blue Skies	
	Holdings	
Case Study:	Creating and Taking Advantage of Economies of Scale-The Ghana and Côte d'Ivoire	
	Experiences	
Case Study:	Creating and Taking Advantage of Economies of Scale Within the Mozambican Cashew	
	Value Chain	
Case Study:	Value Chain Strategies for Market Repositioning—Rwandan Coffee	
Case Study:	Uganda Nile Perch Quality Management and Certification	
Case Study:	How to Encourage Innovation—Mali Mangos	
Case Study:	Identifying Needed Support Services for the Value Chain-Zambia	
Case Study:	Improving the Operating Environment Through Public/Private Dialogue-Botswana	
	Cattle Producers Association	

List of Tables (in Order of Appearance)

- Table 1.African Trade Growth
- Table 2. SSA's Rank Among World Importing and Exporting Regions
- Table 3.Historical Levels of Agriculture Productivity, by Region
- Table 4.Historical Trends of African Productivity, by Country
- Table 5.Key Documents Utilized in the Literature Review
- Table 6.Mozambique Commodities by Price, 2005
- Table 7.Original and Revised Sectors for Intervention in Mozambique
- Table 8.Gap Analysis of the Dominican Cigar Industry vs. Cuban Cigars
- Table 9.Illustrative Gap Analysis
- Table 10.Growth Performance of Ugandan Horticulture Exports, 1995–2002
- Table 11.Rose Sales at Dutch Auction in 2002
- Table 12.Sweetheart Rose Production Cost Structure per Hectare: Uganda vs. Kenya (in USD)
- Table 13.Cost Structure for Cuttings Production per Hectare: Uganda vs. Kenya in USD
- Table 14. Driving Forces and a Comparison between Uganda, Kenya, and the Netherlands
- Table 15.Mozambican Cashew Processing Operations
- Table 16.Some Reasons to Consider Vertical Integration
- Table 17. Cost of Packing Material Before and After AIA
- Table 18.Cost of Shipping Before and After AIA
- Table 19.Rwandan Coffee Production
- Table 20.Recent Rwandan Coffee Production and Prices
- Table 21.Standard and Ordinary Coffee Percentages, 1998–2005
- Table 22.Dunavant Compensation Plan
- Table 23. Sample Checklist of Issues to Address at Various Stages of the PPD Process
- Table 24.
 Types of Business Membership Organizations and Functions
- Table 25.Pursuing Effective Dialogue

List of Figures (in Order of Appearance)

Figure 1	Africa's Share of World Trade	
Figure 2	Porter's Competitiveness Diamond	
Figure 3	Power Relations in Value Chains	
Figure 4	Value Chain Program Implementation Cycle and Tool Themes	
Figure 5	Mozambique Cashew Domestic Value Chain	
Figure 6	Enhancing Productivity and Value Across the Value Chain	
Figure 7	Mongolia: Net Revenue per KG of Cashmere for Each Component of the Value System	
Figure 8	Pakistan Dairy SWOT Analysis	
Figure 9	Competitiveness Diamond Analysis—Key Questions	
Figure 10	Map of Nigeria	
Figure 11	Nigerian Domestic Catfish Farming Value Chain	
Figure 12	Consumers' Reasons for Purchasing Live Catfish	
Figure 13	Consumers' Views of What Factors Would Increase Their Purchases of Fresh Catfish	
Figure 14	Nigerian Domestic Catfish Farming Value Chain—Possible Actions	
Figure 15	Projected Development of the Value Chain for Fresh Catfish	
Figure 16	Morocco-Coordination Within the Citrus Fruit and Tomato Value Chains, Comparison	
C	with Turkey and Spain	
Figure 17	Map of Uganda	
Figure 18	Uganda Floriculture Value Chain and Cluster Map	
Figure 19	Value Chain Analysis for Flowers from Uganda	
Figure 20	Components of Total Cost of Sweetheart Roses in Uganda and Kenya	
Figure 21	Components of Total Cost of Cuttings in Uganda and Kenya	
Figure 22	Uganda's Flower Exports 1994–2006	
Figure 23	Mongolian Meat Export Value Chains	
Figure 24	Deepening the Value Chain Under Two Scenarios	
Figure 25	Map of Kenya	
Figure 26	Kenya Green Bean Value Chain	
Figure 27	Integrated Large Exports Value Chain	
Figure 28	Mozambican Cashew Nut Exports Since the 1970s	
Figure 29	Weaknesses in the Mozambique Cashew Domestic Value Chain	
Figure 30	Mozambique Cashew Domestic Value Chain with Small Processors	
Figure 31	Firm Value Chain	
Figure 32	Improving Value Chains (Before Value Added)	
Figure 33	Improving Value Chains (with Value Added)	
Figure 34	ESA Rose Exports to the European Union	
Figure 35	Bulgarian Wine Value Chain: Traditional	
Figure 36	Bulgarian Wine Value Chain: Integrated	
Figure 37	Pineapple Exports—Ghana, Côte d'Ivoire, and Costa Rica	
Figure 38	European Pineapple Imports	
Figure 39	Ghana's Fresh Pineapple Exports	
Figure 40	Ghana Pineapple Value Chain	
Figure 41	Ghana European Pineapple Exports	
Figure 42	Regions of Mozambique	
Figure 43	Mozambican Cashew Exports, 1961–2000	
Figure 44	Domestic Value Chain for Mozambican Cashews	
Figure 45	Product Positioning—Mongolian Cashmere Industry	
Figure 46	Positioning of Rwandan Coffee in the 1990s vs. World Markets	

Figure 47Rwandan Coffee's Goals for 2010Figure 48Generic Coffee Value ChainFigure 49Results of Rwandan Coffee's Positioning Efforts, 2005Figure 50Rwandan Coffee Positioning Goals Post 2010Figure 51Standards Plotted Against Product ValueFigure 52International and Value-Added StandardsFigure 53Thai Vegetable Export Value ChainsFigure 54Ecuadorian Cacao Value ChainFigure 55Prices and Production of CacaoFigure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	F :	$\mathbf{D}_{\mathrm{max}} = \mathbf{D}_{\mathrm{max}} \mathbf{C}_{\mathrm{max}} \mathbf{f}_{\mathrm{max}} \mathbf{C}_{\mathrm{max}} 1_{\mathrm{max}} \mathbf{f}_{\mathrm{max}} 2010$
Figure 43Generic Conee Value ChainFigure 49Results of Rwandan Coffee's Positioning Efforts, 2005Figure 50Rwandan Coffee Positioning Goals Post 2010Figure 51Standards Plotted Against Product ValueFigure 52International and Value-Added StandardsFigure 53Thai Vegetable Export Value ChainsFigure 54Ecuadorian Cacao Value ChainFigure 55Prices and Production of CacaoFigure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 47	Rwandan Coffee S Goals for 2010
Figure 49Results of Rwandan Coffee S Positioning Enorts, 2003Figure 50Rwandan Coffee Positioning Goals Post 2010Figure 51Standards Plotted Against Product ValueFigure 52International and Value-Added StandardsFigure 53Thai Vegetable Export Value ChainsFigure 54Ecuadorian Cacao Value ChainFigure 55Prices and Production of CacaoFigure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 48	Desults of Dwondon Coffac's Desitioning Efforts 2005
Figure 50Kwandah Conee Positioning Goals Post 2010Figure 51Standards Plotted Against Product ValueFigure 52International and Value-Added StandardsFigure 53Thai Vegetable Export Value ChainsFigure 54Ecuadorian Cacao Value ChainFigure 55Prices and Production of CacaoFigure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 49	Results of Rwandan Coffee S Positioning Efforts, 2005
Figure 51Standards Flotted Against Product ValueFigure 52International and Value-Added StandardsFigure 53Thai Vegetable Export Value ChainsFigure 54Ecuadorian Cacao Value ChainFigure 55Prices and Production of CacaoFigure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 50	Stendarde Dietted Against Dreduct Value
Figure 52International and Value-Added StandardsFigure 53Thai Vegetable Export Value ChainsFigure 54Ecuadorian Cacao Value ChainFigure 55Prices and Production of CacaoFigure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 51	Standards Plotted Against Product Value
Figure 53That Vegetable Export Value ChainsFigure 54Ecuadorian Cacao Value ChainFigure 55Prices and Production of CacaoFigure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 52	International and Value-Added Standards
Figure 54Ecuadorian Cacao Value ChainFigure 55Prices and Production of CacaoFigure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 55	That vegetable Export value Chains
Figure 55Prices and Production of CacaoFigure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 54	Ecuadorian Cacao Value Chain
Figure 56World Production of Cacao BeansFigure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 55	Prices and Production of Cacao
Figure 57Ugandan Nile Perch Value ChainFigure 58ISO 9000 Certification ProcessFigure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 56	World Production of Cacao Beans
Figure 58ISO 9000 Certification ProcessFigure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 57	Ugandan Nile Perch Value Chain
Figure 59Good Manufacturing Practice & HACCPFigure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 58	ISO 9000 Certification Process
Figure 60HAACP implementationFigure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 59	Good Manufacturing Practice & HACCP
Figure 61Uganda's Nile Perch ExportsFigure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 60	HAACP implementation
Figure 62Map of MaliFigure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 61	Uganda's Nile Perch Exports
Figure 63Air-Freight Value Chain for Mali's MangosFigure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 62	Map of Mali
Figure 64Sea-Freight Value Chain for Mali's MangosFigure 65Multi-Modal Value Chain for Mali's Mangos	Figure 63	Air-Freight Value Chain for Mali's Mangos
Figure 65 Multi-Modal Value Chain for Mali's Mangos	Figure 64	Sea-Freight Value Chain for Mali's Mangos
	Figure 65	Multi-Modal Value Chain for Mali's Mangos
Figure 66 Malian Mango Exports in Metric Tons	Figure 66	Malian Mango Exports in Metric Tons
Figure 67 Malian Mango Exports to TFC Holland in Metric Tons	Figure 67	Malian Mango Exports to TFC Holland in Metric Tons
Figure 68 Value of Sri Lankan Cinnamon Exports	Figure 68	Value of Sri Lankan Cinnamon Exports
Figure 69 Cinnamon Export Value Chain	Figure 69	Cinnamon Export Value Chain
Figure 70 Mapping Actual and Potential Business and Financial Services	Figure 70	Mapping Actual and Potential Business and Financial Services
Figure 71 Zambian Cotton Exports, 1990–1994	Figure 71	Zambian Cotton Exports, 1990–1994
Figure 72 Zambia's Cotton Value Chain	Figure 72	Zambia's Cotton Value Chain
Figure 73 Zambian Competition for Cottonseed	Figure 73	Zambian Competition for Cottonseed
Figure 74 The Dunavant Distributor Model	Figure 74	The Dunavant Distributor Model
Figure 75 Zambian Cotton Exports Turn Around	Figure 75	Zambian Cotton Exports Turn Around
Figure 76 Bostwana's Red Meat Value Chain	Figure 76	Bostwana's Red Meat Value Chain
Figure 77 Locations of Botswanan Abattoirs	Figure 77	Locations of Botswanan Abattoirs
Figure 78 Exports of Botswanan Beef, 1990–2004	Figure 78	Exports of Botswanan Beef, 1990–2004
Figure 79 Recommendations for the Botswanan Red-Meat Value Chain	Figure 79	Recommendations for the Botswanan Red-Meat Value Chain
Figure 80 Initial Implementation of Recommendations	Figure 80	Initial Implementation of Recommendations
Figure 81 Next Steps for the Botswanan Red-Meat Value Chain	Figure 81	Next Steps for the Botswanan Red-Meat Value Chain
Figure 82 Kenya's Cut Flower Cluster	Figure 82	Kenva's Cut Flower Cluster
Figure 83 Kenyan Avocado Production, 1994–2003	Figure 83	Kenvan Avocado Production, 1994–2003
Figure 84 Kenyan Avocado Value Chain	Figure 84	Kenyan Avocado Value Chain
Figure 85 Kenyan Avocado Value Chain and Cluster	Figure 85	Kenyan Avocado Value Chain and Cluster
Figure 86 Kenyan Avocado Exports, 1975–2005	Figure 86	Kenyan Avocado Exports, 1975–2005
Figure 87 M&E VC Model	Figure 87	M&E VC Model

Figure 88Sample SPEG Customer Showing Desired Pineapple Qualities